UNIVERSITY OF AMSTERDAM

MSC MATHEMATICS

MASTER THESIS

Coalgebraic geometric logic

Author: Jim de Groot

Examination date: 27 June 2018

Supervisors: Prof. dr. Yde Venema Dr. Nick Bezhanishvili

Korteweg-de Vries institute for mathematics



Institute for Logic, Language and Computation



Abstract Coalgebras are category-theoretic constructions which can be used to model a wide variety of phenomena. Logic is used as a tool for reasoning about properties of coalgebras. In this thesis, we briefly review some known facts about coalgebraic logic and geometric logic, whereupon we develop coalgebraic geometric logic for coalgebras whose state spaces are topological spaces. We define and investigate notions of equivalence and behaviour: various notions bisimulation, modal equivalence and behavioural equivalence. Furthermore, we give a method of lifting a set functor together with a collection of predicate liftings to a sober functor (and a set of *open* predicate liftings for this new functor). Throughout, we connect results with the guiding examples of monotone logic and conditional logic.

Title: Coalgebraic geometric logic Author: Jim de Groot, jim.degroot@student.uva.nl, 6265898 Supervisors: Prof. dr. Yde Venema and Dr. Nick Bezhanishvili Second examiner: Prof. dr. Jan van Mill Examination date: June 27, 2018

Korteweg-de Vries Institute for Mathematics University of Amsterdam Science Park 105–107, 1098 XG Amsterdam http://kdvi.uva.nl

Institute for Logic, Language and Computation University of Amsterdam Science Park 107, 1098 XG Amsterdam http://illc.uva.nl Acknowledgements I would like to thank my supervisors Yde Venema and Nick Bezhanishvili for their outstanding supervision during this project. They introduced to the wonderous world of coalgebraic logic, helped me when I got stuck and guided me when I was lost.

Furthermore, I would like to thank Sebastian Enqvist for the helpful discussion about the construction of final coalgebras, which helped prove theorem 3.22; Clemens Kupke for his help with the proof of lemma A.6 and his useful suggestions for the proof of theorem 3.22; Helle Hansen for the insightful discussions about (generalisations of) Λ bisimulations; and Jan van Mill for being my second reader.

Contents

1	Introduction	5	
2	Coalgebras and coalgebraic logic 2.1 Coalgebras 2.2 Set-based coalgebraic logic	9 9 14	
	2.3 Stone-based coalgebraic logic	16	
3	Coalgebraic geometric logic	24	
	3.1 Geometric logic and duality	25	
	3.2 Coalgebraic geometric logic	29	
	3.3 Examples	37	
	3.4 Bisimulations	42	
4	Lifting functors	47	
	4.1 From Set to Sob	47	
	4.2 From Set to Stone	52	
	4.3 The pro-completion lift from Set to Stone	54	
5	Coalgebraic conditional logic 6		
	5.1 Conditional logic on Stone-coalgebras	60	
	5.1.1 Descriptive conditional models	60	
	5.1.2 Duality	66	
	5.1.3 Bisimulations between descriptive conditional models	68	
	5.2 Geometric conditional logic	72	
	5.2.1 Geometric conditional frames	72	
	5.2.2 An Isbell dual for \mathbb{C}_{kh}	75	
6	Conclusion 8		
7	Populaire samenvatting 83		
Bi	ibliography	83	
т		00	
IU	luex	88	
Α	Appendix	89	
	A.1 Notation	89	
	A.2 The monotone functor on Frm	90	
	A.3 For reference	94	

Introduction

The content of this thesis lies on the intersection of coalgebra, topology and modal logic.

A T-coalgebra is a pair (X, γ) where T is an endofunctor on some category C, X is an object in C and $\gamma: X \to \mathbb{T}X$ is a morphism in C. The object X is also referred to as the state space, γ as the structure map and the category C is called the base category of the coalgebra (X, γ) . Intuitively, the functor T captures the possible outcomes of the structure map applied to a state, and the structure map γ describes the dynamics of the coalgebra (X, γ) .

Coalgebras come with their own generic notions of morphisms, bisimilarity and behaviour and hence are useful to reason about notions related to behaviour and observational indistinguishability. Although the definition of a coalgebra may seem rather abstract, they model a wide variety of structures.

One of the simplest structures which can be described as a coalgebra is a Kripke frame [34, 10]. Kripke frames correspond one-to-one with coalgebras for the powerset functor \mathbb{P} on Set, the category of sets and functions [1]. Moreover, the standard notion of a morphism between Kripke frames corresponds precisely to \mathbb{P} -coalgebra morphisms and the coalgebraic definition of a bisimulation coincides with the recognised definition of bisimulation from modal logic. Even in this simple setting where the base category is Set, many other familiar structures can be viewed as coalgebras as well. Among these are transition systems, non-wellfounded sets and deterministic automata [3, 48]. Set-based coalgebras are also called systems and are well researched [49, 28, 48, 27].

The generality of the theory of coalgebra allows for results which are uniform in \mathbb{T} . Results on the level of coalgebra can then be applied to structures corresponding to a particular choice of \mathbb{T} . Besides, logic can be used as a tool for reasoning about properties of coalgebras, such as bisimilarity. Moss was the first to generalise the concept of modal logic from Kripke frames and models to coalgebraic logic for arbitrary setbased coalgebras [43]. He used so-called relation liftings to define modal operators for propositional logic. This triggered much more research in the area [24, 25, 47, 35, 21, 50, 22]. In [47] Pattinson introduces a different method for defining modal operators, namely via predicate liftings. Coalgebraic logic for set-based coalgebras has been well investigated and is still an active area of research [55, 56, 14, 38].

Coalgebras where the state space is not a set are also useful in modelling various phenomena. For example, trace semantics for non-deterministic automata and context-free grammars can be obtained by modelling these systems as coalgebras over the category of sets and relations [23, 27], and the descriptive frames of modal logic are coalgebras over the category **Stone** of Stone spaces and continuous functions [37]. Some advances have been made on coalgebraic logic for coalgebras whose underlying spaces are Stone spaces [16, 9]. Research about (logics for) coalgebras over arbitrary topological spaces is more scarce, as observed in [44].

In this thesis we investigate coalgebras whose state space is a topological space, but not necessarily a Stone space. That is, we let the base category be some full subcategory of **Top**, the category of topological spaces and continuous functions. Firstly, this is motivated by mathematical curiosity. For instance, the aforementioned descriptive frames are coalgebras for the Vietoris functor on **Stone**. But really, the Vietoris functor is defined on the full category **Top** and can be restricted to **Stone**. This raises the question what coalgebras look like for this definition of the Vietoris functor. Of course, there are many more functors on **Top** whose coalgebras might be of interest. Secondly, in [44] it is suggested that coalgebras over **KHaus**, the category of compact Hausdorff spaces and continuous functions may be useful in economic theory.

The clopen sets of a Stone space are a subbase for the topology. Moreover, they form a Boolean algebra, hence they are closed under taking complements, finite intersections and finite unions. Therefore, if we use the clopen sets as the interpretants for propositional statements, the logic used to study Stone coalgebras, should contain negation, conjunction and disjunction. Since the empty set serves as a bottom element, this logic is just classical propositional logic. The method of predicate lifting then allows one to define additional modal operators.

However, as soon as one leaves the realm of set- and Stone-based coalgebras, classical propositional logic seizes to be a suitable logic to build upon. This is, in part, due to the fact that the open sets of a topological space, which are a natural choice of the interpretants of propositional statements, are not generally closed under taking complements. Therefore, the coalgebraic logic used to study coalgebras over arbitrary topological spaces must be based on some language without negations. An immediate question which arises is what logic we should use to build *coalgebraic logic* on for these coalgebras.

One of the natural candidates for this logic is geometric logic. The language of geometric logic is constructed from a set of propositional statements, arbitrary disjunctions and finite conjunctions [59, 60, 61]. We will see that geometric logic can be viewed as the logic of finite observations. Formulas of geometric logic can be interpreted in the frame of open sets of a topological space. There is a duality between the category of spatial frames and homomorphisms and the full subcategory of **Top** whose objects are sober spaces, which is central to the theory of geometric logic [62].

We modify the method of predicate lifting [47] and use it to define modal operators for geometric logic, which can then be interpreted in models based on coalgebras with a topological space as state space. The duality between sober spaces and spatial frames allows us to view problems from different perspectives. For example, the dual perspective of an enfunctor \mathbb{T} on Sob, the category of sober spaces and continuous functions, gives rise to a concrete construction of a final coalgebra in Coalg(\mathbb{T}).

The aim of this thesis is to develop a theoretical framework of coalgebraic geometric logic. In particular, we investigate the notions of open predicate liftings, geometric models, modal equivalence, bisimilarity and behavioural equivalence.

Outline of the thesis Chapter 2 sets the stage for the rest of the thesis; we fix notation and introduce formally the structures that will play a role in later chapters. It starts with the definition of a coalgebra, concrete examples of coalgebras, and the notions of coalgebra bisimulation and behavioural equivalence. Subsequently, in section 2.2, we define models over set-based coalgebras and we introduce coalgebraic logic which can be interpreted on these models. Furthermore, a different notion of bisimilarity, Λ -

bisimilarity, is presented. In section 2.3 we change the base category of interest to the category of Stone spaces and continuous functions. We generalise Λ -bisimilarity to a notion of bisimilarity between models over Stone coalgebras and investigate how this relates to modal equivalence and behavioural equivalence.

In chapter 3 we focus on coalgebras whose base category is a full subcategory of Top. We start with the definition of geometric logic and thereafter define so-called open predicate liftings and the *coalgebraic* geometric logic induced by a set of open predicate liftings. The models that we use to interpret this logic are coalgebras with a valuation. We investigate the relation between modal equivalence and behavioural equivalence between such models. Subsequently, we study a concrete example of the functor \mathbb{D}_{kh} , the monotone functor on KHaus, and provide a dual description of this functor in terms of frames, called \mathbb{M}_{kr} . Finally, in section 3.4, we define Λ -bisimulations and see how this relates to modal equivalence and behavioural equivalence.

Chapter 4 is devoted to lifting endofunctors from Set to other categories. In section 4.1 we show how one can lift a set functor together with a set of predicate liftings to a sober functor, i.e. an endofunctor on the category of sober spaces and continuous functions. We show that lifting the powerset functor and the monotone functor on Set together with the usual set of predicate liftings yields the Vietoris functor and the monotone functor on KHaus, respectively. The content of section 4.2 is similar to that of section 4.1, but we lift functors to Stone instead. In section 4.3, a different method for lifting a set functor to a Stone is given. We show that this coincides with the method from section 4.2. This provides a partial solution to a question raised in the conclusion of [36].

The final chapter of this thesis is a case study, where we put the developed theory into practice. In section 5.1 we define so-called *descriptive conditional frames*. These generalise conditional frames in the same manner descriptive general frames generalise Kripke frames. We show that descriptive conditional frames are coalgebras for a certain functor \mathbb{C}_{st} on Stone, which arises as the lift of the conditional functor on Set. Moreover, we provide an endofunctor on BA, the category of Boolean algebras and homomorphisms, which is dual to \mathbb{C}_{st} . Besides, we define the notion of descriptive conditional bisimilarity, which turns out the be the equivalent to Λ -bisimilarity but differs in the fact that its definition is structural, whereas Λ -bisimilarity is defined in a non-structural manner. In section 5.2 we investigate geometric conditional frames. These are coalgebras for the functor \mathbb{C}_{kh} on KHaus, which arises from lifting the conditional functor on Set using the method from section 4.1. We give an endofunctor on Frm and prove that its restriction to KRFrm is dual to \mathbb{C}_{kh} .

In appendix A.2 we give a different proof of the fact that \mathbb{M}_{kr} from chapter 3 preserves compactness, which does not depend on the duality with \mathbb{D}_{kh} .

Most notable theorems

- In section 2.3 we define so-called A-bisimulations between models over Stonecoalgebras. In **propositions 2.29 and 2.34 and lemma 2.32** this notion is related to modal equivalence and behavioural equivalence.
- In section 3.4 we define Λ-bisimulations between models for coalgebraic geometric logic. Theorem 3.57 describes the relation of Λ-bisimilarity with modal equivalence and behavioural equivalence.

- For a certain class of set functors, there are two ways of lifting a set functor to a Stone functor. **Theorem 4.24** shows that these coincide on objects.
- We generalise the monotone functor (which is an endofunctor on Stone) to an endofunctor on KHaus and give a description of its algebraic dual, which is an endofunctor on Frm, in theorem 3.41. Similarly, we generalise the conditional functor (on Set) to an endofunctor on KHaus in section 5.2 and show that it has a dual functor on Frm in theorem 5.38.
- Descriptive conditional bisimilarity is a structural notion of bisimilarity between descriptive conditional frames. Theorem 5.23 shows that descriptive conditional bisimilarity, Λ -bisimilarity (for certain Λ), modal equivalence and behavioural equivalence coincide.
- In chapter 3 we use duality to prove that the monotone functor \mathbb{M} on Frm preserves compactness. In **theorem A.3** we give a proof of this fact which only plays on the frame side and does not use the duality with \mathbb{D}_{kh} .

Prerequisites We assume familiarity with basic topology and category theory. Excellent references for these topics are [5, 45, 40]. Furthermore, familiarity with modal logic will be helpful in providing intuition. We refer to [10] for an outstanding introduction to basic modal logic. Besides, the reader is advised to have a look at section A.1, where notational conventions are explained.

Coalgebras and coalgebraic logic

This chapter sets the stage for the rest of the thesis. We present mostly known facts about coalgebra and coalgebraic logic. The main purpose is to fix notation and, through examples, introduce some structures that will play a role in later chapters. We assume familiarity with basic category theory, as standard reference we use [40].

Section 2.1 introduces coalgebras and related concepts of equivalence for an arbitrary category C. Some concrete and some more abstract examples are given. Section 2.2 focusses on the case C = Set and introduces coalgebraic logic for set-coalgebras. Section 2.3 concentrates on the case C = Stone and coalgebraic logic for Stone-coalgebras. So-called Λ -bisimulations are introduced, which are a straightforward generalisation from the same notion on set coalgebras [6], and various notions of equivalence between coalgebras are compared.

2.1 COALGEBRAS

In this section we define coalgebras and corresponding notions of equivalence. We also give a number of illustrative examples. For a thorough (yet accessible) introduction to the theory of coalgebras we refer to [27, 49, 28].

2.1 Definition. Let C be a category and T an endofunctor on C. A T-coalgebra is a pair (X, γ) where X is an object in C, also referred to as the state space, and $\gamma: X \to \mathbb{T}X$ is a morphism in C, known as the transition map or structure map. A T-coalgebra morphism between two T-coalgebras (X, γ) and (X', γ') is a morphism $f: X \to X'$ in C such that the following diagram commutes:

$$\begin{array}{ccc} X & \stackrel{f}{\longrightarrow} & X' \\ \gamma & & & \downarrow \gamma' \\ \mathbb{T}X & \stackrel{T}{\longrightarrow} & \mathbb{T}X' \end{array}$$

The collection of \mathbb{T} -coalgebras and \mathbb{T} -coalgebra morphisms forms a category, which we shall denote by $\mathsf{Coalg}(\mathbb{T})$. The category C is called the **base category** of $\mathsf{Coalg}(\mathbb{T})$. \triangleleft

If the functor \mathbb{T} is clear from the context, we will simply refer to (X, γ) as a coalgebra and to f as a coalgebra morphism or a coalgebra map. In case \mathbb{T} is an endofunctor on Set, \mathbb{T} -coalgebras are also known as **systems**. The standard reference for the theory of systems is [49].

Many known structures can be viewed as a coalgebra. We give some examples.

2.2 Example (Transition systems). Consider the following example of a transition system. Here x_1, x_2, x_3 represent states and the arrows describe the relation between these states.



This can be viewed as a \mathbb{P} -coalgebra (X, γ) with state space $X = \{x_1, x_2, x_3\}$ and transition map $\gamma : X \to \mathbb{P}X$ given by $\gamma(x_1) = \{x_2, x_3\}, \gamma(x_2) = \{x_1, x_3\}$ and $\gamma(x_3) = \emptyset$. The coalgebra (X, γ) then encodes all information of the transition system, that is, given (X, γ) we can recover the given transition system.

In general, transition systems are pairs (X, R) where X is a set and $R \subseteq X \times X$ a relation on X. A transition system (X, R) corresponds to the \mathbb{P} -coalgebra (X, γ) where $\gamma : X \to \mathbb{P}X : x \mapsto \{x' \in X \mid xRx'\}$. Conversely, every \mathbb{P} -coalgebra (X, γ) gives rise to a transition system (X, R) with $R \subseteq X \times X$ defined by xRx' iff $x' \in \gamma(x)$. These constructions give a one-to-one correspondence between transition systems and \mathbb{P} -coalgebras. \triangleleft

2.3 Example (Labelled transition systems). Let us fix a set of labels A and label the transition system from the previous example with $a_1, \ldots, a_4 \in A$.



The result is a labelled transition system (LTS). To make this into a coalgebra we need to adapt our functor. Let $Lab: Set \rightarrow Set$ be the functor defined by

$$\operatorname{Lab} X = \mathbb{P}(A \times X)$$

for a set X and

$$Lab(f)(V) = \{(a, f(x)) \mid (a, x) \in V\}$$

for functions $X \to X'$ and $V \in \mathbb{P}(A \times X)$.

The information of the LTS can be encoded as a Lab-coalgebra as follows: let $X = \{x_1, x_2, x_3\}$ and define $\gamma : X \to Lab X$ by

$$\gamma: \left\{ \begin{array}{l} x_1 \mapsto \{(a_2, x_2), (a_4, x_3)\} \\ x_2 \mapsto \{(a_1, x_1), (a_3, x_3)\} \\ x_3 \mapsto \emptyset \end{array} \right.$$

Then (X, γ) corresponds to the LTS above, i.e. we can retrieve the given LTS from (X, γ) .

More generally, a labelled transition system with labels in A is a pair (X, \rightarrow) where X is a set and $\rightarrow \subseteq X \times A \times X$ a labelled transition map. An LTS (X, \rightarrow) corresponds to the Lab-coalgebra (X, γ) where γ is defined by $\gamma(x) = \{(a, x') \in A \times X \mid (x, a, x') \in \rightarrow\}$. Conversely, given a Lab-coalgebra (X, γ) we can retrieve the LTS corresponding to it by defining $\rightarrow \subseteq X \times A \times X$ by $(x, a, x') \in \rightarrow$ iff $(a, x') \in \gamma(x)$. This gives a 1-1 correspondence between labelled transition systems with labels in A and Lab-coalgebras. For details see example 2.1 in [49].

2.4 Example (Weighed transition systems). Suppose we only want to look at labelled transition systems with labels in the set $\mathbb{R}_{\geq 0}$ such that for each state x the sum of the labels of all outgoing arrows is equal to some fixed $n \in \mathbb{R}_{\geq 0}$. Then we have to modify our functor Lab to Lab_n: Set \rightarrow Set, defined on objects by

$$\operatorname{Lab}_{n} X = \{ V \subseteq \mathbb{R}_{\geq 0} \times X \mid \sum \{ a \mid \exists x \in X \text{ with } (a, x) \in V \} = n \}.$$

For a function $f: X \to X'$ and $V \in \mathbb{L}ab_n X$ let

$$\mathbb{L}ab_n(f)(V) = \{(b_x, f(x)) \mid (a, x) \in V \text{ and } b_x = \sum \{c \mid (c, x) \in V\}\}.$$

It is an easy exercise to see that the aforementioned LTSs correspond precisely to $\mathbb{L}ab_n$ coalgebras.

2.5 Remark. Lab₁-coalgebras are precisely discrete time Markov chains (cf. [33, 46]). A slightly different functor to make Markov chains coalgebraic and many more examples of coalgebras of probabilistic systems can be found in [52].

2.6 Example (Kripke frames). In modal logic, the transition systems of example 2.2 are better known as **Kripke frames**. The standard notion of morphisms between Kripke frames (X, R) and (X', R') is that of a bounded morphism: a set-map $f : X \to X'$ is a **bounded morphism** if for all $x, y \in X$ and $z' \in X'$ we have

- (i) Rxy implies R'f(x)f(y); and
- (ii) R'f(x)z' implies that there exists $z \in X$ with Rxz and f(z) = z'.

It is not hard to show that f is a bounded morphism from (X, R) to (X', R') iff it is a coalgebra morphism between the corresponding \mathbb{P} -coalgebras. This yields the following isomorphism of categories

$$\mathsf{Krip} \cong \mathsf{Coalg}(\mathbb{P}),$$

where Krip denotes the category of Kripke frames and bounded morphisms. \triangleleft

Kripke frames play an important role in modal logic as they are the structures used to interpret *basic modal logic*. The following two examples are the structures that correspond to *monotone modal logic* and *conditional logic*. These will be key ingredients in guiding examples in subsequent chapters.

2.7 Example (Monotone frames). A monotone frame is a pair (X, γ) where X is a set and $\gamma: X \to \check{\mathbb{P}}(\check{\mathbb{P}}X)$ is a monotone neighbourhood function. That is, γ assigns to each state $x \in X$ a collection of subsets of X, called neighbourhoods of x, and whenever $a \in \gamma(x)$ and $a \subseteq b \subseteq X$, we have $b \in \gamma(x)$. A bounded morphism between monotone frames (X, γ) and (X', γ') is a map $f: X \to X'$ such that for all $x \in X$ and $a' \subseteq X'$ we have $f^{-1}[a'] \in \gamma(x)$ iff $a' \in \gamma'(f(x))$.

Monotone frames are coalgebras for the functor \mathbb{D} : Set \rightarrow Set given by

$$\mathbb{D}X = \{ W \subseteq \mathbb{P}X \mid \text{ if } a \in W \text{ and } a \subseteq b \text{ then } b \in W \}.$$

For a morphism $f: X \to X'$ define

$$\mathbb{D}f:\mathbb{D}X\to\mathbb{D}X':W\mapsto\{a'\in\mathbb{P}X'\mid f^{-1}(a')\in W\}.$$

The bounded morphisms correspond precisely to coalgebra morphisms [21, 20, 13]. \triangleleft

In literature, morphisms between Kripke frames and morphisms between monotone frames are both called *bounded morphisms*. It will always be clear from the context which notion of bounded morphisms we mean.

Our next example will be that of conditional frames and conditional frame morphisms. One may find various definitions of conditional frames in the literature [38, 7, 13]. However, conditional frame morphisms have not been defined. In order to avoid confusion let us define the notion of conditional frames, taken from [7], and the corresponding morphisms that we will use here.

2.8 Definition. A conditional frame is a pair (X, ν) where X is a set and $\nu : X \times \mathbb{P}X \to \mathbb{P}X$ a function that satisfies for all $x \in X$ and $a, b \in \mathbb{P}X$

- (i) if $a \cap b = \emptyset$, then $\nu(x, a) \cap b = \emptyset$; and
- (ii) if $a \subseteq b$ and $\nu(x, b) \subseteq a$ then $\nu(x, a) = \nu(x, b)$.

A map $f : X \to X'$ is a **conditional frame morphism** between conditional frames (X, ν) and (X', ν') if for all $x \in X$ and $a' \subseteq X'$,

$$f[\nu(x, f^{-1}(a'))] = \nu'(f(x), a').$$
(2.1)

This definition is motivated by the fact that f is a conditional frame morphism iff the following diagram commutes,

$$\mathbb{P}X \xrightarrow{f^{-1}} \mathbb{P}X'$$

$$\nu(x,-) \downarrow \qquad \qquad \downarrow \nu'(f(x),-)$$

$$\mathbb{P}X \xrightarrow{f[-]} \mathbb{P}X'$$

We write CF for the category of conditional frames and conditional frame morphisms. \triangleleft

2.9 Remark. Condition (i) in the previous definition can be reformulated as $\nu(x, a) \subseteq a$. Therefore our conditions are equivalent to the ones in definition 1 of [7]. We have chosen this slightly altered formulation in view of chapter 5, where they make a difference when dealing with geometric conditional frames.

Let us adopt a coalgebraic perspective on conditional frames and their morphisms.

2.10 Example (Conditional frames). Conditional frames are coalgebras for the functor \mathbb{C} . For a set X, $\mathbb{C}X$ is the collection of functions $h : \mathbb{P}X \to \mathbb{P}X$ that satisfy

(C1) if $a \subseteq X$ then $h(a) \subseteq a$; and

(C2) if $a \subseteq b \subseteq X$ and $h(b) \subseteq a$ then h(a) = h(b).

For a function $f : X \to X'$ define $\mathbb{C}f : \mathbb{C}X \to \mathbb{C}X'$ by $\mathbb{C}f(h)(a) = f[h(f^{-1}(a))]$. We need to check that this is a well-defined functor, i.e. $\mathbb{C}f(h) \in \mathbb{C}X'$ for $h \in \mathbb{C}$. For (C1), let $a \subseteq X'$, then

$$\mathbb{C}f(h)(a) = f[h(f^{-1}(a))] \subseteq f[f^{-1}(a)] \subseteq a.$$

For (C2), assume $a \subseteq b$ and $\mathbb{C}f(h)(b) \subseteq a$. We need to show that $\mathbb{C}f(h)(a) = \mathbb{C}f(h)(b)$. Since $a \subseteq b$ we know $f^{-1}(a) \subseteq f^{-1}(b)$ and because $\mathbb{C}f(h)(b) = f[h(f^{-1}(b))] \subseteq a$ we know $h(f^{-1}(b)) \subseteq f^{-1}(a)$. Now we may apply (C2) to $\mathbb{C}X$ to find $h(f^{-1}(a)) = h(f^{-1}(b))$, from which it follows that

$$\mathbb{C}f(h)(a) = f[h(f^{-1}(a))] = f[h(f^{-1}(b))] = \mathbb{C}f(h)(b).$$

So \mathbb{C} is well defined on morphisms.

There is an isomorphism $\mathsf{CF} \cong \mathsf{Coalg}(\mathbb{C})$ which is given on objects by observing that functions $X \times \mathbb{P}X \to \mathbb{P}X$ satisfying (i) and (ii) from definition 2.8 correspond one-to-one with elements of the set $\mathbb{C}X$. Furthermore, conditional frame morphisms are tailored to coincide with \mathbb{C} -coalgebra morphisms. It is routine to check the details.

2.11 Remark. If we omit (i) and (ii) from definition 2.8 we get the definition of a **selection function frame**. These are known to be coalgebraic [38]. Although the authors only consider the frames, not the morphisms, it is easy to see that the above notion of a morphism is precisely a coalgebra morphism for the functor given in [38].

More examples showcasing the wide scope of coalgebra can be found in a variety of areas such as biology [65], economics [44] and quantum computing [2, 26].

There are two standard notions of equivalence for coalgebras: coalgebra bisimilarity and behavioural equivalence. It is a well-known fact that these two notions coincide for many choices of the functor \mathbb{T} , namely if \mathbb{T} preserves weak pullbacks. These notions will be used throughout this thesis.

2.12 Definition. Let C be a category which has products and a forgetful functor \mathbb{Y} : $C \to Set$. Let T be an endofunctor on C, let (X, γ) and (X', γ') be T-coalgebras, and let $x \in \mathbb{Y}X$ and $x' \in \mathbb{Y}X'$. The states x and x' are called **behaviourally equivalent**, $x \simeq x'$, if there exist a coalgebra (Y, δ) and coalgebra morphisms $f : (X, \gamma) \to (Y, \delta)$ and $f' : (X', \gamma') \to (Y, \delta)$ such that f(x) = f'(x').

Let B be an object in C such that $\mathbb{Y}B \subseteq \mathbb{Y}X \times \mathbb{Y}X'$, with projections $\pi: B \to X$ and $\pi': B \to X'$. B is called a **coalgebra bisimulation** or **Aczel-Mendler bisimulation** between (X, γ) and (X', γ') if there exists a transition map $\beta: B \to \mathbb{T}B$ that makes π and π' coalgebra morphisms. That is, β is such that the following diagram commutes:

$$\begin{array}{cccc} X & & & & & & B & \xrightarrow{\pi'} & X' \\ \gamma & & & & & & \downarrow \beta & & & \downarrow \gamma' \\ \mathbb{T}X & & & & & \mathbb{T}B & \xrightarrow{\pi'} & X' \end{array}$$

Two states $x \in \mathbb{U}X, x' \in \mathbb{U}X'$ are called **bisimilar**, notation $x \Leftrightarrow x'$, if they are linked by a coalgebra bisimulation.

Finally, recall the definition of a final object in a category.

2.13 Definition. An object X in a category C is called **final** if for all objects X' in C there exists a unique morphism $f: X' \to X$. A \mathbb{T} -coalgebra is called final if it is a final object in $\mathsf{Coalg}(\mathbb{T})$.

Let $\mathbb{T} : \mathbb{C} \to \mathbb{C}$ be a functor and suppose the category $\mathsf{Coalg}(\mathbb{T})$ has a final object (Z,ζ) . For each object $\mathfrak{X} = (X,\gamma)$ in $\mathsf{Coalg}(\mathbb{T})$ let $f_{\mathfrak{X}} : X \to Z$ be the unique coalgebra map to (Z,ζ) . Then it is an easy consequence of finality of (Z,ζ) that two states x and x' in two coalgebras (X,γ) and (X',γ') are behaviourally equivalent if and only if $f_{\mathfrak{X}}(x) = f_{\mathfrak{X}'}(x')$. In fact, originally, behavioural equivalence was only defined for categories with a final object; two states were called behaviourally equivalent if they were mapped to the same element under the unique maps to the final object. Definition 2.13 is equivalent to this one in case the category has a final object, but can also be used for categories without a final object.

2.2 Set-based coalgebraic logic

In this section we briefly describe how set-based coalgebras relate to logic. Much more information can be found in e.g. [38, 55, 39, 14]. We start by introducing the idea of a so-called predicate lifting, corresponding language, and models for interpreting this language. Thereafter a different notion of bisimulation is given and some examples are examined.

2.14 Definition. Let \mathbb{T} be an endofunctor Set. A predicate lifting for \mathbb{T} of arity n is a natural transformation

$$\lambda: \mathbb{P}^n \to \mathbb{P} \circ \mathbb{T}.$$

The **dual** of an *n*-ary predicate lifting λ is given by

$$\lambda_X^{\partial} : \check{\mathbb{P}}^n X \to \check{\mathbb{P}} X : (a_1, \dots, a_n) \mapsto \mathbb{T} X \smallsetminus \lambda(X \smallsetminus a_1, \dots, X \smallsetminus a_n).$$

A collection Λ of predicate liftings for \mathbb{T} is called a **similarity type** (for \mathbb{T}), and is said to be **closed under duals** if $\lambda \in \Lambda$ implies $\lambda^{\partial} \in \Lambda$. A similarity type Λ is **separating** for \mathbb{T} if for all sets X and all distinct $x, x' \in \mathbb{T}X$ there exists a $\lambda \in \Lambda$ and $a_1, \ldots, a_n \in \mathbb{P}X$ such that precisely one of x, x' belongs to the set $\lambda_X(a_1, \ldots, a_n)$.

Fix a set Φ of proposition letters.

2.15 Definition. Let \mathbb{T} be a functor on Set. A \mathbb{T} -model is a triple $\mathfrak{X} = (X, \gamma, V)$ where (X, γ) is a \mathbb{T} -coalgebra and $V : \Phi \to \mathbb{P}X$ is a valuation. A \mathbb{T} -model morphism f from (X, γ, V) to (X', γ', V') is a \mathbb{T} -coalgebra morphism $f : (X, \gamma) \to (X', \gamma')$ such that $f^{-1} \circ V' = V$. The collection of \mathbb{T} -models and \mathbb{T} -model morphisms forms a category, $\mathsf{Mod}(\mathbb{T})$.

An Aczel-Mendler bisimulation between two T-models is an Aczel-Mendler bisimulation between the underlying T-coalgebras such that for all $(x, x') \in B$ and $p \in \Phi$, $x \in V(p)$ iff $x' \in V'(p)$.

Every similarity type induces a modal language that we can interpret on T-models.

2.16 Definition. The **language** induced by the similarity type Λ is the set $\mathcal{L}(\Lambda)$ of formulas defined by

$$\varphi ::= \bot \mid p \mid \neg \varphi \mid \varphi_1 \land \varphi_2 \mid \heartsuit^{\lambda}(\varphi_1, \dots, \varphi_n),$$

where $p \in \Phi$ and $\lambda \in \Lambda$ is *n*-ary. The symbols \top, \lor, \rightarrow and \leftrightarrow denote the usual abbreviations. The **semantics** of $\varphi \in \mathcal{L}(\Lambda)$ on a \mathbb{T} -model $\mathfrak{X} = (X, \gamma, V)$ is given inductively by

$$\llbracket p \rrbracket^{\mathfrak{X}} = V(p), \quad \llbracket \varphi_1 \wedge \varphi_2 \rrbracket^{\mathfrak{X}} = \llbracket \varphi_1 \rrbracket^{\mathfrak{X}} \cap \llbracket \varphi_2 \rrbracket^{\mathfrak{X}}, \quad \llbracket \neg \varphi \rrbracket^{\mathfrak{X}} = X \smallsetminus \llbracket \varphi \rrbracket^{\mathfrak{X}}, \\ \llbracket \Diamond^{\lambda}(\varphi_1, \dots, \varphi_n) \rrbracket^{\mathfrak{X}} = \gamma^{-1}(\lambda(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}})),$$

where $p \in \Phi$ and λ ranges over Λ .

Besides Aczel-Mendler bisimulations, other notions of bisimulations between \mathbb{T} -coalgebras and \mathbb{T} -models have been proposed and linked to modal equivalence and behavioural equivalence. We content ourselves with stating the definition of a so-called Λ -bisimulation for future reference, and refer to [17, 6, 22] for more information.

 \triangleleft

2.17 Definition. Let $B \subseteq X \times X'$ be a relation. A pair $(a, a') \in \mathbb{P}X \times \mathbb{P}X'$ is called *B*-coherent if $B[a] \subseteq a'$ and $B^{-1}[a'] \subseteq a$.

Properties of coherent pairs of sets may be found in [22]. We prove one property which is not in [22] for future reference.

2.18 Lemma. Let $B \subseteq X \times X'$ be a relation and (a,a') a B-coherent pair. Then $(X \setminus a, X' \setminus a')$ is B-coherent.

Proof. Assume towards a contradiction that $B[X \setminus a] \notin X' \setminus a'$, then $B[X \setminus a] \cap a' \neq \emptyset$, so some element in $X \setminus a$ is related to an element in a'. But then $B^{-1}[a'] \notin a$, a contradiction. Therefore we must have $B[X \setminus a] \subseteq X' \setminus a'$. In a similar way it can be shown that $B^{-1}[X' \setminus a'] \subseteq X \setminus a$.

2.19 Definition. Let \mathbb{T} be a set functor, i.e. an endofunctor on Set, Λ a collection of predicate liftings for \mathbb{T} and (X, γ, V) and (X', γ', V') two \mathbb{T} -models. A relation $B \subseteq X \times X'$ is called a Λ -bisimulation if for all $\lambda \in \Lambda$, $(x, x') \in B$ and B-coherent pairs (a_i, a'_i) we have

- $x \in V(p)$ iff $x' \in V'(p)$;
- $\gamma(x) \in \lambda_X(a_1, \ldots, a_n)$ iff $\gamma'(x') \in \lambda_{X'}(a'_1, \ldots, a'_n)$.

Two states are called Λ -bisimilar if they are linked by a Λ -bisimulation.

 \triangleleft

The remainder of this section is devoted to examples.

2.20 Example (Normal modal logic). In example 2.6 we showed that \mathbb{P} -coalgebras correspond precisely to Kripke frames. Define $\lambda^{\Box} : \check{\mathbb{P}} \to \check{\mathbb{P}} \circ \mathbb{P}$ by $\lambda_X^{\Box}(a) = \{ \delta \in \mathbb{P}X \mid \delta \subseteq a \}$ and set $\Lambda = \{\lambda^{\Box}\}$. Then $\mathcal{L}(\Lambda)$ is the standard relational semantics for modal logic. We write \Box instead of $\heartsuit^{\lambda^{\Box}}$. If $\mathfrak{X} = (X, \gamma, V)$ is a Kripke model (a \mathbb{P} -model) and $\varphi \in \mathcal{L}(\Lambda)$ is a formula, then

$$\llbracket \Box \varphi \rrbracket^{\mathfrak{X}} = \{ x \in X \mid \gamma(x) \subseteq \llbracket \varphi \rrbracket^{\mathfrak{X}} \},\$$

so $\mathfrak{X}, x \Vdash \Box \varphi$ iff for all $y \in \gamma(x)$ we have $\mathfrak{X}, y \Vdash \varphi$. This yields the usual Kripke semantics of modal logic [10].

It is an easy exercise to show that $\{\lambda^{\Box}\}$ -bisimilar states satisfy precisely the same formulas. Moreover, every Kripke bisimulation is also a $\{\lambda^{\Box}\}$ -bisimulation [6, Example 3.3].

2.21 Example (Monotone modal logic). Example 2.7 shows that monotone frames are \mathbb{D} -coalgebras. Define $\lambda^{\Box} : \check{\mathbb{P}} \to \check{\mathbb{P}} \circ \mathbb{D}$ by $\lambda_X^{\Box}(a) = \{W \in \mathbb{D}X \mid a \in W\}$ and set $\Lambda = \{\lambda^{\Box}\}$. Then $\mathcal{L}(\Lambda)$ is the standard semantics of modal logic. Write \Box for $\heartsuit^{\lambda^{\Box}}$. If $\mathfrak{X} = (X, \gamma, V)$ is a neighbourhood model and φ a formula in $\mathcal{L}(\Lambda)$ then, similar to the previous example, we have $\mathfrak{X}, x \Vdash \Box \varphi$ iff $\llbracket \varphi \rrbracket^{\mathfrak{M}} \in \gamma(x)$. This yields the usual monotone semantics of modal logic [20, 21, 13]. As in the previous example, $\{\lambda^{\Box}\}$ -bisimilar states satisfy precisely the same formulas.

Next, we will look at conditional logic. Conditional logic provides an example of a non-monotone modality: the conditional implication, \Rightarrow . The modality \Rightarrow is meant to express a notion of conditionality which in general is different from the usual implication \rightarrow . A formula $\varphi_1 \Rightarrow \varphi_2$ should be read as "If φ_1 is the case, then usually φ_2 is the case."

For an example, suppose Morty usually visits his grandpa Rick on Fridays. This can be formalised as

Friday
$$\Rightarrow$$
 visit Rick.

However, if Morty is ill he will not visit his grandpa, so

Friday
$$\wedge$$
 ill $\Rightarrow \neg$ (visit Rick).

The non-monotonicity shows itself in the fact that the conclusion is not maintained if more information becomes available. For more information on conditional logic, see [13, 4, 42, 51, 7, 11].

2.22 Example (Conditional logic). The language of conditional logic is given by

$$\varphi ::= \bot \mid p \mid \neg \varphi \mid \varphi_1 \land \varphi_2 \mid \varphi_1 \Rightarrow \varphi_2,$$

with $p \in \Phi$. We abbreviate \vee and \top as usual, and let $\varphi_1 \Downarrow \varphi_2 \coloneqq \neg(\varphi_1 \Rightarrow \neg\varphi_2)$. A possible way to read $\varphi_1 \Rightarrow \varphi_2$ is as: "If φ_1 holds, then usually φ_2 holds as well." On a conditional model (viewed as \mathbb{C} -model, cf example 2.10) $\mathfrak{X} = (X, \gamma, V)$, truth of the proposition letters and of the Boolean cases is treated as usual. Truth of the implication is given by

$$\mathfrak{X}, x \Vdash \varphi_1 \Rightarrow \varphi_2 \quad \text{iff} \quad \gamma(x)(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}) \subseteq \llbracket \varphi_2 \rrbracket^{\mathfrak{X}}.$$

and consequently $\mathfrak{X}, x \Vdash \varphi_1 \Downarrow \varphi_2$ iff $\gamma(x)(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}) \cap \llbracket \varphi_2 \rrbracket^{\mathfrak{X}} \neq \emptyset$. The intuition behind this is that the function $\gamma(x) : \mathbb{P}X \to \mathbb{P}X$ indicates for each set $a \subseteq X$ the relevant states in a. We say that a state x satisfies $\varphi_1 \Rightarrow \varphi_2$ if the relevant states of $\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}$ as seen from x, are all contained in $\llbracket \varphi_2 \rrbracket^{\mathfrak{X}}$.

Define $\lambda^{\Rightarrow} : \check{\mathbb{P}}^2 \to \check{\mathbb{P}} \circ \mathbb{C}$ by $\lambda^{\Rightarrow}(a, b) = \{h : \mathbb{P}X \to \mathbb{P}X \mid h(a) \subseteq b\}$. Then

$$\mathfrak{X}, x \Vdash \varphi_1 \Rightarrow \varphi_2 \quad \text{iff} \quad \gamma(x) \in \lambda^{\Rightarrow}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \llbracket \varphi_2 \rrbracket^{\mathfrak{X}}).$$

This yields conditional semantics [7, 13]. Additionally we may define $\lambda^{\downarrow} : \mathbb{P}^2 \to \mathbb{P} \circ \mathbb{C}$ by $\lambda^{\downarrow}(a, b) = \{h : \mathbb{P}X \to \mathbb{P}X \mid h(a) \cap a \neq \emptyset\}$. Then we have $\mathfrak{X}, x \Vdash \varphi_1 \Downarrow \varphi_2$ iff $\gamma(x) \in \lambda^{\downarrow}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \llbracket \varphi_2 \rrbracket^{\mathfrak{X}})$.

2.23 Remark. The introduction of the modality \Downarrow may seem superfluous at this point. Indeed, it is only an abbreviation so we don't really need a predicate lifting to describe its truth. However, it turns out to be useful when generalising conditional logic for Stone-coalgebras in section 5.1. Besides, when dealing with geometric conditional logic (section 5.2) the modalities \Rightarrow and \Downarrow will no longer be mutually expressible, but relate via a weaker relational.

More examples of the interplay between logic and set-coalgebras can be found in [38].

2.3 STONE-BASED COALGEBRAIC LOGIC

The final section of this chapter is devoted to logic on Stone-coalgebras. Analogously to the set case, we define so-called *clopen* predicate liftings, a language, and models for interpreting this language. We then show that, provided Λ is a *characteristic* set of predicate liftings, the notions of modal equivalence and behavioural equivalence coincide. Thereafter, Λ -bisimulations for models on Stone coalgebras are introduced. The section closes with two examples of well-known Stone-coalgebras which occur in modal logic. **2.24 Definition.** Let \mathbb{T} be an endofunctor on Stone. A clopen predicate lifting of arity n is a natural transformation

$$\lambda: \mathbb{Clp}^n \to \mathbb{Clp} \circ \mathbb{T}.$$

A clopen predicate lifting is said to be **monotone** if for all topological spaces \mathfrak{X} and all $a_1, \ldots, a_n, \delta_1, \ldots, \delta_n \in \mathbb{Clp}(\mathfrak{X})$, if $a_i \subseteq \delta_i$ for all $1 \leq i \leq n$, then $\lambda_{\mathfrak{X}}(a_1, \ldots, a_n) \subseteq \lambda_{\mathfrak{X}}(\delta_1, \ldots, \delta_n)$. The **dual** of a clopen predicate lifting λ is given by $\lambda_{\mathfrak{X}}^{\partial}(a_1, \ldots, a_n) := \mathbb{T}\mathfrak{X} \setminus \lambda(X \setminus a_1, \ldots, X \setminus a_n)$. A collection Λ of predicate liftings for \mathbb{T} is said to be **characteristic** if for every Stone space \mathfrak{X} the collection

$$\{\lambda_{\mathbf{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda, a_i \in \mathbb{Clp}\,\mathbf{X}\}$$

forms a subbase for the topology on $\mathbb{T}\mathfrak{X}$.

The condition for a collection of predicate liftings to be characteristic can be regarded as the topological counterpart of being separated.

2.25 Remark. In [16] the authors define a *topological* predicate lifting as the Stone space variation of a predicate lifting. A **topological predicate lifting** for a Stone functor \mathbb{T} is a natural transformation

$$\lambda: \check{\mathbb{P}}^n \circ \mathbb{U} \to \check{\mathbb{P}} \circ \mathbb{U} \circ \mathbb{T}$$

such that for all Stone spaces \mathfrak{X} and $a_1, \ldots, a_n \in \mathbb{Clp} \mathfrak{X}$ the set $\lambda_{\mathfrak{X}}(a_1, \ldots, a_n)$ is clopen in $\mathbb{T}\mathfrak{X}$. Although $\lambda_{\mathfrak{X}}(a)$ is defined for all subsets $a \subseteq \mathfrak{X}$, the only information that is used in the semantics of the language is the action of $\lambda_{\mathfrak{X}}$ on the clopens of \mathfrak{X} .

If $\lambda : \mathbb{P} \circ \mathbb{U} \to \mathbb{P} \circ \mathbb{U} \circ \mathbb{T}$ is a (unary) topological predicate lifting then we can obtain a clopen predicate lifting λ^r by restricting for each Stone space \mathfrak{X} the map $\lambda_{\mathfrak{X}}$ to $\mathbb{Clp} \mathfrak{X}$. By definition of a topological predicate lifting we have $\lambda_{\mathfrak{X}}^r(a) \in \mathbb{Clp}(\mathbb{T}\mathfrak{X})$ for all $a \in \mathbb{Clp}\mathfrak{X}$, so $\lambda_{\mathfrak{X}}^r$ is indeed a map to $\mathbb{Clp}(\mathbb{T}\mathfrak{X})$. Naturality of λ^r follows immediately from the naturality of λ . For every Stone space \mathfrak{X} the action of $\lambda_{\mathfrak{X}}$ and $\lambda_{\mathfrak{X}}^r$ on clopens of \mathfrak{X} is the same.

The *n*-ary case is similar. So every topological predicate lifting yields a clopen predicate lifting which gives the same language and semantics.

We have not found a converse, i.e., a way to turn each open predicate lifting into a topological predicate lifting. Nor have we found a counterexample that this is not possible. We leave this as an interesting open question.

2.26 Definition. Let \mathbb{T} be a functor on Stone. A \mathbb{T} -model is a triple $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ where (\mathfrak{X}, γ) is a \mathbb{T} -coalgebra and $V : \Phi \to \mathbb{Clp} \mathfrak{X}$ is an admissible valuation of the proposition letters. A \mathbb{T} -model morphism from $(\mathfrak{X}, \gamma, V)$ to $(\mathfrak{X}', \gamma', V')$ is a \mathbb{T} -coalgebra morphism $f : (\mathfrak{X}, \gamma) \to (\mathfrak{X}', \gamma')$ such that $f^{-1} \circ V' = V$. The collection of \mathbb{T} -models and \mathbb{T} -model morphisms forms a category, called $\mathsf{Mod}(\mathbb{T})$.

An Aczel-Mendler bisimulation between two T-models is an Aczel-Mendler bisimulation between the underlying T-coalgebras such that for all $(x, x') \in B$ and $p \in \Phi$, $x \in V(p)$ iff $x' \in V'(p)$.

2.27 Definition. The **language** induced by a collection of clopen predicate liftings Λ is the set $\mathcal{L}(\Lambda)$ of formulas

$$\varphi ::= \bot \mid p \mid \neg \varphi \mid \varphi_1 \land \varphi_2 \mid \heartsuit^{\lambda}(\varphi_1, \dots, \varphi_n),$$

 \triangleleft

where $p \in \Phi$ and $\lambda \in \Lambda$ is *n*-ary. The symbols \top, \lor, \rightarrow and \leftrightarrow denote the usual abbreviations. The **semantics** of $\varphi \in \mathcal{L}(\Lambda)$ on a \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ is given inductively by

$$\llbracket p \rrbracket^{\mathfrak{X}} = V(p), \quad \llbracket \varphi_1 \wedge \varphi_2 \rrbracket^{\mathfrak{X}} = \llbracket \varphi_1 \rrbracket^{\mathfrak{X}} \cap \llbracket \varphi_2 \rrbracket^{\mathfrak{X}}, \quad \llbracket \neg \varphi \rrbracket^{\mathfrak{X}} = X \smallsetminus \llbracket \varphi \rrbracket^{\mathfrak{X}}, \\ \llbracket \heartsuit^{\lambda}(\varphi_1, \dots, \varphi_n) \rrbracket^{\mathfrak{X}} = \gamma^{-1}(\lambda(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}})).$$

A formula φ is called **valid** on $\mathsf{Mod}(\mathbb{T})$ if for every \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and all $x \in \mathfrak{X}$ we have $\mathfrak{X}, x \Vdash \varphi$. Denote the collection of valid formulas of $\mathcal{L}(\Lambda)$ by $\mathbf{Log}(\mathbb{T}, \Lambda)$.

Two states x and x' in two T-models \mathfrak{X} and \mathfrak{X}' are called **modally equivalent**, notation $x \equiv_{\Lambda} x'$, if for all $\varphi \in \mathcal{L}(\Lambda)$, $\mathfrak{X}, x \Vdash \varphi \Leftrightarrow \mathfrak{X}', x' \Vdash \varphi$.

2.28 Proposition. Let \mathbb{T} be an endofunctor on Stone, Λ a set of predicate liftings for \mathbb{T} and f a \mathbb{T} -model morphism from $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ to $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$. Then

$$\mathfrak{X}, x \Vdash \varphi \quad iff \quad \mathfrak{X}', f(x) \Vdash \varphi.$$

Proof. The proof of this lemma is similar to the proof of proposition 3.20, which is in turn similar to the proof of theorem 6.17 in [56]. \Box

The next theorem connects behavioural equivalence to modal equivalence. The proof is inspired by theorem 4.1 in [16].

2.29 Proposition. Let \mathbb{T} be an endofunctor on Stone and Λ a characteristic set of predicate liftings for \mathbb{T} . Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be two \mathbb{T} -models and $x \in \mathfrak{X}, x' \in \mathfrak{X}'$ states in these models. Then x and x' are modally equivalent if and only if they are behaviourally equivalent.

Proof. Our strategy to prove this proposition is to construct a final coalgebra of theories and then exploit that two states are behaviourally equivalent if and only if their theories are the same.

Let \mathfrak{Z} be the collection of maximal satisfiable sets of formulas of $\mathcal{L}(\Lambda)$, with a topology generated by the clopen subbase $\{\widetilde{\varphi} \mid \varphi \in \mathrm{CL}\}$, where $\widetilde{\varphi} = \{\Gamma \in \mathfrak{Z} \mid \varphi \in \Gamma\}$. By definition \mathfrak{Z} is (homeomorphic to) the dual Stone space of the Lindenbaum-Tarski algebra of $\mathrm{Log}(\mathbb{T}, \Lambda)$, so every clopen set is of the form $\widetilde{\varphi}$ for some $\varphi \in \mathcal{L}(\Lambda)$.

For every \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ define a map

$$\operatorname{th}_{\mathfrak{X}}: \mathfrak{X} \to \mathfrak{Z}: x \mapsto \{\varphi \mid \mathfrak{X}, x \Vdash \varphi\}.$$

2.29.A Claim. Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be two \mathbb{T} -models, $x \in \mathfrak{X}, x' \in \mathfrak{X}'$. If $\operatorname{th}_{\mathfrak{X}}(x) = \operatorname{th}_{\mathfrak{X}'}(x')$ then $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) = \mathbb{T} \operatorname{th}_{\mathfrak{X}'}(\gamma'(x'))$.

Proof of claim. Suppose $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) \neq \mathbb{T} \operatorname{th}_{\mathfrak{X}'}(\gamma'(x'))$, then there exists a clopen set $c \in \mathbb{Clp}(\mathbb{TZ})$ such that $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) \in c$ and $\mathbb{T} \operatorname{th}_{\mathfrak{X}'}(\gamma'(x')) \notin c$. Since Λ is characteristic for \mathbb{T} and every clopen set of \mathfrak{Z} is of the form $\widetilde{\varphi}$, there exist $\lambda \in \Lambda$ and $\widetilde{\varphi}_1, \ldots, \widetilde{\varphi}_n$ such that $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1, \ldots, \widetilde{\varphi}_n) \subseteq c$.

Observe th_{\mathfrak{X}} $(\widetilde{\varphi}_i) = \llbracket \varphi_i \rrbracket^{\mathfrak{X}}$ for $1 \leq i \leq n$. The fact that $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1, \ldots, \widetilde{\varphi}_n)$ and naturality of λ yield

$$\gamma(x) \in (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1}(\lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n})) = \lambda_{\mathfrak{X}}(\operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{1}), \dots, \operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{n})) = \lambda_{\mathfrak{X}}(\llbracket \varphi_{1} \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_{n} \rrbracket^{\mathfrak{X}})$$

and similarly $\gamma'(x') \notin \lambda_{\mathfrak{X}'}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \ldots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}})$. Therfore $\mathfrak{X}, x \Vdash \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n)$ and $\mathfrak{X}', x' \not\models \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n)$, so $\operatorname{th}_{\mathfrak{X}}(x) \neq \operatorname{th}_{\mathfrak{X}'}(x')$. This proves the claim.

Let $\Gamma \in \mathfrak{Z}$ and let (\mathfrak{X}, x) be a pointed \mathbb{T} -model such that $\operatorname{th}_{\mathfrak{X}}(x) = \Gamma$. Such a pointed model always exists because the elements of \mathfrak{Z} are assumed to be satisfiable. Define $\zeta(\Gamma) := \mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x))$. This gives rise to a map

$$\zeta: \mathfrak{Z} \to \mathbb{T}\mathfrak{Z}$$

which by the previous claim is well-defined, because it does not depend on the choice of the pointed model (\mathfrak{X}, x) . Moreover, we argue that ζ is continuous:

2.29.B Claim. The map $\zeta : \mathfrak{Z} \to \mathbb{T}\mathfrak{Z}$ is continuous.

Proof of claim. Since Λ is characteristic it suffices to show that $\zeta^{-1}(\lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n))$ is open in \mathfrak{Z} for $\lambda \in \Lambda$ and $\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n \in \operatorname{Clop} \mathfrak{Z}$. Fix such a λ and $\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n$. We will show that

$$\zeta^{-1}(\lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n))=\mathfrak{S}^{\lambda}(\varphi_1,\ldots,\varphi_n).$$

For $\Gamma \in \mathfrak{Z}$, let (\mathfrak{X}, x) be a pointed \mathbb{T} -model with $\operatorname{th}_{\mathfrak{X}}(x) = \Gamma$. Then

$$\zeta(\Gamma) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n}) \Leftrightarrow \mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n})$$

$$\Leftrightarrow \gamma(x) \in \lambda_{\mathfrak{X}}(\operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{1}), \dots, \operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{n}))$$

$$\Leftrightarrow \gamma(x) \in \lambda_{\mathfrak{X}}(\llbracket \varphi_{1} \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_{n} \rrbracket^{\mathfrak{X}})$$

$$\Leftrightarrow \mathfrak{X}, x \Vdash \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n})$$

$$\Leftrightarrow \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \in \operatorname{th}_{\mathfrak{X}}(x)$$

$$\Leftrightarrow \nabla^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \in \Gamma$$

$$\Leftrightarrow \Gamma \in \overline{\heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n})}.$$

This proves continuity of ζ .

We have established that (\mathfrak{Z}, ζ) is a T-coalgebra. Endow (\mathfrak{Z}, ζ) with the valuation $V_{\mathfrak{Z}} : \Phi \to \operatorname{Clop} \mathfrak{Z} : p \mapsto \widetilde{p}$. By construction each map $\operatorname{th}_{\mathfrak{X}}$ is a T-model morphism. It then follows that $\mathfrak{Z}, \Gamma \Vdash \varphi$ iff $\varphi \in \Gamma$: the case $\varphi = p$ holds by definition of $V_{\mathfrak{Z}}$, the Boolean cases follow by an easy induction, and the case $\varphi = \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n)$ follows from the proof of the previous claim. In addition, \mathfrak{Z} is final.

2.29.C Claim. The \mathbb{T} -model $\mathfrak{Z} = (\mathfrak{Z}, \zeta, V_{\mathfrak{Z}})$ is final in $\mathsf{Mod}(\mathbb{T})$.

Proof of claim. Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ be any \mathbb{T} -model and $f : \mathfrak{X} \to \mathfrak{Z}$ a \mathbb{T} -model morphism. It follows from proposition 2.28 that for all $x \in \mathfrak{X}$ we have $\mathfrak{X}, x \Vdash \varphi$ iff $\mathfrak{Z}, f(x) \Vdash \varphi$ iff $\varphi \in f(x)$, so $f(x) = \operatorname{th}_{\mathfrak{X}}(x)$ hence $f = \operatorname{th}_{\mathfrak{X}}$.

The proposition now follows: suppose x and x' are modally equivalent, then $\operatorname{th}_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Z}$ and $\operatorname{th}_{\mathfrak{X}'} : \mathfrak{X}' \to \mathfrak{Z}'$ are \mathbb{T} -model morphisms such that $\operatorname{th}_{\mathfrak{X}}(x) = \operatorname{th}_{\mathfrak{X}'}(x')$, so x and x' are behaviourally equivalent. Conversely, if x and x' are behaviourally equivalent, then we must have $\operatorname{th}_{\mathfrak{X}}(x) = \operatorname{th}_{\mathfrak{X}'}(x')$ so by proposition 2.28 $x \equiv_{\Lambda} x'$.

We will now define a notion of bisimulation between models and relate this to modal equivalence and behavioural equivalence. The following definition of Λ -bisimulation is an adaptation of ideas in [6, 17].

 \diamond

2.30 Definition. Let \mathbb{T} be an endofunctor on Stone and Λ a collection of predicate liftings for \mathbb{T} . Let (\mathfrak{X}, γ) and (\mathfrak{X}', γ') be \mathbb{T} -coalgebras and $B \subseteq \mathfrak{X} \times \mathfrak{X}'$ be a subspace with projections $\pi : B \to \mathfrak{X}$ and $\pi' : B \to \mathfrak{X}'$. Let

be the pullback diagram of the cospan $(\mathfrak{clp}(\pi), \mathfrak{clp}(\pi'))$ in BA. We say that B is a A-bisimulation if for all $(x, x') \in B$ and $\lambda \in \Lambda$ we have

$$\mathfrak{clp}(\pi) \circ \mathfrak{clp}(\gamma) \circ \lambda_{\mathfrak{X}} \circ \overline{\pi}^n = \mathfrak{clp}(\pi') \circ \mathfrak{clp}(\gamma') \circ \lambda_{\mathfrak{X}'} \circ \overline{\pi}'^n.$$

A relation *B* between two T-models $(\mathfrak{X}, \gamma, V)$ and $(\mathfrak{X}', \gamma', V')$ is a Λ -bisimulation if it is a Λ -bisimulation between the underlying T-coalgebras and for all $(x, x') \in B$ and $p \in \Phi$ we have $x \in V(p)$ iff $x' \in V'(p)$. Two states $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are called Λ -bisimilar, notation $x \cong_{\Lambda} x'$, if there is a Λ -bisimulation linking them.

2.31 Remark. Observe that $(a, a') \in \mathbb{Clp} \mathfrak{X} \times \mathbb{Clp} \mathfrak{X}'$ is *B*-coherent, i.e. $B[a] \subseteq a'$ and $B^{-1}[a'] \subseteq a$, if and only if it is in $pb(\mathfrak{clp}(\pi), \mathfrak{clp}(\pi'))$. It follows from unraveling that *B* is a Λ -bisimulation if and only if for all $\lambda \in \Lambda$ and all *B*-coherent pairs of clopens $(a_i, a'_i) \in \mathbb{Clp} \mathfrak{X} \times \mathbb{Clp} \mathfrak{X}'$ we have

$$\gamma(x) \in \lambda_{\mathfrak{X}}(a_1, \dots, a_n) \quad \text{iff} \quad \gamma'(x') \in \lambda_{\mathfrak{X}'}(a_1', \dots, a_n').$$

The following statements are easy to verify.

2.32 Lemma. Let \mathbb{T} be an endofunctor on Stone, Λ a set of predicate liftings for \mathbb{T} and $(\mathfrak{X}, \gamma, V)$ and $(\mathfrak{X}', \gamma', V')$ \mathbb{T} -models. If two states $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are Λ -bisimilar, then they are modally equivalent.

2.33 Proposition. Let \mathbb{T} be an endofunctor on Stone and Λ set of predicate liftings for \mathbb{T} . Every Aczel-Mendler bisimulation between \mathbb{T} -models is a Λ -bisimulation.

If Λ is characteristic, it follows from the previous lemma and proposition combined with proposition 2.29 that Aczel-Mendler bisimilarity implies behavioural equivalence. If moreover \mathbb{T} preserves weak pullbacks, the converse holds as well. The proof of this is similar to theorem 4.3 and the preceding discussion in [49].

However, we do not wish to make this assumption. For example, the Vietoris functor does not preserve weak pullbacks [9, Corollary 4.3]. The next proposition shows that for monotone Λ , behavioural equivalence implies Λ -bisimilarity, without assuming \mathbb{T} to preserve weak pullbacks.

2.34 Proposition. Let Λ be a monotone characteristic set of predicate liftings for a functor \mathbb{T} and suppose two states x and x' in \mathbb{T} -models $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ are behaviourally equivalent. Then x and x' are Λ -bisimilar.

Proof. Since x and x' are behaviourally equivalent, there must be some \mathbb{T} -coalgebra (\mathfrak{Y}, δ) and some coalgebra morphisms $f : \mathfrak{X} \to \mathfrak{Y}, f' : \mathfrak{X}' \to \mathfrak{Y}$ such that f(x) = f'(x'). Let

$$B = \{(u, u') \in \mathbf{X} \times \mathbf{X}' \mid f(u) = f'(u')\},\$$

then clearly xBx'. We claim that B is a Λ -bisimulation.

In order to show that B is a Stone space, it suffices to show that B is closed. To see this, suppose $(u, u') \notin B$. Then $f(u) \neq f'(u')$ and since \mathcal{Y} is Hausdorff there exist disjoint clopens $a, a' \in \mathbb{Clp} \mathcal{Y}$ that contain f(u) and f'(u') respectively. Now $f^{-1}(a) \times (f')^{-1}(a')$ contains (u, u'), is open in $\mathfrak{X} \times \mathfrak{X}'$ and is disjoint from B. Therefore B is closed in $\mathfrak{X} \times \mathfrak{X}'$. It follows from proposition 2.29 that for all $(x, x') \in B$ we have $x \in V(p)$ iff $x' \in V'(p)$.

Let $\lambda \in \Lambda$ be *n*-ary and for $1 \leq i \leq n$ let (a_i, a'_i) be a *B*-coherent pair of clopens. Suppose uBu' and $\gamma(u) \in \lambda_{\mathfrak{X}}(a_1, \ldots, a_n)$. We will show that $\gamma'(u') \in \lambda_{\mathfrak{X}'}(a'_1, \ldots, a'_n)$, the converse direction is similar.

Let us construct for each pair (a_i, a'_i) a clopen set $\mathcal{B}_i \in \mathbb{Clp} \mathcal{Y}$ such that $f[a_i] \subseteq \mathcal{B}_i$ and $(f')^{-1}(\mathcal{B}_i) \subseteq a'_i$. Since a_i is clopen, $f[a_i]$ is closed in \mathcal{Y} , so we may write $f[a_i] = \bigcap \{c \in \mathbb{Clp} \mathcal{Y} \mid f[a_i] \subseteq c\}$. Because continuous maps preserve arbitrary meets, we have

$$\bigcap\{(f')^{-1}(c) \mid f[a_i] \subseteq c \in \mathbb{Clp} \, \mathcal{Y}\} = (f')^{-1}(f[a_i]) \subseteq a'.$$

The collection $\{\mathbf{X}' \smallsetminus (f')^{-1}(c) \mid f[a_i] \subseteq c \in \mathbb{Clp} \, \mathcal{Y}\}$ is an open cover of the (closed hence) compact set $\mathbf{X}' \smallsetminus a'$, so there exists a finite number $c_1, \ldots, c_m \in \mathbb{Clp} \, \mathcal{Y}$ such that $\bigcup_{j=1}^m \mathbf{X}' \smallsetminus (f')^{-1}(c_j)$ covers $\mathbf{X}' \smallsetminus a'$. Set $\mathcal{B}_i = c_1 \cap \cdots \cap c_m$, then $\mathcal{B}_i \in \mathbb{Clp} \, \mathcal{Y}$ and $(f')^{-1}(\mathcal{B}_i) \subseteq a_i$. Moreover $f[a_i] \subseteq \mathcal{B}_i$, hence $a_i \subseteq f^{-1}(\mathcal{B}_i)$.

By monotonicity and naturality of λ we find

$$\gamma(u) \in \lambda_{\mathfrak{X}}(a_1, \ldots, a_n) \subseteq \lambda_{\mathfrak{X}}(f^{-1}(\mathfrak{G}_1), \ldots, f^{-1}(\mathfrak{G}_n)) = (\mathbb{T}f)^{-1}(\lambda_{\mathfrak{Y}}(\mathfrak{G}_1, \ldots, \mathfrak{G}_n)),$$

so $(\mathbb{T}f)(\gamma(u)) \in \lambda_{\mathcal{Y}}(\beta_1, \dots, \beta_n)$. Since f and f' are coalgebra morphisms we have $(\mathbb{T}f)(\gamma(u)) = \delta(f(u)) = \delta(f'(u')) = (\mathbb{T}f')(\gamma'(u'))$ and by monotonicity and naturality of λ again we find

$$\gamma'(u') \in (\mathbb{T}f')^{-1}(\lambda_{\mathfrak{Y}}(\mathfrak{b}_1,\ldots,\mathfrak{b}_n)) = \lambda_{\mathfrak{X}'}((f')^{-1}(\mathfrak{b}_1),\ldots,(f')^{-1}(\mathfrak{b}_n)) \subseteq \lambda_{\mathfrak{X}'}(a'_1,\ldots,a'_n).$$

This proves the proposition.

2.35 Example (Descriptive frames). As a first example of logic on a Stone-coalgebra, we mention descriptive frames for modal logic [37]. Descriptive frames turn out to be coalgebras for the Vietoris functor:

For a topological space \mathfrak{X} let $\mathbb{V}\mathfrak{X}$ be the set of closed subsets of \mathfrak{X} topologised by the subbase

$$\boxplus a := \{ b \in \mathbb{V} \mathfrak{X} \mid b \subseteq a \}, \qquad \boxplus a := \{ b \in \mathbb{V} \mathfrak{X} \mid a \cap b \neq \emptyset \},$$

where a ranges over the opens of \mathfrak{X} . This assignment can be extended to a functor on Top by defining $\mathbb{V}f:\mathbb{V}\mathfrak{X}\to\mathbb{V}\mathfrak{X}'$ to be the direct image of f, for continuous functions $f:\mathfrak{X}\to\mathfrak{X}'$. It is well known that the Vietoris functor restricts to KTop, KHaus and Stone, and that the category of descriptive frames and its morphisms is isomorphic to Coalg($\mathbb{V}_{\text{Stone}}$) [37]. For a thorough survey of properties of the Vietoris functor, see [58].

Descriptive monotone frames are another important example of Stone-coalgebras. Below we give a way to view these as Stone coalgebras which is slightly different from, but equivalent to [16, 21]. The example will also play a role in the next chapter.

2.36 Definition ([20], Definition 7.30). A general monotone frame is a triple (X, μ, A) where (X, μ) is a monotone frame and $A \subseteq \mathbb{P}X$ is a collection of admissible subsets of

X which contains \emptyset and X and is closed under finite intersection, finite union, taking complements and the map

$$m_{\mu}: \mathbb{P}X \to \mathbb{P}X: a \mapsto \{x \in X \mid a \in \mu(x)\}.$$

A general monotone frame morphism from (X, μ, A) to (X', μ', A') is a bounded morphism $f : (X, \mu) \to (X', \mu')$ between the underlying monotone frames such that $f^{-1}(a') \in A$ for all $a' \in A'$.

Let \mathfrak{X} denote the topological space with underlying set X topologised by the clopen subbase A. A general monotone frame is called **differentiated** if $x \in a \Leftrightarrow x' \in a$ for all $a \in A$ implies x = x'. It is called **tight** if for all $x \in X$, $c \in K\mathfrak{X}$ and $u \subseteq X$ we have

- $c \in \nu(x)$ iff every admissible superset $a \supseteq c$ is in $\nu(x)$; and
- $u \in \nu(x)$ iff there exists a closed subset $c \subseteq u$ that is in $\nu(x)$.

A general monotone frame is called **compact** if A is compact. A **descriptive monotone** frame is differentiated, tight and compact general monotone frame.

The following definition is taken from [16] and is equivalent to definition 3.9 in [21].

2.37 Definition. For a Stone space $\mathfrak{X} = (X, \tau)$ define $\mathbb{D}'_{st}\mathfrak{X}$ to be the collection of sets $W \subseteq \mathbb{P}X$ such that $a \in W$ iff there exists a closed $c \subseteq a$ such that every clopen superset of c is in W. Endow $\mathbb{D}'_{st}\mathfrak{X}$ with the topology generated by the clopen subbase

$$\boxplus a \coloneqq \{ W \in \mathbb{D}'_{\mathsf{st}} \mathfrak{X} \mid a \in W \}, \quad \& a \coloneqq \{ W \in \mathbb{D}'_{\mathsf{st}} \mathfrak{X} \mid Xa \notin W \},\$$

where a ranges over $\mathbb{Clp} \mathfrak{X}$.

For continuous functions $f: \mathfrak{X} \to \mathfrak{X}'$ define

$$\mathbb{D}'_{\mathsf{st}}f:\mathbb{D}'_{\mathsf{st}}\mathfrak{X}\to\mathbb{D}'_{\mathsf{st}}\mathfrak{X}':W\mapsto\{a\in\mathbb{P}X\mid f^{-1}(a)\in W\}.$$

Descriptive monotone frames are known to be coalgebras for \mathbb{D}'_{st} . In fact, the category of descriptive conditional frames and general monotone frame morphisms, DMF, is isomorphic to the category of \mathbb{D}'_{st} -coalgebras and \mathbb{D}' -coalgebra morphisms [21],

DMF
$$\cong$$
 Coalg(\mathbb{D}'_{st}).

The functor \mathbb{D}_{st} in the next definition arises from definition 2.37 by replacing the use of clopen sets by open sets. This functor will turn out to be equivalent to \mathbb{D}'_{st} , but allows for a generalisation to the category of compact Hausdorff spaces in section 3.3.

2.38 Definition. Let $\mathfrak{X} = (X, \tau)$ be a Stone space. Let $\mathbb{D}_{st}\mathfrak{X}$ be the collection of sets $W \subseteq \mathbb{P}X$ such that $a \in W$ iff there exists a closed $c \subseteq u$ such that every open superset of c is in W. Endow $\mathbb{D}_{st}\mathfrak{X}$ with the topology generated by the subbase

$$\boxplus a := \{ W \in \mathbb{D}_{st} \mathfrak{X} \mid a \in W \}, \quad \& a := \{ W \in \mathbb{D}_{st} \mathfrak{X} \mid X \setminus a \notin W \},\$$

where a ranges over $\Omega \mathfrak{X}$. For continuous functions $f : \mathfrak{X} \to \mathfrak{X}'$ define $\mathbb{D}_{st}f : \mathbb{D}_{st}\mathfrak{X} \to \mathbb{D}_{st}\mathfrak{X}' : W \mapsto \{a' \in \mathbb{P}X \mid f^{-1}(a') \in W\}.$

2.39 Lemma. If $f : \mathfrak{X} \to \mathfrak{X}'$ is a morphism in Stone, then $\mathbb{D}_{st}f$ is a well-defined continuous function from $\mathbb{D}_{st}\mathfrak{X}$ to $\mathbb{D}_{st}\mathfrak{X}'$.

Proof. $\mathbb{D}_{st}f$ is well-defined. Let $W \in \mathbb{D}_{st}\mathfrak{X}$. We need to show that $\mathbb{D}_{st}f(W) \in \mathbb{D}_{st}\mathfrak{X}'$. Suppose $a' \in \mathbb{D}_{st}f(W)$. Then $f^{-1}[a'] \in W$ and so there exists a closed $c \subseteq f^{-1}[a']$ such that $c \in W$. Since \mathfrak{X} is compact and \mathfrak{X}' is Hausdorff, f[c] is a closed set in \mathfrak{X}' . Besides $f[c] \subseteq a'$. Suppose $f[c] \subseteq b$ for some open $b \in \Omega\mathfrak{X}'$, then $c \subseteq f^{-1}[b]$ so $f^{-1}[b] \in W$ and hence $b \in \mathbb{D}_{st}f(W)$. So all open supersets of f[c] are in $\mathbb{D}_{st}f(W)$, and therefore $f[c] \in \mathbb{D}_{st}f(W)$.

 $\mathbb{D}_{\mathsf{st}}f \text{ is continuous.} \quad \text{For continuity we need to show that both } (\mathbb{D}_{\mathsf{st}}f)^{-1}[\boxplus a'] \text{ and } (\mathbb{D}_{\mathsf{st}}f)^{-1}[\otimes a'] \text{ are open in } \mathbb{D}_{\mathsf{st}}\mathfrak{X}, \text{ whenever } a' \in \Omega(\mathfrak{X}'). \text{ It follows for a straightforward computation that } (\mathbb{D}_{\mathsf{st}}f)^{-1}(\boxplus a') = \boxplus f^{-1}(a'), \text{ which is open in } \mathbb{D}_{\mathsf{st}}\mathfrak{X} \text{ by definition. In a similar way we find } (\mathbb{D}_{\mathsf{st}}f)^{-1}(\otimes a') = \otimes f^{-1}[a'] \in \Omega\mathbb{D}_{\mathsf{st}}\mathfrak{X}. \qquad \Box$

Note that the first part of the previous lemma makes use of the fact that \mathfrak{X} is compact and \mathfrak{X}' is Hausdorff.

For an element $W \in \mathbb{D}_{st} \mathfrak{X}$ its upward closure is defined by $\uparrow(W) \coloneqq \{u \subseteq X \mid \exists u' \in W \text{ s.t. } u' \subseteq u\}$. The following lemma gives a more intuitive characterisation of the action of \mathbb{D}_{st} on morphisms. The proof is straightforward.

2.40 Lemma. Let $f : \mathfrak{X} \to \mathfrak{X}'$ be a continuous map between compact Hausdorff spaces and suppose $W \in \mathbb{D}_{st}\mathfrak{X}$. Then

$$\mathbb{D}_{\mathsf{st}}f(W) = \uparrow(\{f[u] \mid u \in W\}).$$

Finally, let us show that definition 2.38 is equivalent to definition 2.37 when restricted to Stone. It will follow as a corollary that $\mathbb{D}_{st} \mathfrak{X}$ is a Stone space whenever \mathfrak{X} is a Stone space.

2.41 Theorem. Let $\mathfrak{X} = (X, \tau)$ be a Stone space. Then $\mathbb{D}_{st}\mathfrak{X} \cong \mathbb{D}'_{st}\mathfrak{X}$.

Proof. We first show that the sets underlying both topological spaces are the same. It is obvious that $\mathbb{D}_{st}\mathfrak{X} \subseteq \mathbb{D}'_{st}\mathfrak{X}$. Conversely, take $W \in \mathbb{D}'_{st}\mathfrak{X}$. To show that $W \in \mathbb{D}_{st}\mathfrak{X}$ take an arbitrary $a \in W$. By definition of $\mathbb{D}'_{st}\mathfrak{X}$ there exists a closed set $\hbar \subseteq a$ such that all clopen supersets of \hbar are in W. Let δ be any open superset of \hbar . Since the clopen sets form a basis for \mathfrak{X} , for each $x \in a$ we can find a clopen c_x such that $x \in c_x \subseteq \delta$. The set \hbar is covered by a finite amount of such sets because it is closed and \mathfrak{X} is compact. Therefore there is a clopen set c such that $\hbar \subseteq c \subseteq \delta$. By assumption we have $c \in W$, hence $\delta \in W$. This shows that for all $a \in W$ there is a closed subset \hbar of a such that every open superset of \hbar is in W, so $W \in \mathbb{D}_{st}\mathfrak{X}$.

Next let us compare the topologies. It follows immediately form the definitions that $\Omega \mathbb{D}'_{st} \mathfrak{X} \subseteq \Omega \mathbb{D}_{st} \mathfrak{X}$. For the converse, it suffices to show that $\boxplus a, \otimes a \in \Omega \mathbb{D}'_{st} \mathfrak{X}$ for $a \in \Omega \mathfrak{X}$. Suppose $W \in \boxplus a$. Then $a \in W$ hence there is a closed $\hbar \subseteq a$ such that all open supersets of \hbar are in W. Since \mathfrak{X} is a Stone space there exists a clopen c such that $\hbar \subseteq c \subseteq a$. By assumption $c \in W$, so $W \in \boxplus c$. Since $\boxplus c \subseteq \boxplus a$, this proves that every element in $\boxplus a$ has an open neighbourhood in $\Omega \mathbb{D}'_{st} \mathfrak{X}$ contained in $\boxplus a$, hence $\boxplus a \in \Omega \mathbb{D}'_{st} \mathfrak{X}$. The case of \otimes can be treated similarly. \square

2.42 Corollary. The functor \mathbb{D}_{st} is an endofunctor on Stone.

Another guiding example of logic on Stone-coalgebras is that of descriptive conditional frames, which will be developed in chapter 5.

Coalgebraic geometric logic

In this chapter we investigate how one can extend geometric logic (i.e. logic with finite conjunctions and infinite disjunctions) with extra modalities. Some advances have been made in this field: Johnstone [29] defines a point-free, syntactic version of the Vietoris functor, using an extension of geometric logic with two unary operators, \Box and \diamond . Furthermore, in [57] the authors define the so-called Vietoris powerlocale functor $V_{\mathbb{T}}$: Frm \rightarrow Frm for a given set functor \mathbb{T} which satisfies some categorical properties, and take steps towards developing a logic with finite conjunctions, infinite disjunctions and a single modality.

Whereas the authors of [57] use the method of relation lifting to define the new modality, we use a modified form of predicate liftings. Besides, where they take an algebraic point of view, we adopt a topological approach. A category of (certain) topological spaces will form the base category of the coalgebras that we use, and the open sets serve as the interpretants of proposition letters.

In the Stone case, there is a dual equivalence between Stone and BA; the clopen sets in a Stone space, which are the interpretants of the proposition letters, form a Boolean algebra. This allows one to take both a topological and an algebraic point of view, i.e., every endofunctor on Stone gives rise, via this duality, to an endofunctor on BA and vice versa. In the new setting for coalgebraic geometric logic a similar duality is desirable. The open sets of a topological space also form an algebraic structure: a frame. In order to have a dual equivalence between topological spaces and frames, we have to restrict both categories (the category of topological spaces and continuous maps and the category of frames and frame homomorphisms) to suitable full subcategories. We will see in section 3.1 that there are several possibilities for this restriction. It is not a priori clear which of these is the right one. Throughout the chapter we will encounter pros and cons of each of these possibilities.

This chapter is structured as follows: In section 3.1 we lay the foundations for this chapter by investigating geometric logic and dualities. We find three candidates for the base category of coalgebraic geometric logic: the (full) subcategories of **Top** whose objects are sober spaces, compact sober spaces and compact Hausdorff spaces respectively. In the subsequent sections we develop coalgebraic geometric logic (section 3.2), examine two examples (section 3.3) and investigate bisimulations between the models for coalgebraic logic (section 3.4). The choice of base category will be continually remarked upon; where possible we will give definitions and results for all choices and whenever this is not possible we will indicate the problem.

3.1 GEOMETRIC LOGIC AND DUALITY

Before defining geometric logic, we recall some definitions concerning frames.

3.1 Definition. A frame is a complete lattice F in which for all $a \in F$ and $S \subseteq F$ the infinite distributive law holds:

$$a \land \bigvee S = \bigvee \{a \land s \mid s \in S\}.$$

A frame homomorphism is a function between frames that preserves finite meets and arbitrary joins. \triangleleft

3.2 Definition. A **presentation** is a pair $\langle G, R \rangle$ where G is a set of generators and R is a collection of relations between expressions constructed from the generators using arbitrary joints and finite meets.

Let F be a frame. Recall that $\mathbb{Z}F$ is the underlying frame. We say that $\langle G, R \rangle$ **presents** F if there is an assignment $f : G \to \mathbb{Z}F$ of the generators such that (i), (ii) and (iii) hold:

(i) The set $\{f(g) \mid g \in G\}$ generates F.

The assignment f can be extended to an assignment \tilde{f} for any expression x build from the generators in G using \wedge and \vee . We require

- (ii) If x = x' is a relation in R, then $\tilde{f}(x) = \tilde{f}(x')$ in F.
- (iii) For any F' and assignment $f': G \to \mathbb{Z}F'$ satisfying property (ii) there exists a frame homomorphism $h: F \to F'$ such that the diagram

$$\begin{array}{ccc} G & \stackrel{f}{\longrightarrow} & \mathbb{Z}F \\ & & & & \downarrow_{\mathbb{Z}g} \\ & & & & \mathbb{Z}F' \end{array}$$

commutes.

The frame homomorphism from (iii) is necessarily unique, because the image of the generating set $\{f(g) \mid g \in G\}$ under h is determined by the diagram. A detailed account of frame presentations may be found in chapter 4 of [59].

3.3 Remark. We will regularly want to define a frame homomorphism $F \to F'$ from a frame F presented by $\langle G, R \rangle$ to some frame F'. By definition 3.2 it suffices to give an assignment $f': G \to F'$ such that (ii) holds, because this yields a unique frame homomorphism $F \to F'$. By abuse of notation, we will denote the unique frame homomorphism $F \to F'$ such that the diagram in (iii) commutes with f' as well.

The next propositions allows us to define a frame by generators and relations. A proof can be found in [29, Proposition II2.11].

3.4 Proposition. Any presentation by generators and relations presents a frame.

3.5 Definition. A set *B* of elements in a frame *F* is called **directed** if for all $a, b \in B$ there is a $c \in B$ such that $a \leq c$ and $b \leq c$. We denote the disjunction $\forall B$ of a directed set *B* by $\forall B$, that is, the symbol \forall indicates that the set *B* is a directed set.

A collection $B \subseteq \mathbb{P}X$ of subsets of a set X is called directed if for all $a, b \in B$ there is a $c \in B$ such that $a \subseteq c$ and $b \subseteq c$. We write $\bigcup B$ for the union $\bigcup B$ of such a directed set B.

 \triangleleft

The opens of a topological space with the inclusion order provide an example of a frame. In this case the meet and join are simply set-theoretic intersection and union. Indeed, a finite meet of open sets is again open, as is an arbitrary union of opens. In fact, there is a contravariant functor $\mathsf{Top} \to \mathsf{Frm}$ sending a topological space to its frame of open sets.

3.6 Definition. Let \mathfrak{X} be a topological space. Define $\mathfrak{opn} \mathfrak{X}$ to be the frame of open sets of \mathfrak{X} (that is, the collection of open sets ordered by inclusion; it is routine to check that this is indeed a lattice). For a continuous function $f : \mathfrak{X} \to \mathfrak{X}'$ let $\mathfrak{opn} f = f^{-1} : \mathfrak{opn} \mathfrak{X}' \to \mathfrak{opn} \mathfrak{X}$. The map $\mathfrak{opn} : \mathsf{Top} \to \mathsf{Frm}$ is a contravariant functor.

A frame isomorphic to $\mathfrak{opn} \mathfrak{X}$ for some topological space \mathfrak{X} is called **spatial**.

An equivalent definition of spatiality is given in [29, II1.5]. The following definition is stated for future reference.

3.7 Definition. Let *F* be a frame. A filter in *F* is a nonempty upwards closed set *J* such that $a, b \in J$ implies $a \wedge b \in J$. A filter is called **prime** if $a \vee b \in J$ implies $a \in J$ or $b \in J$. A **completely prime filter** is a filter such that for all $S \subseteq A$, $\forall S \in J$ implies there is $a \in S$ with $a \in J$.

For $a, b \in F$ we say that a is **well inside** b, notation: $a \leq b$, if there is a $c \in F$ such that $c \wedge a = \bot$ and $c \vee b = \top$. An element $a \in F$ is called **regular** if $a = \bigvee \{b \in F \mid b \leq a\}$ and a frame is called **regular** if all of its elements are regular. The **negation** of $a \in F$ is defined as $\sim a = \bigvee \{b \in F \mid a \wedge b = \bot\}$.

A frame F is **compact** if for all directed sets $S, \forall S = \top$ implies $\top \in S$.

3.8 Lemma. For all elements a, b in a frame F we have $a \leq b$ iff $\neg a \lor b = \top$.

Proof. See III1.1 in [29].

3.9 Lemma. Finite meets and arbitrary joins of regular elements are regular.

Proof. It is known that $d \le c \le a \le b$ implies $d \le b$. We first show that $c \le a$ and $d \le b$ implies $c \land d \le a \land b$. It is clear that $c \land d \le a$ and $c \land d \le b$. Since $\sim (c \land d) \lor (a \land b) = (\sim (c \land d) \lor a) \land (\sim (c \land d) \lor b) = \top \land \top = \top$ we know $c \land d \le a \land b$.

Now suppose a and b are regular elements, then

$$a \wedge b = \bigvee \{c \mid c \leqslant a\} \wedge \bigvee \{d \mid d \leqslant b\} = \bigvee \{c \wedge d \mid c \leqslant a, d \leqslant b\} \leq \bigvee \{c \mid c \leqslant a \wedge b\} \leq a \wedge b,$$

so $a \wedge b$ is regular. If a_i is regular for all *i* in some index set *I*, then

$$\bigvee_{i \in I} a_i = \bigvee_{i \in I} \left(\bigvee \{ c \mid c \leqslant a_i \} \right) \le \bigvee \left\{ c \mid c \leqslant \bigvee_{i \in I} a_i \right\} \le \bigvee_{i \in I} a_i,$$

so an arbitrary join of regular elements is regular.

Now let us proceed to geometric logic. As stated in the introduction, geometric logic can be viewed as the logic of finitely observable statements. A finitely observable statement is a statement which can be verified in a finite amount of time. For example, the statement

"There exist glow-in-the-dark turtles."

 \triangleleft

To verify this statement, it suffices to find a single glow-in-the-dark turtle. Therefore the statement is finitely observable. On the other hand, to refute the statement, one would have to find all turtles in the world and check that they do not glow in the dark. To be complete, one should also check all past and future turtles. In a practical sense, this statement can never be refuted. Thus, the statement

"Glow-in-the-dark turtles do not exist"

is not finitely observable.¹

The previous discussion shows that finitely observable statements are not closed under taking negations. The reader can easily convince himself that the collection of finitely observable statements is not closed under implications either. However, finitely observable statements are closed under taking arbitrary disjunctions and finite conjunctions. This intuition leads to the following definition of geometric logic.

3.10 Definition. Let Φ be a set of proposition letters. The geometric formulae over Φ are given by

$$\varphi ::= \mathsf{T} \mid p \mid \varphi_1 \land \varphi_2 \mid \bigvee_{i \in I} \varphi_i,$$

where $p \in \Phi$. We abbreviate $\perp = \bigvee \emptyset$. Write \mathcal{GL} for the collection of geometric formulas.

A sequent is a pair of \mathcal{GL} -formulas. We write $\varphi \vdash \psi$ if (φ, ψ) is a sequent. Intuitively, this should be thought of as " φ implies ψ ". A geometric theory over Φ is a collection of sequents that contains the axioms $\varphi \vdash \varphi$ and is closed under the following rules: cut

$$\frac{\varphi \vdash \psi \quad \psi \vdash \chi}{\varphi \vdash \chi},$$

the conjuction rules

$$\varphi \vdash \mathsf{T}, \quad \varphi \land \psi \vdash \varphi, \quad \varphi \land \psi \vdash \psi, \quad \frac{\varphi \vdash \psi \quad \varphi \vdash \chi}{\varphi \vdash \psi \land \chi}$$

the disjunction rules

$$\varphi \vdash \bigvee S \quad (\varphi \in S), \quad \frac{\varphi \vdash \psi \text{ (for all } \varphi \in S)}{\bigvee S \vdash \psi}$$

and frame distributivity

$$\varphi \land \bigvee S \vdash \bigvee \{\varphi \land \psi \mid \psi \in S\}.$$

Let T be a theory. If T contains $\varphi \vdash \psi$ and $\psi \vdash \varphi$ we say that φ and ψ are **equivalent** with respect to T. We call φ and ψ **equivalent** if they are equivalent with respect to every theory.

3.11 Remark. The collection \mathcal{GL} is not generally a set; it may be a proper class. However, frame distributivity allows us to reduce every formula to an equivalent disjunction of finite conjunctions of symbols in Φ . Therefore the collection of formulas modulo equivalence forms a set. Let T be a geometric theory. The rules imply that the Lindenbaum-Tarski algebra, i.e. the set of geometric formulas modulo equivalence with respect to T of a theory is a frame [62]. Accordingly, we shall call it the Lindenbaum-Tarski frame.

¹The existence of glow in the dark turtles has never been refuted. In fact, they have been observed recently. See [19] for a scientific article and [41] for a video.

For more information about the connection between geometric logic and frames we refer to [62]. Topological spaces with a valuation form models for geometric logic.

3.12 Definition. A valuation of a topological space \mathfrak{X} is a map $V : \Phi \to \Omega \mathfrak{X}$. One may define truth of $G\mathcal{L}$ -formulas in $\mathfrak{X} = (\mathfrak{X}, V)$ inductively by

$$\llbracket \mathsf{T} \rrbracket^{\mathfrak{X}} = \mathbb{U}\mathfrak{X}, \quad \llbracket p \rrbracket^{\mathfrak{X}} = V(p), \quad \llbracket \varphi_1 \land \varphi_2 \rrbracket^{\mathfrak{X}} = \llbracket \varphi_1 \rrbracket^{\mathfrak{X}} \cap \llbracket \varphi_2 \rrbracket^{\mathfrak{X}}, \llbracket \bigvee_{i \in I} \varphi_i \rrbracket^{\mathfrak{X}} = \bigcup_{i \in I} \llbracket \varphi_i \rrbracket^{\mathfrak{X}}.$$

 \triangleleft

We write $\mathfrak{X}, x \Vdash \varphi$ iff $x \in \llbracket \varphi \rrbracket^{\mathfrak{X}}$.

In definition 3.6 we have seen the functor \mathfrak{opn} : Top \rightarrow Frm. We will now define a functor in the opposite direction that is right adjoint to \mathfrak{opn} .

3.13 Definition. A point of a frame F is a frame homomorphism $p: F \to 2$, with $2 = \{T, \bot\}$ the two-element frame. Let $\mathfrak{pt}F$ be the collection of points of F endowed with the topology $\{\tilde{a} \mid a \in F\}$, where $\tilde{a} = \{p \in \mathfrak{pt}F \mid p(a) = \top\}$. For a frame homomorphism $f: F \to F'$ define $\mathfrak{pt}f: \mathfrak{pt}F' \to \mathfrak{pt}F$ by $p \mapsto p \circ f$. The assignment \mathfrak{pt} defines a functor Frm \to Top.

A topological space that arises as the space of points of a lattice is called **sober**. The **sobrification** of a topological space \mathfrak{X} is $\mathfrak{pt}(\mathfrak{opn}\mathfrak{X})$.

There is a 1-1 correspondence between the points of a frame and the completely prime filters of the frame: For a completely prime filter F the map $p_F: A \to 2$ defined by $p_F(a) = \top$ iff $a \in F$ and $p_F(a) = \bot$ if $a \notin F$ is a point. Conversely, for a point p the set $p^{-1}(\top)$ is a completely prime filter.

Write SFrm, KSFrm and KRFrm for the full subcategories of Frm whose objects are spatial frames, compact spatial frames and compact regular frames, respectively. For topological spaces, write Sob, KSob and KHaus for the full subcategories of Top whose objects are sober spaces, compact sober spaces and compact Hausdorff spaces respectively. Furthermore, we write \equiv for an equivalence between categories.

3.14 Proposition. The functor pt is a right adjoint to opn. This adjunction restricts to a duality between the category of spatial frames and the category of sober spaces,

$$SFrm \equiv Sob^{op}$$
.

This duality restricts to the dualities

$$KSFrm \equiv KSob^{op}$$

and

$$KRFrm \equiv KHaus^{op}$$
.

For a more thorough exposition of frames and spaces, and a proof of the statements in proposition 3.14 we refer to section C1.2 of [31]. We explicitly mention one isomorphism which is part of this duality, because we will encounter it later on in this thesis.

3.15 Remark. Let \mathfrak{X} be a sober space. Then proposition 3.14 entails that there is an isomorphism $\mathfrak{X} \to \mathfrak{pt}(\mathfrak{opn}\mathfrak{X})$. This isomorphism is given by $x \mapsto p_x$, where p_x is the point given by

$$p_x: \mathfrak{opn} \mathfrak{X} \to 2: \begin{cases} a \mapsto \top & \text{if } x \in a \\ a \mapsto \bot & \text{if } x \notin a \end{cases}$$

for all $x \in \mathbf{X}$ and $a \in \Omega \mathbf{X}$.

3.2 COALGEBRAIC GEOMETRIC LOGIC

As stated in the introduction of this chapter, it is not clear what base space we should use for the models of coalgebraic geometric logic. We have seen at the end of the previous section three candidates, Sob, KSob and KHaus, which all have a dually equivalent algebraic counterpart. We will start this section with defining the language and semantics for coalgebraic geometric logic and thereafter relate the notions of modal equivalence and behavioural equivalence.

The definitions of predicate liftings, geometric \mathbb{T} -models, the language and its semantics are the same whether we are working on Sob, KSob or KHaus. We write C for any of these categories.

3.16 Definition. Let \mathbb{T} be an endofunctor on C. An open predicate lifting for \mathbb{T} is a natural transformation

$$\lambda:\Omega^n\to\Omega\circ\mathbb{T}.$$

A collection of open predicate liftings for \mathbb{T} is called a **geometric modal signature** for \mathbb{T} . An open predicate lifting is called **monotone in its** *i*-th argument if for all a_1, \ldots, a_n, δ we have $\lambda_{\mathfrak{X}}(a_1, \ldots, a_i, \ldots, a_n) \subseteq \lambda_{\mathfrak{X}}(a_1, \ldots, a_i \cup \delta, \ldots, a_n)$ and **monotone** if it is monotone in every argument. A geometric modal signature for a functor \mathbb{T} is called **monotone** if every open predicate lifting in it is monotone, and **characteristic** if for every object \mathfrak{X} in C the collection

$$\{\lambda_{\mathbf{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda \text{ n-ary }, a_i \in \Omega \mathbf{X}\}$$

is a sub-base for the topology on $\mathbb{T}\mathfrak{X}$.

Note that if \mathbb{T} is an endofunctor on Sob that restricts to KSob or KHaus, then an open predicate lifting λ for \mathbb{T} restricts to an open predicate lifting for KSob or KHaus respectively.

The next two definitions are the analogs of definitions 2.26 and 2.27. Recall that Φ is some fixed set of proposition letters.

3.17 Definition. A geometric \mathbb{T} -model for a functor $\mathbb{T} : \mathbb{C} \to \mathbb{C}$ is a triple $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ where (\mathfrak{X}, γ) is a \mathbb{T} -coalgebra and $V : \Phi \to \Omega \mathfrak{X}$ is a valuation of the proposition letters. A map $f : \mathfrak{X} \to \mathfrak{X}'$ is a geometric \mathbb{T} -model morphism from $(\mathfrak{X}, \gamma, V)$ to $(\mathfrak{X}', \gamma', V')$ if f is a coalgebra morphism between the underlying coalgebras and $f^{-1} \circ V' = V$. The collection of geometric \mathbb{T} -models and geometric \mathbb{T} -model morphisms forms a category, which we denote by $\mathsf{Mod}(\mathbb{T})$.

3.18 Definition. The **language** induced by a geometric modal signature is the collection $\text{GML}(\Lambda)$ of formulas defined by the grammar

$$\varphi ::= \mathsf{T} \mid p \mid \varphi_1 \land \varphi_2 \mid \bigvee \varphi_i \mid \heartsuit^{\lambda}(\varphi_1, \dots, \varphi_n),$$

where $p \in \Phi$ and $\lambda \in \Lambda$ is *n*-ary. Abbreviate $\perp := \bigvee \emptyset$.

3.19 Definition. The semantics of $\varphi \in \text{GML}(\Lambda)$ on a geometric \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$, where \mathbb{T} is an endofunctor on C , is given recursively by

$$\llbracket \mathsf{T} \rrbracket^{\mathfrak{X}} = X, \quad \llbracket p \rrbracket^{\mathfrak{X}} = V(p), \quad \llbracket \varphi \wedge \psi \rrbracket^{\mathfrak{X}} = \llbracket \varphi \rrbracket^{\mathfrak{X}} \cap \llbracket \psi \rrbracket^{\mathfrak{X}}, \quad \llbracket \bigvee_{i \in I} \varphi_i \rrbracket^{\mathfrak{X}} = \bigcup_{i \in I} \llbracket \varphi_i \rrbracket^{\mathfrak{X}}, \\ \llbracket \varphi^{\lambda}(\varphi_1, \dots, \varphi_n) \rrbracket^{\mathfrak{X}} = \gamma^{-1}(\lambda_{\mathfrak{X}}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}})).$$

We write $\mathfrak{X}, x \Vdash \varphi$ iff $x \in \llbracket \varphi \rrbracket^{\mathfrak{X}}$. Two states x and x' are called **modally equivalent** if they satisfy the same formulas, notation $x \equiv_{\Lambda} x'$.

 \triangleleft

 \triangleleft

The following proposition shows that morphisms preserve truth. Its proof is similar to the proof of theorem 6.17 in [56]. We give it here for the sake of completeness.

3.20 Proposition. Let $\mathbb{T} : \mathsf{C} \to \mathsf{C}$ be a functor and Λ a geometric modal signature for \mathbb{T} . Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be geometric \mathbb{T} -models and let $f : \mathfrak{X} \to \mathfrak{X}'$ be a geometric \mathbb{T} -model morphism. Then for all $\varphi \in \mathrm{GML}(\Lambda)$ and $x \in \mathfrak{X}$ we have

$$\mathfrak{X}, x \Vdash \varphi \quad i\!f\!f \quad \mathfrak{X}', f(x) \Vdash \varphi.$$

Proof. We will prove that $\llbracket \varphi \rrbracket^{\mathfrak{X}} = (\Omega f) \llbracket \varphi \rrbracket^{\mathfrak{X}'}$ for all formulas φ using induction on the complexity of the formula. The propositional case follows from the definition of a geometric \mathbb{T} -model morphism. The cases \wedge and \vee are routine, so the only case left is $\llbracket \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n) \rrbracket^{\mathfrak{X}} = (\Omega f) \llbracket \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n) \rrbracket^{\mathfrak{X}'}.$

Since f is a coalgebra morphism the left diagram below commutes. Applying Ω to the diagram yields the right commutative diagram.

We observe

$$[\![\varphi^{\lambda}(\varphi_{1}, \dots, \varphi_{n})]\!]^{\mathfrak{X}} = (\Omega\gamma)(\lambda_{\mathfrak{X}}([\![\varphi_{1}]\!]^{\mathfrak{X}}, \dots, [\![\varphi_{n}]\!]^{\mathfrak{X}})) \qquad (\text{definition 3.19})$$

$$= (\Omega\gamma)(\lambda_{\mathfrak{X}}((\Omega f)[\![\varphi_{1}]\!]^{\mathfrak{X}'}, \dots, (\Omega f)[\![\varphi_{n}]\!]^{\mathfrak{X}'})) \qquad (\text{induction})$$

$$= (\Omega\gamma)(\Omega(\mathbb{T}f)(\lambda_{\mathfrak{X}'}([\![\varphi_{1}]\!]^{\mathfrak{X}'}, \dots, [\![\varphi_{n}]\!]^{\mathfrak{X}'}))) \qquad (\text{naturality of }\lambda)$$

$$= (\Omega f)(\Omega\gamma')\lambda_{\mathfrak{X}'}([\![\varphi_{1}]\!]^{\mathfrak{X}'}, \dots, [\![\varphi_{n}]\!]^{\mathfrak{X}'}) \qquad (\text{by (3.1)})$$

$$= (\Omega f)[\![\varphi^{\lambda}(\varphi_{1}, \dots, \varphi_{n})\!]^{\mathfrak{X}'}. \qquad (\text{definition 3.19})$$

This proves the proposition.

We will now relate modal equivalence to behavioural equivalence. It is necessary to restrict our attention to a single base category, namely Sob , rather than the base category C we have been working with until now. The following remark is very important.

3.21 Remark. The notion of behavioural equivalence depends on the base category we are working over! Suppose \mathbb{T} is a functor on Sob, Λ is a geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ are two geometric \mathbb{T} -models. If $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are behaviourally equivalent, then there exists a $\mathfrak{Z} \in \mathsf{Mod}(\mathbb{T})$ and model morphisms $f: \mathfrak{X} \to \mathfrak{Z}$ and $f': \mathfrak{X}' \to \mathfrak{Z}$ such that f(x) = f'(x').

Now suppose that \mathbb{T} restricts to KHaus (write $\mathbb{T}_{\mathsf{KHaus}}$ for this restriction) and \mathfrak{X} and \mathfrak{X}' happen to be compact Hausdorff spaces. If we view \mathfrak{X} and \mathfrak{X}' as geometric $\mathbb{T}_{\mathsf{KHaus}}$ -models, the elements x and x' are not necessarily behaviourally equivalent. Indeed, the model \mathfrak{Z} need not be compact Hausdorff. Therefore, we should be careful with the notion of behavioural equivalence and, if ambiguity arises, specify with respect to which base category the elements are behaviourally equivalent.

It turns out that modal equivalence and behavioural equivalence coincide when C = Sob and the geometric modal signature for a functor is characteristic. The remainder of this section is devoted to proving the following theorem, and a variation thereof.

3.22 Theorem. Let \mathbb{T} be an endofunctor on Sob, Λ a characteristic geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ two geometric \mathbb{T} -models. Then $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are modally equivalent if and only if they are behaviourally equivalent.

In order to prove this theorem, we first investigate the necessary constructions.

3.23 Definition. Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . We call two formulas φ and ψ equivalent on $\mathsf{Mod}(\mathbb{T})$ with respect to Λ , notation $\varphi \equiv_{\mathbb{T},\Lambda} \psi$ if $\mathfrak{X}, x \Vdash \varphi$ iff $\mathfrak{X}, x \Vdash \psi$ for all geometric \mathbb{T} -models \mathfrak{X} and states $x \in \mathfrak{X}$. Denote the equivalence class of φ by $[\varphi]$. Let **Equiv** $(\mathbb{T}, \Lambda, \Phi)$ be the collection of formulas modulo $\equiv_{\mathbb{T},\Lambda}$ and define disjunction and arbitrary conjunction by

$$\begin{split} [\varphi] \wedge [\psi] \coloneqq [\varphi \wedge \psi] \\ \bigvee_{i \in I} [\varphi_i] \coloneqq \Big[\bigvee_{i \in I} \varphi_i\Big]. \end{split}$$

(This is well defined by lemma 3.24.) We call this the **equivalence frame** for \mathbb{T} with respect to Λ .

3.24 Lemma. Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . Then Equiv $(\mathbb{T}, \Lambda, \Phi)$ is indeed a frame.

Proof. We need to show the conjunction and disjunction from definition 3.23 are well defined, that is, they do not rely on the choice of the representatives. Suppose $\varphi_i \equiv_{\mathbb{T},\Lambda} \psi_i$ for all *i* in some index set *I*. Then

$$\mathfrak{X}, x \Vdash \bigvee_{i \in I} \varphi_i \quad \text{iff} \quad \mathfrak{X}, x \Vdash \varphi_i \text{ for some } i \in I \quad \text{iff} \quad \mathfrak{X}, x \Vdash \psi_i \quad \text{iff} \quad \mathfrak{X}, x \Vdash \bigvee_{i \in I} \psi_i.$$

The case for the conjunction is similar.

The theory of a point x in a geometric \mathbb{T} -model \mathfrak{X} is the collection of formulas that are true at x. It is easy to show that the theory of x is a completely prime filter in Equiv $(\mathbb{T}, \Lambda, \Phi)$. This motivate the next definition.

3.25 Definition. Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . Define $\mathfrak{Z} = \mathfrak{pt}(\mathbf{Equiv}(\mathbb{T}, \Lambda, \Phi))$. For every geometric \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ define

$$\operatorname{th}_{\mathfrak{X}}: \mathfrak{X} \to \mathfrak{Z}: x \mapsto \{\varphi \in \operatorname{GML}(\Lambda) \mid \mathfrak{X}, x \Vdash \varphi\}.$$

The space \mathfrak{Z} will turn out to be the state space of a final coalgebra in $\mathsf{Mod}(\mathbb{T})$. We use the letter \mathfrak{Z} because it is the final letter of the roman alphabet. The functor \mathbb{T} has a dual on SFrm.

3.26 Definition. Let \mathbb{T} be a functor on Sob and Λ a characteristic geometric modal signature for \mathbb{T} . Define

$$\mathbb{L}: \mathsf{Frm} \to \mathsf{Frm}$$

by

$$\mathbb{L} = \mathfrak{opn} \circ \mathbb{T} \circ \mathfrak{pt}.$$

Obviously, this functor restricts to an endofunctor on SFrm, where it is (trivially) dual to \mathbb{T} .

We will now endow $\mathbf{Equiv}(\mathbb{T}, \Lambda, \Phi)$ with an \mathbb{L} -algebra structure. For tidiness, and because no confusion is likely to arise, we will write \mathcal{E} for $\mathbf{Equiv}(\mathbb{T}, \Lambda, \Phi)$.

Since Λ is characteristic, the frame $\mathbb{L}\mathscr{E}$ is generated by the sets $\{\lambda_{\mathfrak{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda, a_i \in \Omega \mathfrak{Z}\}$. Therefore we can define an assignment on these generators and use remark 3.3 to extend this to a frame homomorphism $\mathbb{L}\mathscr{E} \to \mathscr{E}$. Recall that by definition all open sets of $\mathfrak{pt}\mathscr{E}$ are of the form $\widetilde{\varphi}$ for some formula $\varphi \in \text{GML}(\Lambda)$.

3.27 Definition. Endow \mathcal{E} with an \mathbb{L} -algebra structure $\delta : \mathbb{L}\mathcal{E} \to \mathcal{E}$, where δ is defined on generators by

$$\delta : \mathbb{L}\mathcal{E} \to \mathcal{E} : \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1, \dots, \widetilde{\varphi}_n) \mapsto [\mathfrak{Q}^{\lambda}(\varphi_1, \dots, \varphi_n)]. \qquad \triangleleft$$

3.28 Lemma. δ is well defined.

Proof. In order to prove that this is well-defined we need to show that the images of the generators of \mathcal{E} satisfy the same relations that they satisfy in $\mathbb{L}\mathcal{E}$. Recall $\mathfrak{Z} = \mathfrak{pt}\mathcal{E}$, then $\mathbb{L}\mathcal{E} = \mathfrak{opn}(\mathbb{T}\mathfrak{Z})$. We need to show that

$$\bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda_{\mathcal{Z}}^{i,j}(\widetilde{\varphi}_1^{i,j}, \dots, \widetilde{\varphi}_{n_{i,j}}^{i,j}) \right) = \bigcup_{k \in K} \left(\bigcap_{\ell \in L_k} \lambda_{\mathcal{Z}}^{k,\ell}(\widetilde{\varphi}_1^{k,\ell}, \dots, \widetilde{\varphi}_{n_{k,\ell}}^{k,\ell}) \right)$$
(3.2)

implies

$$\bigvee_{i \in I} \Big(\bigwedge_{j \in J_i} \heartsuit^{\lambda^{i,j}} (\varphi_1^{i,j}, \dots, \varphi_{n_{i,j}}^{i,j}) \Big) \equiv_{\mathbb{T}, \Lambda} \bigvee_{k \in K} \Big(\bigwedge_{\ell \in L_k} \heartsuit^{\lambda^{k,\ell}} (\varphi_1^{k,\ell}, \dots, \varphi_{n_{k,\ell}}^{k,\ell}) \Big),$$
(3.3)

where the J_i and L_k are finite index sets. We will see that this follows from naturality of λ . Our strategy is to show that the truth sets of the right hand-side and left hand-side of (3.3) coincide in every geometric \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$.

Observe that the map $\operatorname{th}_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Z}$, which sends a point to its theory, is continuous because

$$\operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}) = \llbracket \varphi \rrbracket^{\mathfrak{X}}, \tag{3.4}$$

which is open in \mathfrak{X} for all formulas φ . Compute

$$\begin{split} &\bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda_{\mathfrak{X}}^{i,j} (\llbracket \varphi_1^{i,j} \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_{n_{i,j}}^{i,j} \rrbracket^{\mathfrak{X}}) \right) \\ &= \bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda_{\mathfrak{X}}^{i,j} (\operatorname{th}_{\mathfrak{X}}^{-1} (\widetilde{\varphi}_1^{i,j}), \dots, \operatorname{th}_{\mathfrak{X}}^{-1} (\widetilde{\varphi}_{n_{i,j}}^{i,j})) \right) \qquad \text{(by (3.4))} \\ &= \bigcup_{i \in I} \left(\bigcap_{j \in J_i} (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1} (\lambda_{\mathfrak{Z}}^{i,j} (\widetilde{\varphi}_1^{i,j}, \dots, \widetilde{\varphi}_{n_{i,j}}^{i,j})) \right) \qquad (\text{naturality of } \lambda) \\ &= (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1} \left(\bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda_{\mathfrak{Z}}^{i,j} (\widetilde{\varphi}_1^{i,j}, \dots, \widetilde{\varphi}_{n_{i,j}}^{i,j}) \right) \right) \qquad (\star) \\ &= (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1} \left(\bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda_{\mathfrak{Z}}^{k,\ell} (\widetilde{\psi}_i^{k,\ell}, \dots, \widetilde{\psi}_{n,j}^{k,\ell}) \right) \right) \qquad (\text{assumption (3.2)} \end{split}$$

$$= (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1} \Big(\bigcup_{k \in K} \Big(\bigcap_{\ell \in L_k} \lambda_{\mathfrak{Z}}^{k,\ell} (\widetilde{\psi}_1^{k,\ell}, \dots, \widetilde{\psi}_{n_{k,\ell}}^{k,\ell}) \Big) \Big) \qquad (\text{assumption (3.2)})$$

$$= \bigcup_{k \in K} \left(\bigcap_{\ell \in L_k} (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1} \left(\lambda_{\mathfrak{X}}^{k,\ell} (\widetilde{\psi}_1^{k,\ell}, \dots, \widetilde{\psi}_{n_{k,\ell}}^{k,\ell}) \right) \right)$$
(*)

$$= \bigcup_{k \in K} \left(\bigcap_{\ell \in L_k} \lambda_{\mathfrak{X}}^{k,\ell}(\operatorname{th}_{\mathfrak{X}}^{-1}(\psi_1^{k,\ell}), \dots, \operatorname{th}_{\mathfrak{X}}^{-1}(\psi_{n_{k,\ell}}^{k,\ell})) \right)$$
(naturality of λ)

$$= \bigcup_{k \in K} \Big(\bigcap_{\ell \in L_k} \lambda_{\mathfrak{X}}^{k,\ell} (\llbracket \psi_1^{k,\ell} \rrbracket^{\mathfrak{X}}, \dots, \llbracket \psi_{n_{k,\ell}}^{k,\ell} \rrbracket^{\mathfrak{X}}) \Big).$$
 (by (3.4))

The steps with (\star) hold because inverse images of maps preserve all unions and intersections. This entails that for all geometric T-models and all states x in \mathfrak{X} we have

$$\mathfrak{X}, x \Vdash \bigvee_{i \in I} \Big(\bigwedge_{j \in J_i} \heartsuit^{\lambda^{i,j}}(\varphi_1^{i,j}, \dots, \varphi_{n_{i,j}}^{i,j}) \Big) \quad \text{iff} \quad \mathfrak{X}, x \Vdash \bigvee_{k \in K} \Big(\bigwedge_{\ell \in L_k} \heartsuit^{\lambda^{k,\ell}}(\varphi_1^{k,\ell}, \dots, \varphi_{n_{k,\ell}}^{k,\ell}) \Big),$$

and hence (3.3) holds. Therefore δ is well defined.

Since we defined δ on generators of L \mathcal{E} , by remark 3.3 it extends to a frame homomorphism which, by abuse of notation, we shall also denote by δ . The algebra structure on \mathcal{E} entails a coalgebra structure on \mathcal{Z} .

3.29 Definition. Let $\zeta : \mathfrak{Z} \to \mathbb{T}\mathfrak{Z}$ be the composition

$$\mathfrak{pt} \mathscr{E} \xrightarrow{\mathfrak{pt} \delta} \mathfrak{pt}(\mathbb{L} \mathscr{E}) = \mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt} \mathscr{E}))) \xrightarrow{k_{\mathbb{T}(\mathfrak{pt} \mathscr{E})}^{-1}} \mathbb{T}(\mathfrak{pt} \mathscr{E})$$

Here $k_{\mathbb{T}(\mathfrak{pt}\mathcal{E})} : \mathbb{T}(\mathfrak{pt}\mathcal{E}) \to \mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt}\mathcal{E})))$ is the isomorphism given in remark 3.15. Since $\mathfrak{Z} = \mathfrak{pt}\mathcal{E}$ this indeed defines a map $\mathfrak{Z} \to \mathbb{T}\mathfrak{Z}$.

For an object $\Gamma \in \mathfrak{Z}$, $(\mathfrak{pt}\delta)(\Gamma)$ is the completely prime filter

$$F = \{\lambda(\widetilde{\varphi}_1, \dots, \widetilde{\varphi}_n) \in \mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt}\,\mathcal{E}))) \mid [\mathfrak{O}^{\lambda}(\varphi_1, \dots, \varphi_n)] \in \Gamma\}$$

in $\mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt}\mathscr{E})))$. The element $\zeta(\Gamma)$ is the unique element in $\mathbb{T}(\mathfrak{pt}\mathscr{E})$ that corresponds to F under the isomorphism $k_{\mathbb{T}(\mathfrak{pt}\mathscr{E})}$. By definition of $k_{\mathbb{T}(\mathfrak{pt}\mathscr{E})}$, this is the unique element in the intersection of

$$\{\lambda(\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n) \mid [\heartsuit^\lambda(\varphi_1,\ldots,\varphi_n)] \in \Gamma\}.$$

Moreover, it follows from the definition of $k_{\mathbb{T}(\mathfrak{pt}\mathcal{E})}$ that $[\mathfrak{S}^{\lambda}(\varphi_1,\ldots,\varphi_n)] \notin \Gamma$ implies $\zeta(\Gamma) \notin \lambda(\widetilde{\varphi}_1,\ldots,\widetilde{\varphi}_n)$.

Notation. If no confusion is likely to occur we will omit the square brackets that indicate equivalence classes of formulas in \mathcal{E} . That is, we shall write $\varphi \in \mathcal{E}$ instead of $[\varphi] \in \mathcal{E}$.

3.30 Definition. Let $V_{\mathfrak{Z}} : \Phi \to \Omega \mathfrak{Z}$ be the valuation $p \mapsto \widetilde{p}$.

The triple $\mathfrak{Z} = (\mathfrak{Z}, \zeta, V_{\mathfrak{Z}})$ is a geometric \mathbb{T} -model, because it is a \mathbb{T} -coalgebra with a valuation. We can prove a truth lemma for \mathfrak{Z} :

3.31 Lemma (Truth lemma). We have $\mathfrak{Z}, \Gamma \Vdash \varphi$ iff $\varphi \in \Gamma$.

Proof. Use induction on the complexity of the formula. The propositional case follows immediately from the definition of $V_{\mathfrak{Z}}$. The cases $\varphi = \varphi_1 \wedge \varphi_2$ and $\varphi = \bigvee_{i \in i} \varphi_i$ are routine. So suppose $\varphi = \heartsuit^{\lambda}(\varphi_1, \ldots, \varphi_n)$. We have

$$\begin{aligned} \mathfrak{Z}, \Gamma \Vdash \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) & \text{iff} \quad \zeta(\Gamma) \in \lambda_{\mathfrak{Z}}(\llbracket \varphi_{1} \rrbracket^{3}, \dots, \llbracket \varphi_{n} \rrbracket^{3}) & (\text{definition of } \Vdash) \\ & \text{iff} \quad \zeta(\Gamma) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n}) & (\text{induction}) \\ & \text{iff} \quad \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \in \Gamma. & (\text{definition of } \zeta) \end{aligned}$$

This proves the lemma.

3.32 Proposition. For every geometric \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ the map $\operatorname{th}_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Z}$ is a \mathbb{T} -coalgebra morphism.

 \triangleleft

Proof. We need to show that the following diagram commutes:

$$\begin{array}{c} x \xrightarrow{\operatorname{th}_{\mathfrak{X}}} \mathfrak{Z} \\ \gamma \downarrow & \qquad \qquad \downarrow \zeta \\ \mathbb{T}x \xrightarrow{} \mathbb{T}\operatorname{th}_{\mathfrak{X}}} \mathbb{T}z \end{array}$$

Let $x \in \mathfrak{X}$. Since $\mathbb{T}\mathfrak{Z}$ is sober, hence T_0 , it suffices to show that $\mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x))$ and $\zeta(\operatorname{th}_{\mathfrak{X}}(x))$ are in precisely the same opens of $\mathbb{T}\mathfrak{Z}$. Moreover, we know that the open sets of $\mathbb{T}\mathfrak{Z}$ are generated by the sets $\lambda_{\mathfrak{Z}}(\widetilde{\varphi}_1, \ldots, \widetilde{\varphi}_n)$, so it suffices to show that for all $\lambda \in \Lambda$ and $\varphi_i \in \operatorname{GML}(\Lambda)$ we have

 $\mathbb{T}\operatorname{th}_{\mathfrak{X}}(\gamma(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1},\ldots,\widetilde{\varphi}_{n}) \quad \text{iff} \quad \zeta(\operatorname{th}_{\mathfrak{X}}(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1},\ldots,\widetilde{\varphi}_{n}).$

This follows from the following computation,

$$\begin{split} \mathbb{T} \operatorname{th}_{\mathfrak{X}}(\gamma(x)) &\in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n}) \\ & \text{iff} \quad \gamma(x) \in (\mathbb{T} \operatorname{th}_{\mathfrak{X}})^{-1}(\lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n})) \\ & \text{iff} \quad \gamma(x) \in \lambda_{\mathfrak{X}}(\operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{1}), \dots, \operatorname{th}_{\mathfrak{X}}^{-1}(\widetilde{\varphi}_{n})) \\ & \text{iff} \quad \gamma(x) \in \lambda_{\mathfrak{X}}(\llbracket \varphi_{1} \rrbracket^{\mathfrak{X}}, \dots, \llbracket \varphi_{n} \rrbracket^{\mathfrak{X}}) \\ & \text{iff} \quad \mathfrak{X}, x \Vdash \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \\ & \text{iff} \quad \mathfrak{T}^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \in \operatorname{th}_{\mathfrak{X}}(x) \\ & \text{iff} \quad \zeta(\operatorname{th}_{\mathfrak{X}}(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n}) \\ \end{split}$$
 (definition of th_{\mathfrak{X}}) \\ & \text{iff} \quad \zeta(\operatorname{th}_{\mathfrak{X}}(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{\varphi}_{1}, \dots, \widetilde{\varphi}_{n}) \\ \end{split}

This proves the proposition.

3.33 Theorem. The geometric \mathbb{T} -model $\mathfrak{Z} = (\mathfrak{Z}, \zeta, V_{\mathfrak{Z}})$ is final in $\mathsf{Mod}(\mathbb{T})$.

Proof. Proposition 3.32 states that for every geometric \mathbb{T} -model $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ there exists a \mathbb{T} -coalgebra morphism $\operatorname{th}_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Z}$, so we only need to show that this morphism is unique.

Let $f: \mathfrak{X} \to \mathfrak{Z}$ be any coalgebra morphism. We know from proposition 3.20 coalgebra morphisms preserve truth, so for all $x \in \mathfrak{X}$ we have $\varphi \in f(x)$ iff $\mathfrak{Z}, f(x) \Vdash \varphi$ iff $\mathfrak{X}, x \Vdash \varphi$. Therefore we must have $f(x) = \operatorname{th}_{\mathfrak{X}}(x)$.

We now have all the tools to prove theorem 3.22.

3.22 Theorem. Let \mathbb{T} be an endofunctor on Sob, Λ a characteristic geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ two geometric \mathbb{T} -models. Then $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are modally equivalent if and only if they are behaviourally equivalent.

Proof of theorem 3.22. It follow from proposition 3.32 and theorem 3.33 that $\mathfrak{Z} = (\mathfrak{Z}, \zeta, V_{\mathfrak{Z}})$ is final in $\mathsf{Mod}(\mathbb{T})$. Suppose x and x' are modally equivalent, then $\mathrm{th}_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Z}$ and $\mathrm{th}_{\mathfrak{X}'} : \mathfrak{X}' \to \mathfrak{Z}'$ are \mathbb{T} -model morphisms such that $\mathrm{th}_{\mathfrak{X}}(x) = \mathrm{th}_{\mathfrak{X}'}(x')$, so x and x' are behaviourally equivalent. Conversely, if x and x' are behaviourally equivalent, then we must have $\mathrm{th}_{\mathfrak{X}}(x) = \mathrm{th}_{\mathfrak{X}'}(x')$ so $x \equiv_{\Lambda} x'$.

Theorem 3.22 does not simply carry over to coalgebras on compact Hausdorff spaces. The reason is that the equivalence frame is not necessarily compact or regular. Therefore its dual, which is the prime candidate of the state space of a final coalgebra, need not be compact Hausdorff. This hinders the construction of a final coalgebra of theories. However, with a little work we can achieve a similar result for the base category KSob of compact sober spaces and continuous maps.

3.34 Theorem. Let \mathbb{T} be an endofunctor on KSob, Λ a characteristic geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ two geometric \mathbb{T} -models. Then $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are modally equivalent if and only if they are behaviourally equivalent.

Proof. If x and x' are behaviourally equivalent, then there exists a geometric $\mathbb{T}_{\mathsf{KSob}}$ model $\mathfrak{U} = (\mathfrak{U}, \nu, V_{\mathfrak{U}})$ and \mathbb{T} -model morphisms $f : \mathfrak{X} \to \mathfrak{U}$ and $f' : \mathfrak{X}' \to \mathfrak{U}$ such that f(x) = f'(x'). It follows from proposition 3.20 that x and x' are modally equivalent.

For the converse, we use a procedure similar to the one in between the statement of theorem 3.22 and its proof.

Suppose x and x' are modally equivalent. Define the equivalence relation \equiv_2 on $\text{GML}(\Lambda)$ by $\varphi \equiv_2 \psi$ iff $\llbracket \varphi \rrbracket^{\mathfrak{X}} = \llbracket \psi \rrbracket^{\mathfrak{X}}$ and $\llbracket \varphi \rrbracket^{\mathfrak{X}'} = \llbracket \psi \rrbracket^{\mathfrak{X}'}$. Let \mathscr{E}_2 be the collection of formulae modulo \equiv_2 and denote by $\llbracket \varphi \rrbracket$ the equivalence class of φ in \mathscr{E}_2 . (Note these square brackets mean something different than the ones in definition 3.23.) Define disjunction and conjunction by

$$[\varphi] \land [\psi] \coloneqq [\varphi \land \psi] \quad \text{and} \quad \bigvee_{i \in I} [\varphi_i] \coloneqq \Big[\bigvee_{i \in I} \varphi_i\Big].$$

It is routine to check that \mathcal{E}_2 is indeed a frame. We claim that \mathcal{E}_2 is compact.

3.34.A Claim. The frame \mathcal{E}_2 is compact.

Proof of claim. Suppose $\bigvee_{i \in I} [\varphi_i] = [\intercal]$ in \mathcal{E} . Then $\bigvee_{i \in I} \varphi_i \equiv_2 \intercal$, so by definition

$$\bigcup_{i\in I} \llbracket \varphi_i \rrbracket^{\mathfrak{X}} = \mathfrak{X} \quad \text{and} \quad \bigcup_{i\in I} \llbracket \varphi_i \rrbracket^{\mathfrak{X}'} = \mathfrak{X}'.$$

By compactness of \mathfrak{X} and \mathfrak{X}' there exist finite sets $J, J' \subseteq I$ such that

$$\bigcup_{i \in J} \llbracket \varphi_i \rrbracket^{\mathfrak{X}} = \mathfrak{X} \quad \text{and} \quad \bigcup_{i \in J'} \llbracket \varphi_i \rrbracket^{\mathfrak{X}'} = \mathfrak{X}'.$$

It follows that

 $\bigvee_{i \in J \cup J'} \varphi_i \equiv_2 \mathsf{T}.$

This proves compactness of \mathcal{E}_2 .

Let $\mathcal{Y} \coloneqq \mathfrak{pt}(\mathcal{E}_2)$. Then \mathcal{Y} is a compact sober space. It is easy to see that the theory of a point in \mathfrak{X} or \mathfrak{X}' , i.e. the collection of formulas it satisfies, is a completely prime filter in \mathcal{E}_2 . Therefore we may define

$$\operatorname{th}_{\mathfrak{X}}': \mathfrak{X} \to \mathfrak{Y}: x \mapsto \{\varphi \in \operatorname{GML}(\Lambda) \mid \mathfrak{X}, x \Vdash \varphi\}.$$

For every open set $[\varphi]$ in \mathcal{Y} we have $\operatorname{th}'_{\mathfrak{X}}([\varphi]) = \{x \in \mathfrak{X} \mid \mathfrak{X}, x \Vdash \varphi\} = [\![\varphi]\!]^{\mathfrak{X}}$. Since $[\![\varphi]\!]^{\mathfrak{X}}$ is open in \mathfrak{X} this shows that $\operatorname{th}'_{\mathfrak{X}}$ is continuous. Similarly, we may define a continuous map $\operatorname{th}'_{\mathfrak{X}'} : \mathfrak{X}' \to \mathfrak{Y}$.

Let \mathbb{L} be the functor $\mathbb{L} = \mathfrak{opn} \circ \mathbb{T} \circ \mathfrak{pt}$ on KSFrm. Then \mathbb{L} is dual to \mathbb{T} . Since Λ is characteristic, the frame $\mathbb{L}\mathscr{E}_2$ is generated by the sets

$$\{\lambda y(\widetilde{[\varphi_1]},\ldots,\widetilde{[\varphi_n]}) \mid \lambda \in \Lambda, \widetilde{[\varphi_i]} \in \Omega \mathcal{Y}\}$$

Endow \mathcal{E}_2 with an \mathbb{L} -algebra structure $\delta_2 : \mathbb{L}\mathcal{E}_2 \to \mathcal{E}_2$ defined on generators by

$$\delta: \mathbb{L}\mathscr{E}_2 \to \mathscr{E}_2: \lambda_{\mathfrak{Y}}([\widetilde{\varphi_1}], \dots, [\widetilde{\varphi_n}]) \mapsto [\mathfrak{S}^{\lambda}(\varphi_1, \dots, \varphi_n)].$$

	^
1	``
`	. /
	~

One can proof that δ_2 is well defined in a manner similar to the proof of 3.28. It then follows from remark 3.3 that the assignment δ_2 indeed defines a homomorphism, which we shall also call δ_2 .

We use the algebra structure on \mathcal{E}_2 to define a coalgebra on \mathcal{Y} . Let $\beta : \mathcal{Y} \to \mathbb{T}\mathcal{Y}$ be the composition

$$\mathfrak{pt} \mathscr{E}_2 \xrightarrow{\mathfrak{pt} \delta} \mathfrak{pt}(\mathbb{L} \mathscr{E}_2) = \mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt} \mathscr{E}_2))) \xrightarrow{k_{\mathbb{T}(\mathfrak{pt} \mathscr{E}_2)}^{-1}} \mathbb{T}(\mathfrak{pt} \mathscr{E}_2).$$

Here $k_{\mathbb{T}(\mathfrak{pt} \mathcal{E}_2)} : \mathbb{T}(\mathfrak{pt} \mathcal{E}_2) \to \mathfrak{pt}(\mathfrak{opn}(\mathbb{T}(\mathfrak{pt} \mathcal{E}_2)))$ is the isomorphism given in remark 3.15. Since $\mathcal{Y} = \mathfrak{pt} \mathcal{E}$ this indeed defines a continuous map $\mathcal{Y} \to \mathbb{T} \mathcal{Y}$. It follows from unravelling the definitions that for any $\Gamma \in \mathcal{Y}$, the image $\beta(\Gamma)$ is the unique element x in $\mathbb{T} \mathcal{Y}$ satisfying

$$\beta(\Gamma) \in \lambda \mathfrak{y}(\widetilde{[\varphi_1]}, \dots, \widetilde{[\varphi_n]}) \quad \text{iff} \quad [\heartsuit^{\lambda}(\varphi_1, \dots, \varphi_n)] \in \Gamma$$

Endow the coalgebra (\mathcal{Y}, β) with the valuation $V_{\mathcal{Y}} : \Phi \to \Omega \mathcal{Y} : p \mapsto [\widetilde{p}]$ and set

$$\mathfrak{Y} = (\mathcal{Y}, \beta, V_{\mathcal{Y}}).$$

3.34.B Claim. The maps $\operatorname{th}'_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Y}$ and $\operatorname{th}'_{\mathfrak{X}'} : \mathfrak{X} \to \mathfrak{Y}$ are \mathbb{T} -coalgebra morphism.

Proof of claim. We show this for $\text{th}'_{\mathfrak{X}} : \mathfrak{X} \to \mathfrak{Y}$, the case for $\text{th}'_{\mathfrak{X}'}$ being similar. We need to show that the following diagram commutes:

$$\begin{array}{c} x \xrightarrow{\operatorname{th}'_{\mathfrak{X}}} \mathfrak{Y} \\ \gamma \downarrow & \downarrow_{\beta} \\ \mathbb{T}x \xrightarrow{}_{\mathbb{T}\operatorname{th}'_{\mathfrak{X}}} \mathbb{T}\mathcal{Y} \end{array}$$

Let $x \in \mathfrak{X}$. By the reasoning from proposition 3.32 it suffices to show that for all $\lambda \in \Lambda$ and $\varphi_i \in \text{GML}(\Lambda)$ we have

$$\mathbb{T} \operatorname{th}'_{\mathfrak{X}}(\gamma(x)) \in \lambda_{\mathfrak{Y}}([\widetilde{\varphi_1}], \dots, [\widetilde{\varphi_n}]) \quad \text{iff} \quad \beta(\operatorname{th}'_{\mathfrak{X}}(x)) \in \lambda_{\mathfrak{Y}}([\widetilde{\varphi_1}], \dots, [\widetilde{\varphi_n}]).$$

This follows from the following computation,

$$\mathbb{T} \operatorname{th}_{\mathfrak{X}}'(\gamma(x)) \in \lambda_{\mathfrak{Y}}(\widetilde{[\varphi_{1}]}, \dots, \widetilde{[\varphi_{n}]})$$

$$\operatorname{iff} \quad \gamma(x) \in (\mathbb{T} \operatorname{th}_{\mathfrak{X}}')^{-1}(\lambda_{\mathfrak{Y}}(\widetilde{[\varphi_{1}]}, \dots, \widetilde{[\varphi_{n}]}))$$

$$\operatorname{iff} \quad \gamma(x) \in \lambda_{\mathfrak{X}}((\operatorname{th}_{\mathfrak{X}}')^{-1}(\widetilde{[\varphi_{1}]}), \dots, (\operatorname{th}_{\mathfrak{X}}')^{-1}(\widetilde{[\varphi_{n}]})) \quad (\text{naturality of } \lambda)$$

$$\operatorname{iff} \quad \gamma(x) \in \lambda_{\mathfrak{X}}([\mathbb{T} \varphi_{1}]^{\mathfrak{X}}, \dots, [\mathbb{T} \varphi_{n}]^{\mathfrak{X}}) \quad (\text{by } (3.4))$$

$$\operatorname{iff} \quad \mathfrak{X}, x \Vdash \heartsuit^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \quad (\text{definition of } \Vdash)$$

$$\operatorname{iff} \quad \wp^{\lambda}(\varphi_{1}, \dots, \varphi_{n}) \in \operatorname{th}_{\mathfrak{X}}'(x) \quad (\text{definition of } \operatorname{th}_{\mathfrak{X}}')$$

$$\operatorname{iff} \quad \beta(\operatorname{th}_{\mathfrak{X}}'(x)) \in \lambda_{\mathfrak{Z}}(\widetilde{[\varphi_{1}]}, \dots, \widetilde{[\varphi_{n}]}) \quad (\text{definition of } \beta)$$

This proves the claim.

One can easily see that if $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are modally equivalent, then $\operatorname{th}_{\mathfrak{X}}(x) = \operatorname{th}_{\mathfrak{X}'}(x')$. Since $\mathfrak{Y} = (\mathfrak{Y}, \beta, V_{\mathfrak{Y}})$ is a geometric \mathbb{T} -model and \mathfrak{Y} is a compact sober space, this shows that x and x' are behaviourally equivalent in $\operatorname{\mathsf{Mod}}(\mathbb{T})$.

 \diamond
3.3 EXAMPLES

The Vietoris functor, introduced (in a different form) in [64], has been well investigated in the literature [30, 63, 9, 58]. It is defined on the category of all topological spaces and is known to preserve compactness, the Hausdorff property, and zero-dimensionality [37], among others. Let \mathbb{V}_{kh} denote the restriction of the Vietoris functor to KHaus There is an Isbell dual functor \mathbb{N} for \mathbb{V}_{kh} on the category of compact regular frames, that is, there is a functor $\mathbb{N} : \mathsf{KRFrm} \to \mathsf{KRFrm}$ and a natural isomorphism $\mathfrak{opn} \circ \mathbb{V}_{kh} \cong$ $\mathbb{N} \circ \mathfrak{opn}$. Coalgebras for \mathbb{V}_{kh} are modal compact Hausdorff spaces. These spaces have been extensively examined in [8].

3.35 Definition. The map $\mathbb{N} : \mathsf{KRFrm} \to \mathsf{KRFrm}$ is defined on an object F as the free frame generated by $\Box a, \Diamond a, (a \in F)$ subject to the relations

$(N1) \Box T = T$	$(N2) \Leftrightarrow \perp = \perp$
(N3) $\Box(a \land b) = \Box a \land \Box b$	(N4) $\diamondsuit(a \lor b) = \diamondsuit a \lor \diamondsuit b$
$(N5) \ \Box(a \lor b) \le \Box a \lor \Diamond b$	(N6) $\Box a \land \Diamond b \leq \Diamond (a \land b)$
$(N7) \Box \bigvee A = \bigvee \{ \Box a \mid a \in A \}$	(N8) $\diamondsuit \bigvee A = \bigvee \{ \diamondsuit a \mid a \in A \}.$

For frame morphisms $f: F \to F'$ define $\mathbb{N}f: \mathbb{N}F \to \mathbb{N}F'$ on generators by $(\mathbb{N}f)(\Box a) = \Box(f(a))$ and $(\mathbb{N}f)(\diamondsuit a) = \diamondsuit(f(a))$.

In [57] a different but equivalent definition of \mathbb{N} is given, which uses the cover modality.

We state the following proposition for future reference. A proof may be found in [30] or in section III.4 of [29]. An overview about dual functors for \mathbb{V} for different base spaces can be found in [58].

3.36 Proposition. There is a natural isomorphism $\mathfrak{opn} \circ \mathbb{V}_{\mathsf{KHaus}} \cong \mathbb{N} \circ \mathfrak{opn}$.

In the remainder of this section we focus on a generalisation of the monotone functor from definition 2.38 and an Isbell dual for it. Observe that the assignment from definition 2.38 in fact defines an endofunctor on KHaus.

3.37 Definition. Let $\mathfrak{X} = (X, \tau)$ be a compact Hausdorff space. Let $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ be the collection of sets $W \subseteq \mathbb{P}X$ such that $a \in W$ iff there exists a closed $c \subseteq u$ such that every open superset of c is in W. Endow $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ with the topology generated by the subbase

$$\boxplus a := \{ W \in \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \mid a \in W \}, \quad \&a := \{ W \in \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \mid X \setminus a \notin W \}.$$

where a ranges over $\Omega \mathfrak{X}$. For continuous functions $f : \mathfrak{X} \to \mathfrak{X}'$ define $\mathbb{D}_{\mathsf{kh}} f : \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \to \mathbb{D}_{\mathsf{kh}} \mathfrak{X}' : W \mapsto \{a \in \mathbb{P}X \mid f^{-1}(a) \in W\}.$

The proof that $\mathbb{D}_{\mathsf{kh}}f$ is indeed a well-defined continuous function is the same as in lemma 2.39. We can easily show that \mathbb{D}_{kh} preserves compactness.

3.38 Lemma. If \mathfrak{X} is a compact space then so is $\mathbb{D}\mathfrak{X}$.

Proof. By the Alexander subbasis theorem it suffices to show that any cover of the form

$$\bigcup_{i\in I} \boxplus a_i \cup \bigcup_{j\in J} \otimes \mathcal{B}_j$$

has a finite subcover. So suppose the above covers $\mathbb{D}\mathfrak{X}$. Since $\emptyset \in \mathbb{D}\mathfrak{X}$ and $\emptyset \notin \boxplus a$ for any open set $a \in \Omega\mathfrak{X}$, we must have $|J| \ge 1$. Furthermore, we must have $k \in I$ such that a_k is a superset of $X \setminus b_j$ for some $j \in J$, because otherwise the up-set $\uparrow \{X \setminus b_j \mid j \in J\}$, where each member is a superset of at least one of the b_j , is not in the cover.

We know now that \mathbb{D}_{kh} preserves compactness. However, we do not yet know whether \mathbb{D}_{kh} preserves the Hausdorff property as well. We will not prove this directly, but first give an Isbell dual \mathbb{M} of \mathbb{D}_{kh} , i.e. a functor on Frm such that for all compact Hausdorff spaces \mathfrak{X} we have

$$\mathbb{M}(\mathfrak{opn}\,\mathfrak{X})\cong\mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$$

We then show that $\mathbb{M}(\mathfrak{opn} \mathfrak{X})$ is compact regular, and it follows from proposition 3.14 that $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ is compact Hausdorff.

3.39 Definition. Let *F* be a frame. Let $\mathbb{M}F$ be the frame generated by the set $\mathbb{M}F = \{\Box a, \Diamond a \ (a \in F)\}$ subject to the relations

$(M1) \ \Box(a \land b) \le \Box a$	$(M2) \diamond a \leq \diamond (a \lor b)$
(M3) $\Box a \land \diamondsuit b \leq \bot$ whenever $a \land b \leq \bot$	(M4) $\Box a \lor \diamondsuit b \ge \top$ whenever $a \lor b \ge \top$
$(M5) \Box \bigvee A = \bigvee \{ \Box a \mid a \in A \}$	$(M6) \diamondsuit \forall A = \forall \{ \diamondsuit a \mid a \in A \},\$

where $a, b \in F$ and A is a directed subset of F. For a homomorphism $f: F \to F'$ define $\mathbb{M}f: \mathbb{M}F \to \mathbb{M}F'$ on generators by $\Box a \mapsto \Box f(a)$ and $\Diamond a \mapsto \Diamond f(a)$. The assignment \mathbb{M} defines a functor on Frm. \triangleleft

The following proposition closely resembles that of proposition III4.3 in [29].

3.40 Proposition. If F is a regular frame, then so is MF.

Proof. We need to show that for all $c \in \mathbb{M}F$ we have $c = \bigvee \{d \in \mathbb{M}F \mid d \leq c\}$. It follows from lemma 3.9 that it suffices to focus on the generators of $\mathbb{M}F$. Let $a \in F$, then we know $\bigvee \{d \in \mathbb{M}F \mid d \leq \Box a\} \leq \Box a$. Suppose $b \leq a$ in F, then by lemma 3.8 $\sim b \vee a = \top$ and hence $\diamondsuit \lor b \vee \Box a \geq \top$. Also $\sim b \land b = \bot$ so it follows from (M3) that $\diamondsuit \lor b \land \Box b = \bot$. This proves $\Box b \leq \Box a$, because the element $\diamondsuit \sim b$ is such that $\diamondsuit \sim b \vee \Box a = \top$ and $\diamondsuit \sim b \land \Box b = \bot$. Since F is regular and $\{b \in F \mid b \leq a\}$ is directed, it follows that

$$\Box a = \Box \bigvee \{ b \in F \mid b \leq a \} = \bigcup \{ \Box b \in \mathbb{M}F \mid b \leq a \} \le \bigcup \{ d \in \mathbb{M}F \mid d \leq \Box a \}$$

so $\Box a = \bigvee \{ d \in \mathbb{M}F \mid d \in \Box a \}$. In a similar fashion one may show that $\Diamond a = \bigvee \{ d \in \mathbb{M}F \mid d \in \Diamond a \}$. This proves the lemma. \Box

We now focus on a duality between \mathbb{D}_{kh} and a restriction of \mathbb{M} . The proof of the next theorem is similar to the proof of proposition III4.6 in [29]. The main difference with the proof in [29] is the way we define a point of $\mathfrak{pt}(\mathfrak{M}(\mathfrak{opn}\mathfrak{X}))$ from a given element of $\mathbb{D}_{kh}\mathfrak{X}$. This is (of course) due to the fact that \mathbb{V}_{kh} and \mathbb{D}_{kh} are different functors.

3.41 Theorem. If X is a compact Hausdorff space then

 $\mathfrak{pt}(\mathbb{M}(\mathfrak{opn}\mathfrak{X})) \cong \mathbb{D}_{\mathsf{kh}}\mathfrak{X}.$

Proof. Define a map

$$\varphi: \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \to \mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X})): W \mapsto p_W$$

where we define p_W on generators by

$$p_W: \mathbb{M}(\mathfrak{opn}\,\mathfrak{X}) \to 2: \begin{cases} \Box a \mapsto \top & \text{iff } a \in W \\ \Diamond a \mapsto \bot & \text{iff } X \smallsetminus a \in W \end{cases}$$

Conversely, for a point $p \in \mathfrak{pt}(\mathfrak{M}(\mathfrak{opn} \mathfrak{X}))$ let

$$W_p \coloneqq \uparrow \{X \smallsetminus a \mid p(\diamondsuit a) = \bot\}$$

This gives rise to a map $\psi : \mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X})) \to \mathbb{D}_{\mathsf{kh}}\mathfrak{X}$. It is clear that $W_p \in \mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ because it is the up-set of a collection of closed sets; indeed, for each $\delta \in W_p$ there exists a closed subset $X \setminus a \subseteq \delta$ with $p(\diamondsuit a) = \bot$ and by definition all open supersets of $X \setminus a$ are in W_p .

We will show that the p_W are well-defined, that φ is a bijection and that φ is continuous.

3.41.A Claim. If $W \in \mathbb{D}_{kh} \mathfrak{X}$ then $p_W : \mathbb{M}(\mathfrak{opn} \mathfrak{X}) \to 2$ is a point.

Proof of claim. Since p_W is a frame homomorphisms defined on generators, it suffices to check that the $p_W(\Box a)$ and $p_W(\Diamond a)$ (where the *a* range over $\Omega \mathfrak{X}$) satisfy (M1) through (M6) from definition 3.39. Let us check (M1), (M3) and (M5), items (M2), (M4) and (M6) being similar.

- (M1) If $p_W(\Box(a \cap b)) = \top$ then $a \cap b \in W$. Since W is upward closed $a \in W$, so $p_W(\Box a) = \top$.
- (M3) If $a \cap b = \emptyset$ then $a \subseteq X \setminus b$. Suppose $p_W(\Box a) = \top$ then $a \in W$ so $X \setminus b \in W$ so $p_W(\Diamond b) = \bot$ hence $p_W(\Box a) \wedge p_W(\Diamond b) = \bot$.
- (M5) We claim that for all $W \in \mathbb{D}\mathfrak{X}$ and directed sets $A \subseteq \Omega\mathfrak{X}$ we have $\bigcup A \in W$ iff there is $a \in A$ with $a \in W$. The direction from right to left follows from the fact that Wis upwards closed. Conversely, suppose $\bigcup A \in W$, then there is a closed set $\hbar \subseteq \bigcup A$ with $\hbar \in W$. The elements of A now cover the closed therefore compact set \hbar , so there is a finite $A' \subseteq A$ with $\hbar \subseteq \bigcup A'$ and since A is directed there is $a \in A$ with $\bigcup A' \subseteq a$. As $\hbar \in W$ and $\hbar \subseteq a$ it follows that $a \in W$.

Now we have $p_W(\Box \bigcup A) = 1$ iff $\bigcup A \in W$ iff there is $a \in A$ with $a \in W$ iff $\bigvee \{p_W(\Box a) \mid a \in A\} = 1$.

3.41.B Claim. For all $p \in \mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X}))$ we have $X \setminus a \in W_p$ iff $p(\diamondsuit a) = \bot$ and $a \in W$ iff $p(\square a) = \top$.

Proof of claim. If $p(\diamondsuit a) = \bot$ then $X \smallsetminus a \in W_p$. Conversely, Suppose $X \smallsetminus a \in W_p$, then there is some b with $p(\diamondsuit b) = \bot$ and $X \lor b \subseteq X \lor a$. Therefore $a \subseteq b$ and $p(\diamondsuit a) \le p(\diamondsuit b) = \bot$. This proves $X \lor a \in W_p$ iff $p(\diamondsuit a) = \bot$.

If $a \in W_p$ then there is $X \setminus b \subseteq a$ in W_p , so $p(\Diamond b) = \bot$. Then $a \cup b = X$, so it follows from (M4) of 3.39 that $p(\Box a) = \top$. If $a \notin W_p$ and $a' \notin a$, then there exists b with $b \cap a' = \emptyset$ and $b \cup a = \mathfrak{X}$. Since $X \setminus b \subseteq a$, set set $X \setminus b$ is not in W_p and hence we must have $p(\Diamond b) = \top$. As $a' \cap b = \emptyset$ it follows from (M3) that $p(\Box a') = p(\emptyset) = \bot$. Now we use (M5) and the fact that $a = \bigvee \{a' \mid a' \notin a\}$ (this is true because \mathfrak{X} is assumed to be compact Hausdorff so $\mathfrak{opn}\mathfrak{X}$ is compact regular) to find

$$p(\Box a) = \bigvee \{ p(\Box a') \mid a' \leqslant a \} = \bigvee \{ \bot \mid a' \leqslant a \} = \bot.$$

It follows that $a \in W_p$ iff $p(\Box a) = \top$.

 \diamond

3.41.C Claim. The maps φ and ψ define a bijection between $\mathbb{D}_{kh} \mathfrak{X}$ and $\mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X}))$.

Proof of claim. For $p \in \mathfrak{pt}(\mathbb{M}(\mathfrak{opn}\mathfrak{X}))$ and $W \in \mathbb{D}_{kh}\mathfrak{X}$ we will show that $p_{W_p} = p$ and $W_{p_W} = W$. In order to prove that (the frame homomorphisms) p and p_{W_p} coincide, it suffices to show that they coincide on the generators of $\mathbb{M}(\mathfrak{opn}\mathfrak{X})$. By definition and claim 3.41.B have

$$p(\Box a) = \top$$
 iff $a \in W_p$ iff $p_{W_p}(\Box a) = \top$

and

$$p(\diamondsuit a) = \bot$$
 iff $X \smallsetminus a \notin W_p$ iff $p_{W_p}(\diamondsuit a) = \bot$.

In order to show that $W = W_{p_W}$ it suffices to show that $X \setminus a \in W$ iff $X \setminus a \in W_{p_W}$ for all open sets a, because elements of $\mathbb{D}_{\mathsf{kh}} \mathfrak{X}$ are uniquely determined by the closed sets they contain. This follows immediately from the definitions and claim 3.41.B,

$$X \smallsetminus a \in W$$
 iff $p_W(\diamondsuit a) = \bot$ iff $X \smallsetminus a \in W_{p_W}$.

3.41.D Claim. The map $\varphi : \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \to \mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X}))$ is continuous.

Proof of claim. The opens of $\mathfrak{pt}(\mathbb{M}(\mathfrak{opn} \mathfrak{X}))$ are generated by $\square a = \{p \mid p(\square a) = \top\}$ and $\Diamond a = \{p \mid p(\Diamond a) = \top\}$, for $a \in \Omega \mathfrak{X}$. We have

$$\varphi^{-1}(\widehat{\Box a}) = \varphi^{-1}(\{p \mid p(\Box a) = \mathsf{T}\}) = \{W \in \mathbb{D}\mathfrak{X} \mid a \in W\} = \boxplus a$$

and similarly $\varphi^{-1}(\widehat{\diamond a}) = \otimes a$. Since $\boxplus a$ and $\otimes a$ are open in $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$, this proves continuity of φ .

We showed that φ is a bijective continuous function, hence a homeomorphism. This completes the proof of the theorem.

As announced before definition 3.39 we will now prove that \mathbb{D}_{kh} sends a compact Hausdorff space to a compact Hausdorff space.

3.42 Corollary. If \mathfrak{X} is a compact Hausdorff space, then so is $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$.

Proof. Since \mathfrak{X} is compact Hausdorff the frame $\mathfrak{opn}\mathfrak{X}$ is compact regular. By proposition 3.40 $\mathbb{M}(\mathfrak{opn}\mathfrak{X})$ is regular. It follows from lemma 3.38 that $\mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$ is compact hence by theorem 3.41 $\mathbb{M}(\mathfrak{opn}\mathfrak{X})$ is compact. So $\mathbb{M}(\mathfrak{opn}\mathfrak{X})$ is compact regular. Since $\mathfrak{pt}(\mathbb{M}(\mathfrak{opn}\mathfrak{X})) \cong \mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ the latter is compact Hausdorff by proposition 3.14. \Box

3.43 Remark. The proof of corollary 3.42 may appear incoherent, because of its many switches between the frame side and the topological side. However, it demonstrates precisely the power of a duality like the one given in theorem 3.41: regularity is easier to prove on the frame side, whereas for compactness the topological setting was more insightful.

It can be proven that \mathbb{M} preserves compactness on the frame side as well. This proof is given in section A.2 in the appendix.

Denote by \mathbb{M}_{kr} the restriction of \mathbb{M} to KRFrm. Theorem 3.41 yields a map $\mathbb{M}_{kr}(\mathfrak{opn} \mathfrak{X}) \rightarrow \mathfrak{opn}(\mathbb{D}_{kh}\mathfrak{X})$ for a compact Hausdorff space \mathfrak{X} given by

$$\mathbb{M}_{kr}(\mathfrak{opn} \mathfrak{X}) \xrightarrow{\mathfrak{opn} \circ \mathfrak{pt}} \mathfrak{opn}(\mathfrak{pt}(\mathbb{M}_{kr}(\mathfrak{opn} \mathfrak{X}))) \xrightarrow{\mathfrak{opn} \varphi} \mathfrak{opn}(\mathbb{D}_{kh} \mathfrak{X})$$

Unravelling the definitions shows that, on generators, it is given by $\Box a \mapsto \equiv a$ and $\Diamond a \mapsto \otimes a$.

3.44 Definition. For every compact Hausdorff space \mathfrak{X} define $\eta_{\mathfrak{X}} : \mathbb{M}_{kr}(\mathfrak{opn}\mathfrak{X}) \to \mathfrak{opn}(\mathbb{D}_{kh}\mathfrak{X})$ on generators by $\eta_{\mathfrak{X}}(\Box a) = \boxplus a$ and $\eta_{\mathfrak{X}}(\Diamond a) = \otimes a$. By the preceding discussion $\eta_{\mathfrak{X}}$ is a well-defined frame isomorphism.

It turns out that the maps $\eta_{\mathcal{X}}$ constitute a natural isomorphism.

3.45 Proposition. The collection $\eta = (\eta_X)_{X \in \mathsf{KH}_{\mathsf{aus}}}$ is a natural isomorphism.

Proof. The isomorphism part is proposition 3.41, so we only need to show naturality. That is, for any morphism $f: \mathfrak{X} \to \mathfrak{X}'$ in KHaus, the following diagram commutes,

(Since \mathfrak{opn} is a contravariant functor, the horizontal arrows are reversed.) For this, suppose $\Box a'$ is a generator of $\mathbb{M}_{kr}(\mathfrak{opn} \mathfrak{X}')$. Then

$$\mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}f) \circ \eta_{\mathfrak{X}'}(\Box a) = \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}f)(\boxplus a) \qquad \text{By definition 3.44}$$
$$= (\mathbb{D}_{\mathsf{kh}}f)^{-1}(\boxplus a) \qquad \text{By definition of opn}$$
$$= \boxplus f^{-1}(a) \qquad \text{By lemma 2.39}$$
$$= \eta_{\mathfrak{X}}(\Box f^{-1}(a)) \qquad \text{By definition 3.44}$$
$$= \eta_{\mathfrak{X}} \circ \mathbb{M}_{\mathsf{kr}}(f^{-1}(\Box a)) \qquad \text{By definition of } \mathbb{M}$$
$$= \eta_{\mathfrak{X}} \circ \mathbb{M}_{\mathsf{kr}}(\mathfrak{opn} f)(\Box a). \qquad \text{By definition of opn}$$

and by analogous reasoning $\Omega \mathbb{D}_{\mathsf{kh}} f \circ \eta_{\mathfrak{X}'}(\Diamond a) = \eta_{\mathfrak{X}} \circ \mathbb{M}_{\mathsf{kr}}(\mathfrak{opn} f)(\Diamond a)$. This proves that the diagram commutes. \Box

Applying proposition A.6 to the previous proposition yields the following corollary. **3.46 Corollary.** There is a dual equivalence

$$Alg(\mathbb{M}_{kr}) \equiv^{op} Coalg(\mathbb{D}_{kh}).$$

The functor \mathbb{D}_{kh} is the analog of the monotone functor from example 2.7 and gives rise to monotone logic on compact Hausdorff spaces. The fact that for both \mathbb{V}_{kh} and \mathbb{D}_{kh} the functor given on objects by taking a free frame modulo the axioms of the logic, suggests that KHaus is the most suitable base category for coalgebraic geometric logic.

The functor \mathbb{M} whose restriction is dual to \mathbb{D}_{kh} is defined for arbitrary frames, not just for compact regular frames. Similarly, the action of \mathbb{D}_{kh} on objects could be applied to any topological space, not just to compact Hausdorff spaces. (As we have seen in lemma 2.39 the morphisms, as they are defined now, do rely on the fact that we work over the category of compact Hausdorff spaces.) We have not been able to extend the duality from theorem 3.41 to either sober spaces and spatial frames, or compact sober spaces and compact spatial frames. Even more important, we have not found a proof that \mathbb{D} preserves sobriety or that \mathbb{M} preserves spatiality. Neither have we found counterexamples showing that it is false. We leave this as an interesting open question.

For the Vietoris functor and its dual more is known, for an overview see [58]. However, we still do not know whether the Vietoris functor preserves sobriety, even less so if there is a dual equivalence with an endofunctor on SFrm that holds in this generality.

A third example of a functor which induces logic on compact Hausdorff-coalgebras is that of the conditional functor. This functor gives rise to a duality similar to the duality from theorem 3.41 and will be treated in chapter 5.

3.4 BISIMULATIONS

In this section we give various notions of bisimulations between models for coalgebraic geometric logic and compare them. As in section 3.2 we use the symbol C to denote either of the categories Sob, KSob and KHaus.

3.47 Definition. Let \mathbb{T} be an endofunctor on C . An Aczel-Mendler bisimulation between two geometric \mathbb{T} -models is an Aczel-Mendler bisimulation B between the underlying \mathbb{T} -coalgebras such that for all $(x, x') \in B$ and $p \in \Phi$, $x \in V(p)$ iff $x' \in V'(p)$. We say that two states x and x' are Aczel-Mendler bisimilar and write $x \rightleftharpoons x'$ if there is an Aczel-Mendler bisimulation linking them.

The following lemma is an immediate consequence of proposition 3.20.

3.48 Lemma. Let \mathbb{T} be a functor on C , Λ a geometric modal signature for \mathbb{T} and x and x' states in two $\mathfrak{X}, \mathfrak{X}'$ two geometric \mathbb{T} -models. Then $x \Leftrightarrow x'$ implies $x \equiv_{\Lambda} x'$.

If Λ is characteristic and C = Sob or C = KSob, it follows from the previous lemma combined with theorems 3.22 and 3.34 that Aczel-Mendler bisimilarity implies behavioural equivalence. If moreover \mathbb{T} preserves weak pullbacks, the converse holds as well. The proof of this is similar to theorem 4.3 and the preceding discussion in [49].

However, we do not wish to make this assumption. For example, the Vietoris functor does not preserve weak pullbacks [9, Corollary 4.3]. Similarly to section 2.3 we define Λ -bisimulations for C-coalgebras. They are an adaptation of ideas in [6, 17]. Under some conditions on Λ , Λ -bisimilarity coincides with behavioural equivalence for any \mathbb{T} .

3.49 Definition. Let \mathbb{T} be an endofunctor on C , Λ a geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ geometric \mathbb{T} -models. A Λ -bisimulation between \mathfrak{X} and \mathfrak{X}' is a relation $B \subseteq \mathbb{U}\mathfrak{X} \times \mathbb{U}\mathfrak{X}'$ such that for all $(x, x') \in B$ and $p \in \Phi$ and all B-coherent pairs of opens $(a_i, a_i') \in \Omega\mathfrak{X} \times \Omega\mathfrak{X}'$ we have

$$x \in V(p)$$
 iff $x' \in V'(p)$

and

 $\gamma(x) \in \lambda_{\mathfrak{X}}(a_1, \dots, a_n) \quad \text{iff} \quad \gamma'(x') \in \lambda_{\mathfrak{X}'}(a_1', \dots, a_n').$ (3.5)

Two states are called Λ -bisimilar if there is a Λ -bisimulation linking them, notation: $x \rightleftharpoons x'$.

3.50 Remark. Let \mathbb{T} be an endofunctor on C , Λ a geometric modal signature for \mathbb{T} and $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ geometric \mathbb{T} -models. Suppose $B \subseteq \mathfrak{X} \times \mathfrak{X}'$ with the subspace topology is an object in C . Let $\pi : B \to \mathfrak{X}$ and $\pi' : B \to \mathfrak{X}'$ be the respective projections and

the pullback diagram of the cospan $(\mathfrak{opn}(\pi), \mathfrak{opn}(\pi'))$ in Frm. Then a pair $(a, a') \in \Omega \mathfrak{X} \times \Omega \mathfrak{X}'$ is *B*-coherent if and only if it is in \mathfrak{P} . If we unravel the definitions we find

that equation 3.5 holds if and only if for all $\lambda \in \Lambda$ the following diagram commutes:

$$\begin{array}{c|c} \mathfrak{opn}(\mathfrak{X})^{n} & \stackrel{(\overline{\pi})^{n}}{\longleftarrow} \mathcal{P}^{n} & \stackrel{(\overline{\pi}')^{n}}{\longrightarrow} \mathfrak{opn}(\mathfrak{X}')^{n} \\ \lambda_{\mathfrak{X}} & & \downarrow^{\lambda_{\mathfrak{X}'}} \\ \mathfrak{opn}(\mathfrak{T}\mathfrak{X}) & & \mathfrak{opn}(\mathfrak{T}\mathfrak{X}') \\ \mathfrak{opn}(\gamma) & & \downarrow^{\mathfrak{opn}(\gamma')} \\ \mathfrak{opn}(\mathfrak{X}) & \stackrel{\mathfrak{opn}(B)}{\longleftarrow} \mathfrak{opn}(B) & \stackrel{\mathfrak{opn}(\chi')}{\longleftarrow} \mathfrak{opn}(\mathfrak{X}') \end{array}$$
(3.6)

This point of view elucidates the similarity with Λ -bisimilarity for set-based coalgebras in [6].

As is to be expected, Λ -bisimilar states satisfy the same formulas.

3.51 Lemma. Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . Then $\Leftrightarrow_{\Lambda} \subseteq \equiv_{\Lambda}$.

Proof. Suppose $x \in \mathfrak{X}$ and $x' \in \mathfrak{X}'$ are Λ -bisimilar and B a Λ -bisimulation with xBx'. We will show that for all formulas φ we have $\mathfrak{X}, x \Vdash \varphi$ iff $\mathfrak{X}', x' \Vdash \varphi$ using induction on the complexity of the formula. The propositional case is by definition. If φ is a finite meet or an arbitrary join of formulas then the lemma is obvious. Suppose $\mathfrak{X}, x \Vdash \mathfrak{S}^{\lambda}(\varphi_1, \ldots, \varphi_n)$, then $\gamma(x) \in \lambda_{\mathfrak{X}}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}}, \ldots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}})$. We have $B[\llbracket \varphi_i \rrbracket^{\mathfrak{X}}] \subseteq \llbracket \varphi_i \rrbracket^{\mathfrak{X}'}$ by the induction hypothesis, so by definition of a Λ -bisimulation we find $\gamma'(x') \in \lambda_{\mathfrak{X}'}(\llbracket \varphi_1 \rrbracket^{\mathfrak{X}'}, \ldots, \llbracket \varphi_n \rrbracket^{\mathfrak{X}'})$ and hence $\mathfrak{X}', x' \Vdash \mathfrak{S}^{\lambda}(\varphi_1, \ldots, \varphi_n)$. The converse direction is proven symmetrically.

It turns out that the collection of Λ -bisimulations between two geometric \mathbb{T} -models forms a complete lattice.

3.52 Proposition. Let Λ be a geometric modal signature of a functor $\mathbb{T} : \mathsf{C} \to \mathsf{C}$ and let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be two geometric \mathbb{T} -models. The collection of Λ -bisimulations between \mathfrak{X} and \mathfrak{X}' forms a complete lattice.

Proof. It is obvious that the collection of Λ -bisimulations is a poset. We will show that this collection is closed under taking arbitrary unions; the result then follows from theorem 4.2 in [12].

Let I be some index set and for all $j \in J$ let B_j be Λ -bisimulations between \mathfrak{X} and \mathfrak{X}' and set $B = \bigcup_{i \in J} B_i$. We claim that B is a Λ -bisimulation.

Let (a_i, a'_i) be a *B*-coherent pairs of opens. Suppose xBx' and $\gamma(x) \in \lambda_{\mathfrak{X}}(a_1, \ldots, a_n)$. Then there is $j \in J$ with xB_jx' . As $B_j[a_i] \subseteq B[a_i] \subseteq a'_i$ and $B_j^{-1}[a'_i] \subseteq B^{-1}[a'_i] \subseteq a_i$, all *B*-coherent pairs (a_i, a'_i) are also B_j -coherent. As B_j is a Λ -bisimulation we have $\gamma'(x') \in \lambda_{\mathfrak{X}'}(a'_1, \ldots, a'_n)$.

Every Aczel-Mendler bisimulation is a Λ -bisimulation.

3.53 Proposition. Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . Then $\Leftrightarrow \subseteq \Leftrightarrow_{\Lambda}$.

Proof. Suppose B is an Aczel-Mendler bisimulation and let β be the map that turns B into a coalgebra, then the following diagram commutes:

$$\begin{array}{cccc} \mathbf{x} & & & & \\ \mathbf{x} & & & \\ \gamma & & & \\ \gamma & & & \\ \mathbf{x} & & \\ \mathbf{x} & & \\ \mathbf{x} & & \\ \mathbf{x} & \\ \mathbf{$$

We will show that B is a Λ -bisimulation. By definition $x \in V(p)$ iff $x' \in V'(p)$ whenever xBx'. We prove the forth condition from definition 3.49. Let $\lambda \in \Lambda$ and $(x,x') \in B$. Suppose $(a_1,a'_1), \ldots, (a_n,a'_n)$ are B-coherent pairs of opens and $\gamma(x) \in$ $\lambda_{\mathfrak{X}}(a_1,\ldots,a_n)$. Then we have

$$\beta(x, x') \in (\mathbb{T}\pi)^{-1}(\lambda_{\mathfrak{X}}(a_{1}, \dots, a_{n}))$$
(follows from (3.7))

$$= \lambda_{B}(\pi^{-1}(a_{1}), \dots, \pi^{-1}(a_{n_{\lambda}}))$$
(naturality of λ)

$$\subseteq \lambda_{B}((\pi')^{-1} \circ \pi'[\pi^{-1}(a_{1})], \dots, (\pi')^{-1} \circ \pi'[\pi^{-1}(a_{n})])$$
(monotonicity of λ)

$$= \lambda_{B}((\pi')^{-1}(B[a_{1}]), \dots, (\pi')^{-1}(B[a_{n}]))$$
(B[a] = $\pi_{2} \circ \pi_{1}^{-1}(a)$)

$$\subseteq \lambda_{B}((\pi')^{-1}(a'_{1}), \dots, (\pi')^{-1}(a'_{n}))$$
(monotonicity of λ)

$$= (\mathbb{T}\pi')^{-1}(\lambda_{\mathfrak{X}'}(a'_{1}, \dots, a'_{n})).$$
(naturality of λ)

Therefore

$$\gamma'(x') = (\mathbb{T}\pi')(\beta(x,x')) \in \lambda_{\mathfrak{X}'}(a'_1,\ldots,a'_{n_{\lambda}})$$

 \triangleleft

as desired.

It follows from lemma 3.51 and theorem 3.22 that, provided Λ is a characteristic geometric modal signature for an endofunctor \mathbb{T} on Sob or KSob, Λ -bisimilarity implies behavioural equivalence. In order to prove a converse statement we need a slight strengthening of the definition of open predicate liftings. We define the notion of a strong open predicate lifting in the next definition and subsequently prove that for strong monotone sets of predicate liftings, behavioural equivalence implies Λ -bisimilarity.

3.54 Definition. Let \mathbb{T} be an endofunctor on C. A strong open predicate lifting for \mathbb{T} is a natural transformation

$$\lambda: \check{\mathbb{P}}^n \circ \mathbb{U} \to \check{\mathbb{P}} \circ \mathbb{U} \circ \mathbb{T}$$

such that for objects \mathfrak{X} in C and $a_1, \ldots, a_n \in \Omega \mathfrak{X}$ the set $\lambda_{\mathfrak{X}}(a_1, \ldots, a_n)$ is open in $\mathbb{T}\mathfrak{X}$. A strong open predicate lifting λ is said to be **monotone in its** *i*-th argument if for all (not necessarily open) subsets $a_1, \ldots, a_n, \delta \in \mathbb{P}(\mathbb{U}\mathfrak{X})$ we have $\lambda_{\mathfrak{X}}(a_1, \ldots, a_i, \ldots, a_n) \subseteq$ $\lambda_{\mathfrak{X}}(a_1, \ldots, a_i \cup \delta, \ldots, a_n)$ and it is called **monotone** if it is monotone in every argument. The dual of a strong open predicate lifting λ is λ^{∂} defined by

$$\lambda_{\mathfrak{X}}^{\partial}(a_1,\ldots,a_n) = \mathfrak{X} \smallsetminus \lambda_{\mathfrak{X}}(\mathfrak{X} \smallsetminus a_1,\ldots,\mathfrak{X} \smallsetminus a_n).$$

A collection of strong open predicate liftings Λ is called a strong geometric modal signature. A strong geometric modal signature is **closed under duals** if $\lambda \in \Lambda$ implies $\lambda^{\partial} \in \Lambda$ and **characteristic** if for every $\boldsymbol{\mathcal{X}}$ in C the collection

$$\{\lambda_{\mathbf{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda \text{ n-ary}, a_i \in \Omega \mathbf{X}\}$$

forms a sub-base for the topology on $\mathbb{T}\mathfrak{X}$.

It follows immediately from the definitions that every strong open predicate lifting restricts to a "normal" open predicate lifting. We call an open predicate lifting λ strong if there exists a strong open predicate liftings that restricts to λ and strong monotone if there exists a monotone strong open predicate lifting that restricts to it. A geometric modal signature is called strong monotone if every open predicate lifting in it is strong monotone.

3.55 Example. The box- and diamond-lifting for both basic modal logic and monotone modal logic are strong monotone open predicate liftings. \triangleleft

The proof of the following proposition is similar to the proof of proposition 2.34 and is inspired by the proof of theorem 4.1 in [16].

3.56 Proposition. Let \mathbb{T} be an endofunctor on C and Λ a strong monotone geometric modal signature for \mathbb{T} . Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be two \mathbb{T} -models. Then $\simeq_{\mathsf{Coalg}}(\mathbb{T}) \subseteq \cong_{\Lambda}$.

Proof. Suppose x and x' are behaviourally equivalent. Then there is some geometric \mathbb{T} -model $\mathfrak{U} = (\mathfrak{U}, \nu, V_{\mathfrak{U}})$ and there are \mathbb{T} -model morphisms $f : \mathfrak{X} \to \mathfrak{U}$ and $f' : \mathfrak{X}' \to \mathfrak{U}$ such that f(x) = f'(x'). We will define a Λ -bisimulation B linking x and x'.

Let B be the pullback of f and f' in Top,

$$B = \{(u, u') \in X \times X' \mid f(u) = f'(u')\}.$$

Then clearly xBx'. It follows from proposition 3.20 that u and u' satisfy precisely the same formulas whenever $(u, u') \in B$.

Suppose $\lambda \in \Lambda$ is *n*-ary and for $1 \leq i \leq n$ let (a_i, a'_i) be a *B*-coherent pair of opens. Suppose uBu' and $\gamma(u) \in \lambda_{\mathfrak{X}}(a_1, \ldots, a_n)$. We will show that $\gamma'(u') \in \lambda_{\mathfrak{X}'}(a'_1, \ldots, a'_n)$, the converse direction is similar. By monotonicity and naturality of λ we obtain

$$\gamma(u) \in \lambda_{\mathfrak{X}}(a_1, \dots, a_n) \subseteq \lambda_{\mathfrak{X}}(f^{-1}(f[a_1]), \dots, f^{-1}(f[a_n])) = (\mathbb{T}f)^{-1}(\lambda_{\mathfrak{Y}}(f[a_1], \dots, f[a_n])),$$

so $(\mathbb{T}f)(\gamma(u)) \in \lambda_{\mathcal{Y}}(f[a_1], \dots, f[a_n])$. (Note that the $f[a_i]$ need not be open in \mathcal{Y} , but $\lambda_{\mathcal{Y}}(f[a_1], \dots, f[a_n])$ is defined because λ is assumed to be strong.) Since f and f' are coalgebra morphisms and f(u) = f'(u') we have $(\mathbb{T}f)(\gamma(u)) = \delta(f(u)) = \delta(f'(u')) = (\mathbb{T}f')(\gamma'(u'))$. Coherence of (a_i, a'_i) and monotonicity and naturality of λ yield

$$\begin{aligned} \gamma'(u') &\in (\mathbb{T}f')^{-1}(\lambda y(f[a_1], \dots, f[a_n])) \\ &= \lambda_{\mathfrak{X}'}((f')^{-1}(f[a_1]), \dots, (f')^{-1}(f[a_n])) \\ &= \lambda_{\mathfrak{X}'}(B[a_1], \dots, B[a_n]) \end{aligned} \qquad (naturality of \lambda) \\ &\subseteq \lambda_{\mathfrak{X}'}(a'_1, \dots, a'_n). \end{aligned}$$

This proves the proposition.

Let \mathbb{T} be an endofunctor on C and Λ a geometric modal signature for \mathbb{T} . The following diagram summarises the results from theorems 3.22, 3.34 and 3.34, lemma 3.51 and proposition 3.56. The arrows indicate that one form of equivalence implies the other. The numbers indicate that the implication is under a certain condition.



Here (1) holds if \mathbb{T} preserves weak pullbacks, (2) is true when \mathbb{T} is a functor on Sob or a functor on KSob, and Λ is characteristic, and (3) holds when Λ is strong. This yields the following theorem.

3.57 Theorem. Let \mathbb{T} be an endofunctor on Sob and Λ a characteristic strong monotone geometric modal signature.

(i) If x and x' are two states in two geometric \mathbb{T} -models, then

 $x \cong_{\Lambda} x'$ iff $x \equiv_{\Lambda} x'$ iff $x \simeq_{\mathsf{Mod}(\mathbb{T})} x'$.

(ii) If \mathbb{T} restricts to the endofunctor $\mathbb{T}_{\mathsf{KSob}}$ on KSob and x, x' are two states in two geometric $\mathbb{T}_{\mathsf{KSob}}$ -models, then

$$x \simeq_{\Lambda} x'$$
 iff $x \equiv_{\Lambda} x'$ iff $x \simeq_{\mathsf{Mod}(\mathbb{T}_{\mathsf{KSob}})} x'$.

(iii) If \mathbb{T} restricts to the endofunctor $\mathbb{T}_{\mathsf{KHaus}}$ on KHaus and x and x' are two states in two geometric $\mathbb{T}_{\mathsf{KHaus}}$ -models, then

$$x \Leftrightarrow_{\Lambda} x' \quad iff \quad x \equiv_{\Lambda} x'.$$

3.58 Remark. A priori, we cannot apply the previous theorem to the monotone functor \mathbb{D}_{kh} . However, we will see in example 4.9 that \mathbb{D}_{kh} is the restriction of a certain functor on Sob.

From the point of view of bisimulations, the category of sober spaces and continuous maps seems to be an appropriate base category for coalgebraic geometric logic. However, we saw in section 3.3 that looking at concrete examples suggests that the category of compact Hausdorff spaces is a natural choice of base category. We leave it as an interesting direction for further research to investigate how these concrete examples of functors fit in the general theory we have developed.

Lifting functors

This chapter is devoted to lifting endofunctors from Set to the categories Sob and Stone. A functor \mathbb{T} together with a set of predicate liftings Λ for \mathbb{T} induces a functor on Sob and on Stone, each accompanied by a set of open respectively clopen predicate liftings. This will give rise to analogs of set-based frames. For example, the functor \mathbb{D}_{st} (from definition 2.38) corresponding to descriptive monotone frames is the lift of the functor \mathbb{D} (see example 2.7) with respect to a certain set of predicate liftings.

Section 4.1 presents a way of lifting set functors with respect to a set of predicate liftings Λ to functors on Sob. This is inspired by the method of lifting a set functor to a functor on Stone from [36], which is presented in section 4.2.

A different method of lifting a set functor to a functor on Frm has been developed in [57]. This trivially yields a lifted functor $\text{Sob} \rightarrow \text{Sob}$ by using the duality between frames and topological spaces (see section 3.1). The two differences of this way of lifting with the method that we present in section 4.1 is that is *does not* depend on a set of predicate liftings, but it *does* put restrictions on the functor \mathbb{T} that is being lifted. We suspect a connection between these two ways of lifting a set functor to a sober functor, and leave this as an interesting open question.

In section 4.2 and 4.3 set functors are lifted to Stone functors in two ways. The first way is a reformulation of the method in [36] and is similar to the procedure in section 4.1. It relies on a given collection of predicate liftings Λ . The second way (lift via pro-completions) only works for a certain class of set functors, but does not depend on a collection of predicate liftings. We prove that for such a T and a suitable choice of Λ the two ways of lifting coincide on objects (theorem 4.24) thus partially answering a question posed in the conclusion of [36].

4.1 FROM Set TO Sob

In this section we lift a functor $\mathbb{T} : \mathsf{Set} \to \mathsf{Set}$ together with a set of predicate liftings Λ for \mathbb{T} to a functor $\ddot{\mathbb{T}}_{\Lambda} : \mathsf{Top} \to \mathsf{Top}$ which restricts to $\ddot{\mathbb{T}}_{\Lambda} : \mathsf{Sob} \to \mathsf{Sob}$. To define the action of $\ddot{\mathbb{T}}_{\Lambda}$ on objects, for a topological space \mathfrak{X} we take the following steps:

- Step 1. Construct a frame of the images of predicate liftings applied to the open sets of \mathfrak{X} (viewed simply as subsets of $\mathbb{T}(\mathbb{U}\mathfrak{X})$), this is $\dot{\mathbb{F}}_{\Lambda}\mathfrak{X}$;
- Step 2. Quotient $\dot{\mathbb{F}}_{\Lambda} \mathfrak{X}$ with a suitable relation that ensures $\bigvee_{\delta \in B} \lambda(\delta) = \lambda(\bigvee B)$ whenever λ is monotone;
- Step 3. Employ the functor $\mathfrak{pt}: \mathsf{Frm} \to \mathsf{Sob}$ to obtain a sober topological space.

This is the content of definitions 4.1, 4.3 and 4.5. It is an adaptation of section 4 in [36].

Recall that \mathbb{U} is the forgetful functor which sends a topological space to its underlying set, and \mathbb{Q} is the contravariant functor sending a set to its powerset Boolean algebra.

4.1 Definition. Let \mathbb{T} : Set \rightarrow Set be a functor and Λ a collection of predicate liftings for \mathbb{T} . We define a contravariant functor $\dot{\mathbb{F}}_{\Lambda}$: Top \rightarrow Frm. For a topological space \mathfrak{X} define $\dot{\mathbb{F}}_{\Lambda}\mathfrak{X}$ to be the subframe of $\mathbb{Q}(\mathbb{T}(\mathbb{U}\mathfrak{X}))$ generated by the set

$$\{\lambda_{\mathbb{U}\mathfrak{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda \text{ n-ary, } a_1,\ldots,a_n \in \Omega\mathfrak{X}\}.$$

That is, we close this set under finite intersections and arbitrary unions in $\mathbb{Q}(\mathbb{T}(\mathbb{U}\mathfrak{X}))$. For a continuous function $f: \mathfrak{X} \to \mathfrak{X}'$ let $\dot{\mathbb{F}}_{\Lambda}f: \dot{\mathbb{F}}_{\Lambda}\mathfrak{X}' \to \dot{\mathbb{F}}_{\Lambda}\mathfrak{X}$ be the restriction of $\mathbb{Q}(\mathbb{T}(\mathbb{U}f))$ to $\dot{\mathbb{F}}_{\Lambda}\mathfrak{X}'$.

4.2 Lemma. The map $\dot{\mathbb{F}}_{\Lambda}$ defines a contravariant functor.

Proof. We need to show that $\dot{\mathbb{F}}_{\Lambda}$ is well defined on morphisms and that it is functoral. To show that the action of $\dot{\mathbb{F}}_{\Lambda}$ on morphisms is well-defined, it suffices to show that $(\dot{\mathbb{F}}_{\Lambda}f)(\lambda_{\mathbb{UX}}(a'_1,\ldots,a'_n)) \in \dot{\mathbb{F}}_{\Lambda}(\mathfrak{X})$ for all generators $\lambda_{\mathbb{UX}}(a'_1,\ldots,a'_n)$ of $\dot{\mathbb{F}}_{\Lambda}\mathfrak{X}'$, because frame homomorphisms preserve finite meets and all joins. This holds by naturality of λ :

$$(\dot{\mathbb{F}}_{\Lambda}f)(\lambda_{\mathbb{U}\mathfrak{X}'}(a_1,\ldots,a_n)) = (\mathbb{T}f)^{-1}(\lambda_{\mathbb{U}\mathfrak{X}'}(a_1,\ldots,a_n)) = \lambda_{\mathbb{U}\mathfrak{X}}(f^{-1}(a_1),\ldots,f^{-1}(a_n)).$$

By continuity of f we have $f^{-1}(a_i) \in \Omega \mathfrak{X}$ so the latter is indeed in $\dot{\mathbb{F}}_{\Lambda} \mathfrak{X}$. Functoriality of $\dot{\mathbb{F}}_{\Lambda}$ follows from functoriality of $\mathbb{Q} \circ \mathbb{T} \circ \mathbb{U}$.

4.3 Definition. Let Λ be a collection of predicate liftings for a set functor \mathbb{T} and let \mathfrak{X} be a topological space. Let $\ddot{\mathbb{F}}_{\Lambda}\mathfrak{X}$ be the quotient of $\dot{\mathbb{F}}_{\Lambda}\mathfrak{X}$ with respect to the congruence ~ generated by

$$\bigvee_{\boldsymbol{b}\in B}\lambda(a_1,\ldots,a_{i-1},\boldsymbol{b},a_{i+1},\ldots,a_n)\sim\lambda(a_1,\ldots,a_{i-1},\boldsymbol{\forall}B,a_{i+1},\ldots,a_n)$$

for all $a_i \in \Omega \mathfrak{X}$, $B \subseteq \Omega \mathfrak{X}$ directed, and $\lambda \in \Lambda$ monotone in its *i*-th argument. Write $q_{\mathfrak{X}} : \dot{\mathbb{F}}_{\Lambda} \mathfrak{X} \to \ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}$ for the quotient map and [x] for the equivalence class in $\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}$ of an element $x \in \dot{\mathbb{F}}_{\Lambda} \mathfrak{X}$. For a continuous function $f : \mathfrak{X} \to \mathfrak{X}'$ define $\ddot{\mathbb{F}}_{\Lambda} f : \ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}' \to \ddot{\mathbb{F}}_{\Lambda} \mathfrak{X} : [\lambda_{\mathbb{U}\mathfrak{X}}(a_1, \ldots, a_n)] \mapsto [\dot{\mathbb{F}}_{\Lambda}(\lambda_{\mathbb{U}\mathfrak{X}}(a_1, \ldots, a_n))].$

4.4 Lemma. The assignment $\ddot{\mathbb{F}}_{\Lambda}$ defines a functor.

Proof. We need to prove functoriality of \mathbb{F}_{Λ} and that $\mathbb{F}_{\Lambda}f$ is well defined for every continuous map $f: \mathfrak{X} \to \mathfrak{X}'$.

In order to show that $\ddot{\mathbb{F}}_{\Lambda}$ is well defined, it suffices to show that $\ddot{\mathbb{F}}_{\Lambda}f$ is invariant under the congruence ~. If $f: \mathfrak{X} \to \mathfrak{X}'$ is a continuous, then

$$\begin{aligned} \bigvee_{\delta \in B} (\dot{\mathbb{F}}_{\Lambda} f) (\lambda_{\mathbb{U} \mathfrak{X}'}(a_{1}, \dots, a_{i-1}, \delta, a_{i+1}, \dots, a_{n})) \\ &= \bigvee_{\delta \in B} (\mathbb{T} f)^{-1} (\lambda_{\mathbb{U} \mathfrak{X}'}(a_{1}, \dots, a_{i-1}, \delta, a_{i+1}, \dots, a_{n})) \\ &= \bigvee_{\delta \in B} \lambda_{\mathbb{U} \mathfrak{X}'}(f^{-1}(a_{1}), \dots, f^{-1}(a_{i-1}), f^{-1}(\delta), f^{-1}(a_{i+1}), \dots, f^{-1}(a_{n})) \\ &\sim \lambda_{\mathbb{U} \mathfrak{X}'}(f^{-1}(a_{1}), \dots, f^{-1}(a_{i-1}), f^{-1}(\sqrt{B}), f^{-1}(a_{i+1}), \dots, f^{-1}(a_{n})) \\ &= \dot{\mathbb{F}}_{\Lambda} f(\lambda_{\mathbb{U} \mathfrak{X}'}(a_{1}, \dots, a_{i-1}, \sqrt{B}, a_{i+1}, \dots, a_{n})) \end{aligned}$$

so $\ddot{\mathbb{F}}_{\Lambda}f$ is indeed invariant under the congruence. In the ~-step, we use that $\{f^{-1}(\boldsymbol{b}) \mid \boldsymbol{b} \in B\}$ is directed in $\Omega \mathfrak{X}$. Functoriality of $\ddot{\mathbb{F}}_{\Lambda}f$ follows from functoriality of $\mathbb{Q} \circ \mathbb{T} \circ \mathbb{U}$. So $\ddot{\mathbb{F}}_{\Lambda}$: Top \rightarrow Frm is a functor.

We are now ready to define the sober Kupke-Kurz-Pattinson lift to Sob for a functor on Set.

4.5 Definition. Define the sober Kupke-Kurz-Pattinson lift (KKP lift for short) of \mathbb{T} with respect to Λ to be the functor

$$\ddot{\mathbb{T}}_{\Lambda} = \mathfrak{pt} \circ \ddot{\mathbb{F}}_{\Lambda}.$$

This is a functor $\mathsf{Top} \to \mathsf{Top}$ and because \mathfrak{pt} lands in Sob it restricts to an endofunctor on Sob. We shall view $\ddot{\mathbb{T}}_{\Lambda}$ as an endofunctor on Sob.

The next definition and lemma describe how to lift a predicate lifting to an open predicate lifting. Recall that \mathbb{Z} is the forgetful functor which sends a frame to its underlying set.

4.6 Definition. Let Λ be a collection of predicate liftings for a set functor \mathbb{T} . A predicate lifting $\lambda : \mathbb{P}^n \to \mathbb{P} \circ \mathbb{T}$ in λ induces an open predicate lifting $\ddot{\lambda} : \Omega \to \Omega \circ \ddot{\mathbb{T}}$ for $\ddot{\mathbb{T}}$ via

$$\Omega^{n} \mathfrak{X} \xrightarrow{\lambda_{U\mathfrak{X}}} \mathbb{Z}(\dot{\mathbb{F}}_{\Lambda} \mathfrak{X}) \xrightarrow{q_{\mathfrak{X}}} \mathbb{Z}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}) \xrightarrow{\mathbb{Z}k_{\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}}} \Omega(\mathfrak{pt}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X})) = \Omega(\ddot{\mathbb{T}} \mathfrak{X}).$$

By $\lambda_{\mathbb{U}\mathfrak{X}}$ actually means the restriction of $\lambda_{\mathbb{U}\mathfrak{X}}$ to $\Omega^n\mathfrak{X} \subseteq \check{\mathbb{P}}(\mathbb{U}\mathfrak{X})$. The map $k_{\mathbb{F}\mathfrak{X}}$ is the frame homomorphism given by $a \mapsto \{p \in \mathfrak{pt}(\mathbb{F}_\Lambda\mathfrak{X}) \mid p(a) = 1\}$. Let $\ddot{\Lambda} = \{\ddot{\lambda} \mid \lambda \in \Lambda\}$ be the collection of lifted predicate liftings. Then $\ddot{\Lambda}$ is a geometric modal signature for $\ddot{\mathbb{T}}_{\Lambda}$. \triangleleft

4.7 Lemma. The assignment $\ddot{\lambda}$ is a natural transformation.

Proof. Let $f: \mathfrak{X} \to \mathfrak{X}'$ be a continuous function, then the following diagram commutes in Set,

$$\begin{array}{cccc} \Omega^{n} \mathfrak{X}' & \xrightarrow{\lambda_{\mathbb{U}\mathfrak{X}'}} & \mathbb{Z}(\dot{\mathbb{F}}_{\Lambda} \mathfrak{X}') & \xrightarrow{\mathbb{Z}q_{\mathfrak{X}'}} & \mathbb{Z}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}') & \xrightarrow{\mathbb{Z}k_{\mathbb{F}_{\Lambda}\mathfrak{X}'}} & \Omega(\mathfrak{pt}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}')) \\ (f^{-1})^{n} & & \downarrow f^{-1} & & \downarrow f^{-1} & & \downarrow \Omega(\mathfrak{pt}(f^{-1})) \\ \Omega^{n} \mathfrak{X} & \xrightarrow{\lambda_{\mathbb{U}\mathfrak{X}}} & \mathbb{Z}(\dot{\mathbb{F}}_{\Lambda} \mathfrak{X}) & \xrightarrow{\mathbb{Z}q_{\mathfrak{X}}} & \mathbb{Z}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}) & \xrightarrow{\mathbb{Z}k_{\mathbb{F}_{\Lambda}\mathfrak{X}}} & \Omega(\mathfrak{pt}(\mathbb{F}_{\Lambda}\mathfrak{X})) \end{array}$$

Commutativity of the left square follows from naturality of λ , commutativity of the middle square follows from the proof of lemma 4.4 and commutativity of the right square can be seen as follows: let $a'_1, \ldots, a'_n \in \Omega \mathfrak{X}$, then

$$\Omega(\mathfrak{pt}(f^{-1})) \circ \mathbb{Z}k_{\mathbb{F}_{\Lambda}\mathfrak{X}'}(\lambda_{\mathbb{U}\mathfrak{X}'}(a_{1}',\ldots,a_{n}')) = \{q \in \mathfrak{pt}(\mathbb{F}_{\Lambda}\mathfrak{X}) \mid q \circ f^{-1}(\lambda_{\mathbb{U}\mathfrak{X}'}(a_{1}',\ldots,a_{n}')) = 1\}$$
$$= \mathbb{Z}k_{\mathbb{F}_{\Lambda}\mathfrak{X}}(f^{-1}(\lambda_{\mathbb{U}\mathfrak{X}'}(a_{1}',\ldots,a_{n}'))).$$

So $\hat{\lambda}$ is an open predicate lifting.

The nature of the definitions of $\ddot{\mathbb{T}}$ and $\ddot{\Lambda}$ yields the following desirable result.

4.8 Proposition. Let $\mathbb{T} : \mathsf{Set} \to \mathsf{Set}$ be a functor and Λ a collection of predicate liftings for \mathbb{T} . Then $\ddot{\Lambda}$ is characteristic for $\ddot{\mathbb{T}}_{\Lambda}$.

Proof. Let \mathfrak{X} be a sober space. We need to show that the collection

$$\{\lambda(a_1,\ldots,a_n) \mid \lambda \in \Lambda \text{ } n\text{-ary}, a_i \in \Omega \mathfrak{X}\}$$
(4.1)

forms a subbase for the topology on $\ddot{\mathbb{T}}_{\Lambda} \mathfrak{X}$. An arbitrary open set of $\ddot{\mathbb{T}}_{\Lambda} \mathfrak{X}$ is of the form $\widetilde{\mathfrak{X}} = \{p \in \mathfrak{pt}(\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}) \mid p(x) = 1\}$, for $x \in \ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}$. An arbitrary element of $\ddot{\mathbb{F}}_{\Lambda} \mathfrak{X}$ is the equivalence class of an arbitrary union of finite intersections of elements of the form $\{\lambda_{\mathfrak{U}\mathfrak{X}}(a_1,\ldots,a_n), \text{ for } \lambda \in \Lambda \text{ and } a_1,\ldots,a_n \in \Omega \mathfrak{X}.$ So we may write

$$x = \bigcup_{i \in I} (\bigcap_{j \in J_i} [\lambda_{\mathbb{U}\mathcal{X}}^{i,j}(a_1^{i,j}, \dots, a_{n_{i,j}}^{i,j})])$$

for some possibly index set I, finite index sets J_i , $\lambda^{i,j} \in \Lambda$ and open sets $a_k^{i,j} \in \Omega \mathfrak{X}$. Unraveling the definitions gives

$$\widetilde{x} = \bigcup_{i \in I} \left(\bigcap_{j \in J_i} \left[\lambda_{\mathbb{UX}}^{i,j}(a_1^{i,j}, \dots, a_{n_{i,j}}^{i,j}) \right] \right)$$
$$= \bigcup_{i \in I} \left(\bigcap_{j \in J_i} \ddot{\lambda}_{\mathcal{X}}^{i,j}(a_1^{i,j}, \dots, a_{n_{i,j}}^{i,j}) \right)$$

This shows that the open sets in (4.1) indeed form a subbase for the open sets of $\ddot{\mathbb{T}}_{\Lambda} \mathfrak{X}$. \Box

Let us put our theory to action to (re)obtain the monotone functor on KHaus from definition 2.38.

4.9 Example (The monotone functor). Recall the set functor \mathbb{D} from example 2.7: $\mathbb{D}: X \to \{W \in \check{\mathbb{P}P}X \mid W \text{ is up-closed under inclusion order}\}$. The box and diamond are given by the predicate liftings $\lambda^{\Box}, \lambda^{\diamondsuit}: \check{\mathbb{P}} \to \check{\mathbb{P}} \circ \mathbb{D}$ defined by

$$\lambda_X^{\square}(a) \coloneqq \{ W \in \mathbb{D}X \mid a \in W \}, \quad \lambda_X^{\diamondsuit}(a) \coloneqq \{ W \in \mathbb{D}X \mid (X \smallsetminus a) \notin W \},$$

where $X \in \text{Set.}$ Furthermore recall from definition 3.37 that for a compact Hausdorff space \mathfrak{X} the space $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ is the subset of $\mathbb{D}(\mathbb{U}\mathfrak{X})$ of collections of sets W satisfying for all $u \subseteq \mathbb{U}\mathfrak{X}$ that $u \in W$ iff there exists a closed $c \subseteq u$ such that every open superset of cis in W. So $\mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \subseteq \mathbb{D}(\mathbb{U}\mathfrak{X})$. The set $\mathbb{D}_{\mathsf{kh}}\mathfrak{X}$ is topologised by the subbase

$$\boxplus a := \{ W \in \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \mid a \in W \}, \quad \& a := \{ W \in \mathbb{D}_{\mathsf{kh}} \mathfrak{X} \mid (\mathfrak{X} \setminus a) \notin W \}.$$

By theorem 3.41 the functor $\mathbb{M} : \mathsf{Frm} \to \mathsf{Frm}$ from definition 3.39 is such that $\mathbb{M}(\mathfrak{opn} \mathfrak{X}) \cong \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$ whenever \mathfrak{X} is a compact Hausdorff space.

Let \mathfrak{X} be a compact Hausdorff space. We claim that $\mathbb{F}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}}\mathfrak{X} = \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$. Define a map $\varphi : \mathbb{M}(\mathfrak{opn}\mathfrak{X}) \to \mathbb{F}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}}\mathfrak{X}$ on generators by $\square a \mapsto [\lambda^{\square}(a)]$ and $\diamondsuit a \mapsto [\lambda^{\diamondsuit}(a)]$. This is well-defined because the $[\lambda^{\square}(a)], [\lambda^{\diamondsuit}(a)]$ satisfy relations (M1) – (M6) from definition 3.39 and it is surjective because the image of φ contains the generators of $\mathbb{F}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}}\mathfrak{X}$.

So we only need to show injectivity of φ . Our strategy to prove this is to define a map $\psi : \ddot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamond}\}} \mathfrak{X} \to \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$ and show that it is inverse to φ on the level of sets. Since a set-theoretic inverse suffices we do not need to prove that ψ is a homomorphism; we just want it to be well-defined. Instead of defining $\psi : \ddot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamond}\}} \to \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$ directly, we will give a well-defined map $\psi' : \dot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamond}\}} \mathfrak{X} \to \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$ whose kernel contains the kernel of the quotient map $q_{\mathfrak{X}} : \dot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamond}\}} \to \ddot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamond}\}}\mathfrak{X}$. This then yields the map ψ we require. In a diagram:

$$\dot{\mathbb{F}}_{\{\lambda^{\Box},\lambda^{\diamond}\}} \mathfrak{X} \xrightarrow{\psi'} \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \\
\overset{q_{\mathcal{X}}}{\overset{\psi}{\mathbb{F}}_{\{\lambda^{\Box},\lambda^{\diamond}\}}} \mathfrak{X} \qquad (4.2)$$

Define $\psi' : \dot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamondsuit}\}} \mathfrak{X} \to \mathbb{M}(\mathfrak{opn}\mathfrak{X})$ on generators by $\lambda^{\Box}(a) \mapsto \boxplus a$ and $\lambda^{\diamondsuit}(a) \mapsto \otimes a$. In order to show that this assignments yields a well-defined map (hence extends to a frame homomorphism by remark 3.3) we need to show that the presentation of an element in $\dot{\mathbb{F}}_{\{\lambda^{\Box}, \lambda^{\diamondsuit}\}}\mathfrak{X}$ does not affect its image under ψ' . That is, if

$$\bigcup_{i \in I} \left(\bigcap_{j \in J_i} \lambda^{\square}(a_{i,j}) \cap \bigcap_{j' \in J'_i} \lambda^{\diamondsuit}(a_{i,j'}) \right) = \bigcup_{k \in K} \left(\bigcap_{\ell \in L_k} \lambda^{\square}(a_{k,\ell}) \cap \bigcap_{\ell' \in L'_k} \lambda^{\diamondsuit}(a_{k,\ell'}) \right), \tag{4.3}$$

where J_i, J'_i, L_k and L'_k are finite index sets, then

$$\bigcup_{i\in I} \Big(\bigcap_{j\in J} \psi'(\lambda^{\Box}(a_{i,j})) \cap \bigcap_{j'\in J'} \psi'(\lambda^{\diamond}(a_{i,j'}))\Big) = \bigcup_{k\in K} \Big(\bigcap_{\ell\in L} \psi'(\lambda^{\Box}(a_{k,\ell})) \cap \bigcap_{\ell'\in L'} \psi'(\lambda^{\diamond}(a_{k,\ell'}))\Big).$$

As stated we have $\mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \subseteq \mathbb{D}(\mathbb{U}\mathfrak{X})$. Observe

$$\psi'(\lambda^{\square}(a)) = \boxplus a = \{ W \in \mathbb{D}(\mathbb{U}\mathfrak{X}) \mid a \in W \} \cap \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) = \lambda^{\square}(a) \cap \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}).$$

and similarly $\psi'(\lambda^{\diamond}(a)) = \lambda^{\diamond}(a) \cap \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})$. Suppose the identity in (4.3) holds, then we have

$$\begin{split} &\bigcup_{i\in I} \Big(\bigcap_{j\in J} \psi'(\lambda^{\Box}(a_{i,j})) \cap \bigcap_{j'\in J'} \psi'(\lambda^{\diamond}(a_{i,j'})) \Big) \\ &= \bigcup_{i\in I} \Big(\bigcap_{j\in J} (\lambda^{\Box}(a_{i,j}) \cap \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})) \cap \bigcap_{j'\in J'} (\lambda^{\diamond}(a_{i,j'}) \cap \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X})) \Big) \\ &= \bigcup_{i\in I} \Big(\mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \cap \bigcap_{j\in J} \lambda^{\Box}(a_{i,j}) \cap \bigcap_{j'\in J'} \lambda^{\diamond}(a_{i,j'}) \Big) \\ &= \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \cap \bigcup_{i\in I} \Big(\bigcap_{j\in J} \lambda^{\Box}(a_{i,j}) \cap \bigcap_{j'\in J'} \lambda^{\diamond}(a_{i,j'}) \Big) \\ &= \mathbb{U}(\mathbb{D}_{\mathsf{kh}}\mathfrak{X}) \cap \bigcup_{k\in K} \Big(\bigcap_{\ell\in L} \lambda^{\Box}(a_{k,\ell}) \cap \bigcap_{\ell'\in L'} \lambda^{\diamond}(a_{k,\ell'}) \Big) \\ &= \bigcup_{k\in K} \Big(\bigcap_{\ell\in L} \psi'(\lambda^{\Box}(a_{k,\ell})) \cap \bigcap_{\ell'\in L'} \psi'(\lambda^{\diamond}(a_{k,\ell'})) \Big). \end{split}$$

So ψ' is well defined.

It is easy to see that $\bigvee_{\delta \in B} \lambda(\delta) \sim \lambda(\bigvee B)$ implies $(\bigvee_{\delta \in B} \lambda(\delta), \lambda(\bigvee B)) \in \ker(\psi)$ for $\lambda \in \{\lambda^{\Box}, \lambda^{\diamondsuit}\}$. Since these pairs generate of the congruence from definition 4.3, we have $\sim = \ker(q_{\mathbf{X}}) \subseteq \ker(\psi')$ and hence there exists a map $\psi : \mathbb{F}_{\{\lambda^{\Box}, \lambda^{\diamondsuit}\}} \mathbf{X} \to \mathfrak{opn}(\widehat{\mathbb{T}}\mathbf{X})$ such that the diagram in (4.2) commutes. Therefore ψ is (well) defined on generators by $[\lambda^{\Box}(a)] \mapsto \Box a$ and $[\lambda^{\diamondsuit}(a)] \mapsto \diamondsuit a$. One can easily check that $\psi \circ \varphi = \mathrm{id}$ and $\varphi \circ \psi = \mathrm{id}$ by looking at the action on the generators. It follows that φ is injective.

This entails that for compact Hausdorff spaces \mathfrak{X} ,

$$\ddot{\mathbb{D}}_{\{\lambda^{\Box},\lambda^{\diamondsuit}\}}\mathfrak{X} = \mathbb{D}_{\mathsf{kh}}\mathfrak{X},$$

Furthermore, it can be seen that for continuous maps $f : \mathfrak{X} \to \mathfrak{X}'$ we have $\mathbb{F}_{\{\lambda^{\Box}, \lambda^{\diamondsuit}\}} f = \mathfrak{opn}(\mathbb{D}_{\mathsf{kh}} f)$. As a consequence, when restricted to KHaus we have

$$(\ddot{\mathbb{D}}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}})_{\uparrow\mathsf{KHaus}} = \mathbb{D}_{\mathsf{kh}},$$

that is, lifting the monotone functor on Set with respect to the box/diamond lifting yields the monotone functor on KHaus from definition 3.37.

4.10 Example. Using similar techniques as in the previous example, one can show that, when restricted to KHaus, the sober Kupke-Kurz-Pattinson lift of \mathbb{P} with respect to the usual box and diamond lifting coincides with the Vietoris functor. (An algebraic description similar to the one in theorem 3.41 is given in definition 3.35. It is shown in proposition III4.6 of [29] that this algebraic description is dual to \mathbb{V}_{kh} .)

A third example of this method of lifting functors can be found in section 5.2: The functor on KHaus corresponding to geometric conditional frames arises as the restriction of the lift of the functor \mathbb{C} from example 2.10 with respect to the predicate liftings $\lambda^{\Rightarrow}, \lambda^{\downarrow}$ from example 2.22.

4.2 FROM Set TO Stone

We turn our attention from sober spaces to Stone spaces. Let \mathbb{T} : Set \rightarrow Set be a functor and Λ a collection of predicate liftings for \mathbb{T} . Along the same lines as the previous section, we construct a lifted version $\widehat{\mathbb{T}}_{\Lambda}$: Stone \rightarrow Stone (the Kupke-Kurz-Pattinson lift).

4.11 Definition. Let $\mathbb{T} : \mathsf{Set} \to \mathsf{Set}$ be a functor and Λ a set of predicate liftings for \mathbb{T} . For a Stone space \mathfrak{X} let $\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}$ be the sub-Boolean algebra of $\mathbb{Q}(\mathbb{T}\mathfrak{X})$ generated by the set

$$\{\lambda_{\mathbb{U}\mathfrak{X}}(a_1,\ldots,a_n) \mid \lambda \in \Lambda, a_1,\ldots,a_n \in \mathbb{C}lp\,\mathfrak{X}\}.$$
(4.4)

That is, $\widehat{\mathbb{B}}_{\Lambda} \mathfrak{X}$ is obtained by closing the set in (4.4) under taking complements, finite intersections and finite unions. For a continuous map $f: \mathfrak{X} \to \mathfrak{X}'$ let $\widehat{\mathbb{B}}_{\Lambda} f: \widehat{\mathbb{B}}_{\Lambda} \mathfrak{X}' \to \widehat{\mathbb{B}}_{\Lambda} \mathfrak{X}$ be the restriction of $\mathbb{Q}(\mathbb{T}f)$ to $\widehat{\mathbb{B}}_{\Lambda} \mathfrak{X}'$.

4.12 Lemma. The assignment $\widehat{\mathbb{B}}_{\Lambda}$ defines a contravariant functor.

Proof. We need to show that $\widehat{\mathbb{B}}_{\Lambda}$ is functorial and that $\widehat{\mathbb{B}}_{\Lambda}f$ is well defined for every continuous function between Stone spaces $f: \mathfrak{X} \to \mathfrak{X}'$. For the latter, it suffices to show that for all generators $\lambda_{\mathfrak{X}'}(a'_1, \ldots, a'_n)$ of $\widehat{\mathbb{B}}\mathfrak{X}'$ we have $(\widehat{\mathbb{B}}_{\Lambda}f)(\lambda_{\mathfrak{X}'}(a'_1, \ldots, a'_n)) \in \widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}$, because the homomorphism $\mathbb{Q}(\mathbb{T}f)$ (of which $\widehat{\mathbb{B}}_{\Lambda}f$ is a restriction) preserves taking complements, finite intersections and finite unions. By naturality of λ we have

$$(\widehat{\mathbb{B}}_{\Lambda}f)(\lambda_{\mathfrak{X}'}(a_1',\ldots,a_n')) = (\mathbb{T}f)^{-1}(\lambda_{\mathfrak{X}'}(a_1',\ldots,a_n')) = \lambda_{\mathfrak{X}}(f^{-1}(a_1'),\ldots,f^{-1}(a_n'))$$

and because the $f^{-1}(a'_i)$ are clopen in \mathfrak{X} , the result is in $\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}$. Functoriality of $\widehat{\mathbb{B}}_{\Lambda}$ follows from functoriality of $\mathbb{Q} \circ \mathbb{T}$.

4.13 Definition. Let \mathbb{T} : Set \rightarrow Set be a functor and Λ a set of predicate liftings for \mathbb{T} . The **Kupke-Kurz-Pattinson lift** of \mathbb{T} to Stone is

$$\widehat{\mathbb{T}}_{\Lambda} \coloneqq \mathfrak{uf} \circ \widehat{\mathbb{B}}_{\Lambda}. \qquad \qquad \triangleleft$$

4.14 Definition. Let Λ be a set of predicate liftings for a functor \mathbb{T} . Then any $\lambda : \mathbb{P}^n \to \mathbb{P} \circ \mathbb{T}$ in Λ induces a clopen predicate lifting $\widehat{\lambda} : \mathbb{Clp} \to \mathbb{Clp} \circ \widehat{\mathbb{T}}_{\Lambda}$ for $\widehat{\mathbb{T}}_{\Lambda}$ via

$$\mathbb{Clp}^{n} \mathfrak{X} \xrightarrow{\lambda_{\mathbb{U}\mathfrak{X}}} \mathbb{Z}(\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}) \xrightarrow{\mathbb{Z}k_{\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}}} \mathbb{Clp}(\mathfrak{uf}(\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X})) = \mathbb{Clp}(\widehat{\mathbb{T}}_{\Lambda}\mathfrak{X}).$$

(To be precise, the domain of $\lambda_{\mathbb{U}\mathfrak{X}}$ is $\mathbb{Q}^n(\mathbb{U}\mathfrak{X})$. We only need the restriction to $\mathbb{C}lp^n\mathfrak{X}$. We still write $\lambda_{\mathbb{U}\mathfrak{X}}$ for this restriction in order to avoid clutter.) The map $k_{\mathbb{F}\mathfrak{X}}$ is the isomorphism given by Stone duality. Naturality of $\widehat{\lambda}$ follows from naturality of λ . We state some properties of lifted predicate liftings, the proofs of which are left to the reader.

4.15 Lemma. Let \mathbb{T} be a set functor, Λ a collection of predicate liftings for \mathbb{T} , $\widehat{\mathbb{T}}_{\Lambda}$ the KKP lift of \mathbb{T} with respect to Λ and $\widehat{\Lambda} = \{\widehat{\lambda} \mid \lambda \in \Lambda\}$.

- (i) If $\lambda \in \Lambda$ is monotone, then so is $\widehat{\lambda}$.
- (ii) If Λ is monotone, then so is $\widehat{\Lambda}$.
- (iii) If Λ is closed under duals, then so is $\widehat{\Lambda}$.

4.16 Remark ([50], proposition 43). As an instance of the Yoneda lemma, the *n*-ary predicate liftings for a set functor \mathbb{T} are in one-to-one correspondence with subsets of $\mathbb{T}(2^n)$. (The set 2^n is the *n*-fold product of the two-element space $2 = \{\mathsf{T}, \bot\}$.) To an *n*-ary predicate lifting λ assign the subset $\lambda_{2^n}(\pi_1^{-1}(\{\mathsf{T}\}), \ldots, \pi_n^{-1}(\{\mathsf{T}\})) \subseteq \mathbb{T}(2^n)$, where $\pi_i : 2^n \to 2$ is the *i*-th projection. Conversely, for $C \subseteq \mathbb{T}(2^n)$, the assignment $\lambda_X^C(a_1, \ldots, a_n) := (\mathbb{T}\langle \chi_{a_1}, \ldots, \chi_{a_n} \rangle)^{-1}[C] \subseteq \mathbb{T}X$ yields a predicate lifting for \mathbb{T} . The angled brackets denote the tupling of the indicator functions χ_{a_i} .

The correspondence gives a canonical collection of predicate liftings for each set functor \mathbb{T} , namely the collection of all predicate liftings for \mathbb{T} .

Example 10 in [36] states that \mathbb{P} with the collection Λ of all predicate liftings gives $\widehat{\mathbb{P}}_{\Lambda} = \mathbb{V}_{st}$. A different way to obtain \mathbb{V}_{st} as a lifted functor is by lifting \mathbb{P} with respect to the usual predicate lifting for box and diamond. Before we can show this in example 4.19, we need the following definition and lemma. For a proof we refer to [29, Section III.4].

4.17 Definition. The map \mathbb{N}_{ba} : $BA \to BA$ is defined on an object B as the free Boolean algebra generated by $\Box a, \Diamond a, (a \in B)$ subject to the relations

 $(V1) \Box T = T$ $(V2) \diamond \bot = \bot$ $(V3) \Box (a \land b) = \Box a \land \Box b$ $(V4) \diamond (a \lor b) = \diamond a \lor \diamond b$ $(V5) \Box (a \lor b) \leq \Box a \lor \diamond b$ $(V6) \Box a \land \diamond b \leq \diamond (a \land b)$

For a homomorphism $f : B \to B'$ define $\mathbb{N}_{\mathsf{ba}}f : \mathbb{N}_{\mathsf{ba}}B \to \mathbb{N}_{\mathsf{ba}}B'$ on generators by $\mathbb{N}_{\mathsf{ba}}(\Box a) \mapsto \Box(f(a))$ and $\mathbb{N}_{\mathsf{ba}}(\Diamond a) \mapsto \Diamond(f(a))$.

4.18 Lemma. If \mathfrak{X} is a Stone space and \mathbb{V}_{st} is the Vietoris functor on Stone, then

$$\chi:\mathfrak{clp}(\mathbb{V}_{st}\mathfrak{X})\to\mathbb{N}_{ba}(\mathfrak{clp}\mathfrak{X})$$

defined on generators by $\boxplus a \mapsto \Box a$ and $\otimes a \mapsto \Diamond a$ is an isomorphism.

We will now show that the Kupke-Kurz-Pattinson lift of the powerset functor \mathbb{P} with respect to the box- and diamond lifting (given below) yields the Vietoris functor on Stone.

4.19 Example. Recall that for the powerset functor \mathbb{P} , the box- and diamond-lifting $\lambda^{\Box}, \lambda^{\diamond} : \check{\mathbb{P}} \to \check{\mathbb{P}}\mathbb{P}$ are given by

$$\lambda_X^{\square} : a \mapsto \{ b \in \mathbb{P}X \mid b \subseteq a \}, \quad \lambda_X^{\diamondsuit} : a \mapsto \{ b \in \mathbb{P}X \mid b \cap a \neq \emptyset \}.$$

Let \mathfrak{X} be a Stone space. We claim that $\widehat{\mathbb{P}}_{\{\lambda^{\Box},\lambda^{\diamondsuit}\}} = \mathbb{V}_{st}$.

Let \mathfrak{X} be a Stone space. We will show that $\widehat{\mathbb{B}}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}}\mathfrak{X} \cong \mathfrak{clp}(\mathbb{V}\mathfrak{X})$. Define φ : $\mathfrak{clp}(\mathbb{V}_{\mathfrak{st}}\mathfrak{X}) \to \widehat{\mathbb{B}}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}}\mathfrak{X}$ on generators by $\boxplus a \mapsto \lambda^{\square}(a)$ and $\&a \mapsto \lambda^{\diamondsuit}a$. By lemma 4.18 we may view φ as a morphism $\mathbb{N}_{\mathsf{ba}}(\mathfrak{clp} \mathfrak{X}) \to \widehat{\mathbb{B}}_{\{\lambda^{\Box}, \lambda^{\Diamond}\}} \mathfrak{X}$ so in order to show that φ is a well-defined Boolean algebra homomorphism, it suffices to show that the $\lambda^{\Box}(a)$ and $\lambda^{\diamond}(a)$ (where a ranges over $\mathbb{Clp}\mathfrak{X}$) satisfy relation (V1) – (V6) from definition 4.17. We leave this verification to the reader.

The map φ is surjective because the generators of $\widehat{\mathbb{B}}_{\{\lambda^{\Box},\lambda^{\Diamond}\}}$ are all in its image. So we only need to show injectivity. Observe that, when seen as subsets of $\mathbb{P}(\mathbb{UX})$, we have

$$\boxplus a = \lambda^{\square}(a) \cap \mathbb{V}_{st} \mathfrak{X} \quad \text{and} \quad \otimes a = \lambda^{\diamondsuit}(a) \cap \mathbb{V}_{st} \mathfrak{X}.$$

Since λ^{\diamond} is the dual of λ^{\Box} , every element of $\mathfrak{clp}(\mathbb{V}_{st}\mathfrak{X})$ is a finite union of finite intersections of generators (i.e. elements of the form $\boxplus a, \otimes a$, where $a \in \mathfrak{clp} \mathfrak{X}$). (Indeed, we do not need to take complements because the generators are closed under complementation in \mathfrak{X} .) So we may write an arbitrary element $x \in \mathfrak{clp}(\mathbb{V}_{st}\mathfrak{X})$ as x = $\bigcup_{i=1}^{n} \left(\bigcap_{j=1}^{m_{i}} \boxplus a_{i,j} \cap \bigcap_{j'=1}^{m'_{i}} \otimes a_{i,j'} \right).$ It is easy to see that $x = \varphi(x) \cap \mathbb{V}_{st} \mathcal{X}$. Therefore, if x, x' are elements of $\mathfrak{clp}(\mathbb{V}_{st} \mathcal{X})$ and $\varphi(x) = \varphi(x')$, then x = x'. This shows that φ is injective. It follows from unraveling the definitions that $\widehat{\mathbb{P}}_{\{\lambda^{\square}, \lambda^{\Diamond}\}} f = \mathbb{V}_{st} f$ when f is a continuous

map between Stone spaces. Therefore

$$\widehat{\mathbb{P}}_{\{\lambda^{\square},\lambda^{\diamondsuit}\}} = \mathbb{V}_{\mathsf{st}},$$

that is, the Kupke-Kurz-Pattinson lift of \mathbb{P} with respect $\{\lambda^{\Box}, \lambda^{\diamondsuit}\}$ to Stone gives the Vietoris functor on Stone.

We have seen in example 4.10 that, when restricted to KHaus, the lift of \mathbb{P} with respect to $\{\lambda^{\Box}, \lambda^{\diamondsuit}\}$ to a functor on **Sob** coincides with the Vietoris functor, i.e. $(\mathbb{P}_{\{\lambda^{\Box}, \lambda^{\diamondsuit}\}})_{\uparrow \mathsf{KHaus}} =$ \mathbb{V}_{kh} . This implies

$$(\ddot{\mathbb{P}}_{\{\lambda^{\Box},\lambda^{\diamondsuit}\}})_{\uparrow \mathsf{Stone}} = \widehat{\mathbb{P}}_{\{\lambda^{\Box},\lambda^{\diamondsuit}\}}.$$

A natural question that arises is how $\ddot{\mathbb{T}}_{\Lambda}$ compares to $\widehat{\mathbb{T}}_{\Lambda}$ in general. We leave this as an interesting open question.

We give another example.

4.20 Example. Let \mathbb{D} be the monotone functor on Set and $\lambda^{\Box}, \lambda^{\diamond}$ the usual predicate liftings for \mathbb{D} . Using similar reasoning as in example 4.19, it can be seen that $\widehat{\mathbb{D}}_{\{\lambda^{\Box},\lambda^{\diamondsuit}\}} =$ \mathbb{D}_{st} .

In section 5.1 the objects of study are descriptive conditional frames. The functor that corresponds to descriptive conditional frames can be obtained by lifting $\mathbb C$ from example 2.10 with respect to the predicate liftings $\lambda^{\Rightarrow}, \lambda^{\downarrow}$ from example 2.22 to a Stone functor. We will see that $\widehat{\mathbb{C}}_{\{\lambda \Rightarrow, \lambda^{\Downarrow}\}}$ is not the restriction of $\widehat{\mathbb{C}}_{\{\lambda \Rightarrow, \lambda^{\Downarrow}\}}$ to Stone.

4.3THE PRO-COMPLETION LIFT FROM Set TO Stone

In the final section of this chapter we define a different way to lift a certain class of set functors to Stone functors and we compare this lift with the Kupke-Kurz-Pattinson lift from definition 4.13. We assume that the set functor \mathbb{T} that we work with sends finite sets to finite sets and preserves cofiltered diagrams (that is, \mathbb{T} sends cofiltered diagrams to cofiltered diagrams). Let FinSet and FinStone be the full subcategories of Set and Stone whose objects are finite sets and finite Stone spaces, respectively. We make use of the following important observations.

Observation 1. The category Stone is equivalent to the pro-completion of FinStone [29, Theorem VI2.3]. Essentially, this means that every Stone space is an inverse limit of finite Stone spaces. For details, see chapter VI of [29].

Observation 2. A topology on a finite set is a Stone topology if and only if it is the discrete topology. Therefore every function between finite Stone spaces is continuous. It follows that the category FinSet of finite sets and functions is isomorphic to the category FinStone of finite Stone spaces and continuous functions.

Observation 3. Since \mathbb{T} restricts to FinSet and FinSet is isomorphic to FinStone, the restriction of \mathbb{T} to FinSet defines a functor on FinStone.

Observation 4. A finite quotient $q_{\mathcal{Y}}: \mathfrak{X} \twoheadrightarrow \mathfrak{Y}$ of a Stone space \mathfrak{X} is simply a partition of \mathfrak{X} into clopen subsets. For $x \in \mathfrak{X}$, let [x] be the equivalence class of x in \mathfrak{Y} . For each $[x] \in \mathfrak{Y}$ the set $q_{\mathfrak{Y}}^{-1}([x])$ is an element of this partition. Moreover, since \mathfrak{Y} is finite $\{[x]\}$ is clopen for each $[x] \in \mathfrak{Y}$, so by continuity of $q_{\mathfrak{Y}}$ the set $q_{\mathfrak{Y}}^{-1}([x]) \subseteq \mathfrak{X}$ is clopen in \mathfrak{X} .

Observation 5. The category BA has all limits and colimits because it is the category of models of an algebraic theory (for details see [18]). Since Stone is dually equivalent to BA it has all limits and colimits as well.

The first observation suggests that we could maybe define a functor $\overline{\mathbb{T}}$ on Stone by defining its action on finite Stone spaces and then taking the pro-completion. Although there is a neat way of lifting \mathbb{T} to a map on the collection of Stone spaces, we have not yet found a suitable way of defining $\overline{\mathbb{T}}f$, for morphisms f in Stone.

In the following definition we define the action of $\overline{\mathbb{T}}$ on objects. When we say that \mathcal{Y} is a finite quotient of \mathcal{X} , we take this literally, i.e. $\mathcal{Y} = \mathcal{X}/\sim$ for some equivalence relation \sim on \mathcal{X} . Therefore, each finite quotient \mathcal{Y} of \mathcal{X} comes equipped with a unique quotient map $q_{\mathcal{Y}} : \mathcal{X} \to \mathcal{Y}$.

4.21 Definition. Let \mathbb{T} be a set functor which sends finite sets to finite sets. Let \mathfrak{X} be a Stone space. Let \mathcal{C} be the collection of finite quotients of \mathfrak{X} ,

 $C = \{ \mathbf{\mathcal{Y}} \mid \mathbf{\mathcal{Y}} \text{ is a finite quotient of } \mathbf{\mathcal{X}} \}.$

For each $\mathcal{Y} \in \mathcal{C}$ let $q_{\mathcal{Y}} : \mathcal{X} \to \mathcal{Y}$ be the unique corresponding quotient map. For all $\mathcal{Y}, \mathcal{Y}' \in \mathcal{C}$, add to \mathcal{C} those continuous maps $f : \mathcal{Y} \to \mathcal{Y}'$ that satisfy $f \circ q_{\mathcal{Y}} = q_{\mathcal{Y}'}$, i.e. such that the following commutes:



Then C is a diagram in Stone and in FinStone.

Let $\mathbb{T}C$ be the diagram obtained from applying \mathbb{T} to the diagram C. Define $\overline{\mathbb{T}X}$ to be the categorical limit

 $\lim \mathbb{T} \mathcal{C}$

of the diagram $\mathbb{T}C$ in Stone (see figure 4.1). This exists by observation 5 above. We call this the lift via pro-completion.



Figure 4.1: The coloured area represents the diagram $\mathbb{T}C$.

Although, formally, a diagram in C is a functor $F : J \rightarrow C$, where J is some index category, we shall think of a diagram as a collection of objects and morphisms between them in C.

4.22 Definition. A diagram C in some category C is filtered if:

- (i) For any two objects X, X' in C there exists an object Y and morphisms $f: X \to Y, f': X' \to Y$ in C; and
- (ii) For any two parallel morphisms $f, f': X \to X'$ in C there is a morphism $g: X' \to Y$ in C such that gf = gf'.

The definition of a **cofiltered diagram** is the same, but with the direction of the morphisms reversed. \triangleleft

The diagram C from definition 4.21 is cofiltered.

4.23 Lemma. Let \mathfrak{X} be a Stone space and C the diagram of finite quotients of \mathfrak{X} as defined in definition 4.21. Then C is a cofiltered diagram.

Proof. For the first item of definition 4.22, suppose \mathcal{Y}_1 and \mathcal{Y}_2 are two finite quotients of \mathcal{X} . We may view them as finite partitions of \mathcal{X} into clopen sets, say, $\mathcal{Y}_1 = \{a_1, \ldots, a_n\}$ and $\mathcal{Y}_2 = \{b_1, \ldots, b_m\}$. Let $\mathcal{Y}_3 \coloneqq \{a_i \cap b_j \mid 1 \le i \le n, 1 \le j \le m\}$. Clearly the $a_i \cap b_j$ are clopen in \mathcal{X} , so \mathcal{Y} is a finite partition of \mathcal{X} , hence a finite quotient. Define $f_i : \mathcal{Y}_3 \to \mathcal{Y}_i$ by $f_i([x]_3) \mapsto [x]_i$ for i = 1, 2, where $[x]_i$ denotes the equivalence class of x in \mathcal{Y}_i . It is easy to check that the f_i are well defined and in \mathcal{C} .

For the second item, let $f_1, f_2 : \mathcal{Y}_2 \to \mathcal{Y}_1$ be parallel morphisms in \mathcal{C} . Then the identity morphism on \mathcal{Y}_2 has the desired property:



Indeed, for all $[x]_2 \in \mathcal{Y}$ we have $f_1(\mathrm{id}_{\mathcal{Y}_2}([x]_2)) = f_1(q_2(x)) = q_1(x) = f_2(q_2(x)) = f_2(\mathrm{id}_{\mathcal{Y}_2}([x]_2)).$

The question whether this lift via pro-completions and the KKP lift coincide was posed in the conclusion of [36]. We conclude this section by answering this question affirmatively for the objects.

4.24 Theorem. Let \mathbb{T} be a set functor that restricts to FinSet and which preserves cofiltered diagrams. Then on objects the canonical lift given in definition 4.21 and the Kupke-Kurz-Pattinson lift of \mathbb{T} to Stone with respect to the collection Λ of all predicate liftings coincide.

Proof. Let \mathfrak{X} be a Stone space and C the collection of finite quotients as defined in definition 4.21. We will compare the clopen set Boolean algebras of $\overline{\mathbb{T}}\mathfrak{X}$ and $\widehat{\mathbb{T}}\mathfrak{X}$. The latter, $\widehat{\mathbb{B}}_{\Lambda}\mathfrak{X}$, is generated as a sub-Boolean algebra of $\mathbb{Q}\mathbb{T}\mathbb{U}\mathfrak{X}$ by the elements

$$\{\lambda_X^C(a_1,\ldots,a_n) \mid n \ge 0, C \subseteq \mathbb{T}(2^n), a_i \in \operatorname{Clop} \mathfrak{X}\},\$$

where λ^C denotes the *n*-ary predicate lifting corresponding to the subset $C \subseteq \mathbb{T}(2^n)$, cf. proposition 4.16. Abbreviate $\widehat{\mathscr{B}} \coloneqq \widehat{\mathbb{B}}_{\Lambda} \mathfrak{X}$. (Note that we implicitly use Stone duality to say that $\mathfrak{clp}(\widehat{\mathbb{T}}\mathfrak{X}) = \mathfrak{clp}(\mathfrak{uf}\widehat{\mathscr{B}}) = \widehat{\mathscr{B}}$.)

Let $\overline{\mathcal{B}}$ be the clopen set Boolean algebra of $\overline{\mathbb{T}}\mathfrak{X}$. Then by the duality of Stone and BA the Boolean algebra $\overline{\mathcal{B}}$ is a colimit of the diagram $\mathfrak{opn}(\mathbb{T}\mathcal{C})$ in BA obtained by applying $\mathfrak{clp} \circ \mathbb{T} \circ \mathbb{U}$ to the diagram \mathcal{C} . (After applying $\mathbb{T} \circ \mathbb{U}$ we view the objects as finite Stone spaces rather than finite sets.)

Observe that, if we view $\mathbb{T}(\mathbb{U}(\mathcal{Y}_i))$ as a Stone space, we have $\mathfrak{clp}(\mathbb{T}(\mathbb{U}(\mathcal{Y}_i))) = \mathbb{Q}(\mathbb{T}(\mathbb{U}(\mathcal{Y}_i)))$. Moreover, for all morphisms f in \mathcal{C} we have $(\mathbb{T}(\mathbb{U}f))^{-1} = \mathfrak{clp}(\mathbb{T}(\mathbb{U}f)) = \mathbb{Q}(\mathbb{T}(\mathbb{U}f))$. Accordingly, the map $(\mathbb{T}(\mathbb{U}q_i))^{-1} : \mathfrak{clp}(\mathbb{T}\mathcal{Y}_i) \to \mathbb{Q}\mathbb{T}\mathbb{U}\mathcal{X}$ is a Boolean algebra morphism for each of the Boolean algebras $\mathfrak{clp}(\mathbb{T}(\mathbb{U}(\mathcal{Y}_i)))$. Applying $\mathbb{Q} \circ \mathbb{T} \circ \mathbb{U}$ to the diagram \mathcal{C} shows that these morphisms commute with the morphisms in $\mathfrak{clp}(\mathbb{T}\mathcal{C})$. Therefore, by the definition of a colimit, there exists a homomorphism $f : \overline{\mathcal{B}} \to \mathbb{Q}(\mathbb{T}(\mathbb{U}\mathcal{X}))$. (See figure 4.2.)



Figure 4.2: The coloured area represents the diagram $\mathfrak{clp}(\mathbb{T}C)$.

Let \mathcal{B} be the sub-Boolean algebra of \mathbb{QTUX} generated by the clopen sets $(\mathbb{T}q_i)^{-1}(a)$ where *i* ranges over *I* and *a* ranges over the clopens of \mathcal{Y}_i . Then im $f = \mathcal{B}$. We will now show that *f* is in fact injective, hence an isomorphism.

4.24.A Claim. We have an isomorphism of Boolean algebras $\overline{\mathcal{B}} \cong \mathcal{B}$.

Proof of claim. We already have an onto homomorphism $f:\overline{\mathcal{B}} \to \mathcal{B}$, so we only need to show that f is injective. By lemma A.8 it suffices to show that $f(z) = \bot$ implies $z = \bot$ for all $z \in \overline{\mathcal{B}}$.

Given $q_i : \mathfrak{X} \to \mathfrak{Y}_i$, define $r_i : \mathfrak{Y}_i \to \mathfrak{X}$ by choosing for each equivalence class in \mathfrak{Y}_i a representative. Then r_i is a section of q_i , that is, $q_i \circ r_i = \mathrm{id}_{\mathfrak{Y}_i}$.

Let $z \in \mathfrak{clp}(\overline{\mathbb{T}}\mathfrak{X}) = \overline{\mathscr{B}}$ such that $f(z) = \bot$. By lemma 4.23 the diagram C is cofiltered, so since \mathbb{T} preserves cofiltered diagrams the diagram $\overline{\mathbb{T}}C$ with limit $\overline{\mathbb{T}}\mathfrak{X}$ is cofiltered as well, hence by duality the diagram $\mathfrak{clp}(\mathbb{T}C)$ with limit $\overline{\mathscr{B}}$ is filtered. It follows from a routine argument that z comes from some $y \in \mathfrak{clp}(\mathbb{T}\mathfrak{Y}_i)$ for some $i \in I$, so $f(z) = (\mathbb{T}q_i)^{-1}(y)$. Then $(\mathbb{T}q_i)^{-1}(y) = \bot$. As $(\mathbb{T}r_i)^{-1} \circ (\mathbb{T}q_i)^{-1} = (\mathbb{T}\operatorname{id}_{\mathfrak{Y}_i})^{-1} = \mathbb{T}\operatorname{id}_{\mathfrak{Y}_i} = \operatorname{id}_{\mathbb{T}\mathfrak{Y}_i}$, we see that $(\mathbb{T}q_i)^{-1}$ has a left inverse, hence is injective. Therefore $(\mathbb{T}q_i)^{-1}(y) = \bot$ implies $y = \bot$ which in turn implies $z = \bot$. So f is injective.

So it suffices to compare $\widehat{\mathcal{B}}$ and \mathcal{B} , which are both sub-Boolean algebras of \mathbb{QTUX} . 4.24.B Claim. Every generator of $\widehat{\mathcal{B}}$ is in \mathcal{B} , hence $\widehat{\mathcal{B}}$ is a sub-Boolean algebra of \mathcal{B} .

Proof of claim. Pick $n \ge 0$, $C \subseteq \mathbb{T}(2^n)$ and $a_1, \ldots, a_n \in \mathbb{Clp} \mathfrak{X}$ and consider

$$\lambda^C(a_1,\ldots,a_n) = (\mathbb{T}\langle \chi_{a_1},\ldots,\chi_{a_n}\rangle)^{-1}[C].$$

This is an arbitrary generator of $\widehat{\mathcal{B}}$. We will show that $\lambda^{C}(a_{1},\ldots,a_{n}) \in \mathcal{B}$. Recall that $\chi_{a_{1}}$ is the indicator function $\mathfrak{X} \to 2$ and $\langle \chi_{a_{1}},\ldots,\chi_{a_{n}} \rangle$ is the tupling of indicator functions.

For every $a \in \mathbb{Clp} \mathfrak{X}$, let \mathfrak{Y}_a be the finite quotient of \mathfrak{X} given by $\{a, \mathfrak{X} \setminus a\}$ and let q_a be the corresponding quotient map. (This is well-defined because obviously q_a is continuous.) Let $i_k : 2 \to \mathfrak{Y}_{a_k}$ be the isomorphism which sends \top to the equivalence class a_k in \mathfrak{Y}_k and \perp to $\mathfrak{X} \setminus a_k$. Then the tupling $i = \langle i_1, \ldots, i_n \rangle$ is an isomorphism $i : 2^n \to \mathfrak{Y}_{a_1} \times \cdots \times \mathfrak{Y}_{a_n}$ is an isomorphism. Set $q = \langle q_{a_1}, \ldots, q_{a_n} \rangle$. Then the upper square in the left diagram below commutes.

Let $F := \{\bigcap_{i=1}^{n} \mathcal{B}_i \mid \mathcal{B}_i \in \mathcal{Y}_{a_i}\}$ and $j : \mathcal{Y}_{a_1} \times \cdots \times \mathcal{Y}_{a_n} \to F : (\mathcal{B}_1, \ldots, \mathcal{B}_n) \mapsto \bigcap \mathcal{B}_i$. Then F is a finite quotient of \mathfrak{X} . Call the corresponding quotient map q_F . The lower triangle in the left diagram commutes by construction. Applying \mathbb{T} to the left diagram yields the right diagram.



Since *i* is an isomorphism, so is $\mathbb{T}i$. Now we get

$$(\mathbb{T}\langle\chi_{a_1},\ldots,\chi_{a_n}\rangle)^{-1}[C] = (\mathbb{T}q)^{-1}[(\mathbb{T}i)[C]] = (\mathbb{T}q_F)^{-1}\underbrace{[\mathbb{T}j\circ\mathbb{T}i[C]]}_{\subseteq\mathbb{T}F}.$$

This shows that $\lambda^{C}(a_{1},\ldots,a_{n}) \in \mathcal{B}$. Since this holds for any generator $\lambda^{C}(a_{1},\ldots,a_{n})$ of $\widehat{\mathcal{B}}$, we have $\widehat{\mathcal{B}} \subseteq \mathcal{B}$.

Our next claim is the converse of claim 4.24.B.

4.24.C Claim. Every generator of \mathcal{B} is in $\widehat{\mathcal{B}}$, hence \mathcal{B} is a sub-Boolean algebra of $\widehat{\mathcal{B}}$.

Proof of claim. Let $\mathcal{Y} = \{a_1, \ldots, a_n\}$ and let $q_{\mathcal{Y}} : \mathcal{X} \twoheadrightarrow \mathcal{Y}$ be a finite quotient of \mathcal{X} and let $B \subseteq \mathbb{T}\mathcal{Y}$ be a subset. (Since $\mathbb{T}\mathcal{Y}$ is finite B is automatically clopen.) Then the following diagram commutes



where $q,\,i$ and j are as before. It is obvious that the diagram commutes. Apply $\mathbb T$ to the diagram to see that

$$(\mathbb{T}q)^{-1}[B] = (\mathbb{T}q)^{-1}[(\mathbb{T}j)^{-1}[B]] = (\mathbb{T}\langle \chi_{a_1}, \dots, \chi_{a_n} \rangle)^{-1}[(\mathbb{T}i)^{-1} \circ (\mathbb{T}j)^{-1}[B]].$$

Note that $(\mathbb{T}i)^{-1} \circ (\mathbb{T}j)^{-1}[B] \subseteq \mathbb{T}(2^n)$. This proves that every generator of \mathcal{B} is in $\widehat{\mathcal{B}}$, hence $\mathcal{B} \subseteq \widehat{\mathcal{B}}$.

We conclude that the clopen set Boolean algebras of $\overline{\mathbb{T}}\mathfrak{X}$ and $\widehat{\mathbb{T}}\mathfrak{X}$ coincide. Therefore $\overline{\mathbb{T}}\mathfrak{X} \cong \widehat{\mathbb{T}}\mathfrak{X}$. This proves the theorem.

This concludes the general theory that we treat in this thesis. The next chapter will consist of a case study where we will see parts of the developed theory in action.

Coalgebraic conditional logic

Recall from section 2.2 that the modality \Rightarrow (the conditional arrow) expresses a notion of conditionality which generally differs from the implication \rightarrow , and a formula $\varphi \Rightarrow \psi$ may be thought of as "If φ holds, then usually ψ holds as well." Conditional logics have a long history and tradition in philosophical logic [53, 13, 54]. They are used in various applications such as non-monotonic interference, belief change and the analysis of intentions. For more information on conditional logic we refer to [13, 4, 42].

This chapter consists of two parts. The first part contains the generalisation of conditional frames (from example 2.10) to *descriptive* conditional frames. We show that these can be seen as coalgebras for a certain functor \mathbb{C}_{st} on Stone and give an endofunctor on BA dual to \mathbb{C}_{st} . Also we generalise conditional bisimulations from [7] to descriptive conditional bisimulations and show how they relate to Λ -bisimulations (where Λ is the set of clopen predicate liftings that corresponds to conditional logic), modal equivalence and behavioural equivalence.

In the second part we develop geometric conditional logic by adding two modalities, \Rightarrow and \downarrow , to the language of geometric logic via predicate liftings. We generalise conditional frames to geometric conditional frames, show that these are coalgebras for a certain functor and give an Isbell dual of this functor.

5.1 CONDITIONAL LOGIC ON Stone-COALGEBRAS

In this section we develop conditional logic on so-called descriptive conditional models. We define descriptive conditional models and show how they can be viewed as a coalgebras. In subsection 5.1.2 we investigate duality with Boolean algebras. In subsection 5.1.3 we define descriptive conditional bisimulations, which are an adaptation of ideas in [7], and we show that descriptive conditional bisimilarity coincides with modal equivalence, behavioural equivalence and Λ -equivalence (for a suitable Λ).

5.1.1 Descriptive conditional models

Recall from definition 2.8 that a conditional frame is a pair (X, ν) where X is a set and $\nu : X \times \mathbb{P}X \to \mathbb{P}X$ a function satisfying certain properties. We extend this by adding to such a pair a collection $A \subseteq \mathbb{P}X$ of admissible subsets of X. We require A to be a Boolean algebra that is closed under certain properties, such that the truth set of every formula (see definition 5.12) is in A. Besides, A will serve as the clopen subbase for a topology on X. But this is jumping ahead; let us start with the definition of a general conditional frame.

5.1 Definition. A general conditional frame is a triple (X, ν, A) where

- X is a set;
- $A \subseteq \mathbb{P}X$ contains \emptyset and X and is closed under Boolean operations and the map $m_{\nu} : A \times \mathbb{P}X \to \mathbb{P}X$ defined by

$$m_{\nu}(a, b) \coloneqq \{x \in X \mid \nu(x, a) \subseteq b\}$$

- $\nu: X \times A \to \mathbb{P}X$ is a map that satisfies for all $x \in X$ and $a, b \in A$,
 - (i) if $a \cap b = \emptyset$, then $\nu(x, a) \cap b = \emptyset$,
 - (ii) if $a \subseteq b$ and $\nu(x, b) \subseteq a$, then $\nu(x, a) = \nu(x, b)$.

A general conditional frame morphism from (X, ν, A) to (X', ν', A') is a map $f: X \to X'$ such that $f^{-1}(a') \in A$ for all $a' \in A'$ and $f[\nu(x, f^{-1}(a'))] = \nu'(f(x), a')$ for all $x \in X, a' \in A'$.

A general conditional model is a tuple (X, ν, A, V) where (X, ν, A) is a general conditional frame and $V : \Phi \to A$ is a valuation of the proposition letters. A general conditional model morphism $f : (X, \nu, A, V) \to (X', \nu', A', V')$ is a descriptive conditional frame morphism $f : (X, \nu, A) \to (X', \nu', A')$ such that $f^{-1} \circ V' = V$.

As in the set case (cf. remark 2.9) condition (i) is equivalent to $\nu(x, a) \subseteq a$ for all $x \in X, a \in A$.

5.2 Definition. Let $\mathfrak{X} = (X, \nu, A)$ be a general conditional frame. Write \mathfrak{X} for the topological space with underlying set X and a topology generated by the subbase A, and $K\mathfrak{X}$ for the collection of closed subsets of \mathfrak{X} . Call a general conditional frame **differentiated** if for all distinct $x, x' \in X$ there is a witness $a \in A$ such that $x \in a$ and $x' \notin a$, **closed** if $\nu(x, a) \in K\mathfrak{X}$ for all $x \in X$ and $a \in A$, and **compact** if $\bigcap A_0 \neq \emptyset$ whenever $A_0 \subseteq A$ has the finite intersection property.

A descriptive conditional frame is a differentiated, closed, compact general conditional frame. The category of descriptive conditional frames and general descriptive frame morphisms is denoted by DCF. A descriptive conditional model is a general conditional model whose underlying general conditional frame is descriptive. Let DCM be the category whose objects are descriptive conditional models and whose morphisms are general conditional model morphisms.

It turns out that descriptive conditional frames can be viewed as coalgebras.

5.3 Definition. Let $\mathbb{C}_{st}\mathfrak{X}$ be the set of maps $h : \mathbb{C}lp \mathfrak{X} \to K\mathfrak{X}$ such that for all $a, b \in \mathbb{C}lp \mathfrak{X}$,

(C1) $h(a) \subseteq a$; and

(C2) if $a \subseteq b$ and $h(b) \subseteq a$, then h(a) = h(b).

Endow $\mathbb{C}_{st} \mathfrak{X}$ with a topology generated by the subbase

$$\boxplus (a, b) \coloneqq \{h \in \mathbb{C}_{\mathsf{st}} \mathfrak{X} \mid h(a) \subseteq b\}, \quad \& (a, b) \coloneqq \{h \in \mathbb{C}_{\mathsf{st}} \mathfrak{X} \mid h(a) \cap b \neq \emptyset\}$$

for $a, b \in \operatorname{Clop} \mathfrak{X}$. Observe that $\mathbb{C}_{\mathsf{st}} \mathfrak{X} \setminus \boxplus (a, b) = \otimes (a, \mathfrak{X} \setminus b)$, so this is in fact a *clopen* subbase. For a continuous map $f : \mathfrak{X} \to \mathfrak{X}'$ between Stone spaces, define $\mathbb{C}_{\mathsf{st}} f : \mathbb{C}_{\mathsf{st}} \mathfrak{X} \to \mathbb{C}_{\mathsf{st}} \mathfrak{X}'$ by

$$\mathbb{C}_{\mathsf{st}}f(h)(a) = f[h(f^{-1}(a))]$$

for $h \in \mathbb{C}_{st} \mathfrak{X}$ and $a \in \mathbb{Clp} \mathfrak{X}$.

 \triangleleft

5.4 Lemma. Let $f : \mathfrak{X} \to \mathfrak{X}'$ be a continuous map between Stone spaces. The map $\mathbb{C}_{st}f : \mathbb{C}_{st}\mathfrak{X} \to \mathbb{C}_{st}\mathfrak{X}'$ is well defined and continuous.

Proof. In order to prove that $\mathbb{C}_{st} f$ is well defined we need to show that for each $h \in \mathbb{C}_{st} \mathfrak{X}$ and $a \in \mathbb{Clp} \mathfrak{X}$ the set $f[h(f^{-1}(a))] \subseteq \mathfrak{X}'$ is closed and that $\mathbb{C}_{st} f(h)$ satisfies (C1) and (C2). Since a is clopen in \mathfrak{X}' , $f^{-1}(a)$ is clopen in \mathfrak{X} . By construction $h(f^{-1}(a))$ is closed in \mathfrak{X} , and since \mathfrak{X} is compact and \mathfrak{X}' is Hausdorff, f is a closed map so $f[h(f^{-1}(a))]$ is closed in \mathfrak{X}' . The map $\mathbb{C}_{st} f(h)$ satisfies (C1) and (C2) for similar reasoning as in paragraph 2.10, hence it is indeed an element of $\mathbb{C}_{st} \mathfrak{X}'$.

For continuity, it suffices to show that $(\mathbb{C}_{st}f)^{-1}(\boxplus(a, b))$ is clopen in $\mathbb{C}_{st}\mathfrak{X}$, whenever $a, b \in \mathbb{Clp}\mathfrak{X}'$. Unravelling the definitions yields

$$(\mathbb{C}_{\mathsf{st}}f)^{-1}(\boxplus(a, b)) = \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid \mathbb{C}_{\mathsf{st}}f(h) \in \boxplus(a, b)\}$$
$$= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid \mathbb{C}_{\mathsf{st}}f(h)(a) \subseteq b\}$$
$$= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid f[h(f^{-1}(a))] \subseteq b\}$$
$$= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid h(f^{-1}(a)) \subseteq f^{-1}(b)\}$$
$$= \boxplus(f^{-1}(a), f^{-1}(b)).$$

Clearly the latter is clopen in $\mathbb{C}_{st}\mathfrak{X}$. This proves the lemma.

We have not yet proved that $\mathbb{C}_{st}\mathfrak{X}$ is indeed a Stone space whenever \mathfrak{X} is. The next two lemmas will show this.

5.5 Lemma. Let $\mathfrak{X} = (X, \tau)$ be a topological space. If \mathfrak{X} is zero-dimensional, then so is $\mathbb{C}_{st}\mathfrak{X}$. If \mathfrak{X} is a Hausdorff space, then so is $\mathbb{C}_{st}\mathfrak{X}$.

Proof. $\mathbb{C}_{st}\mathfrak{X}$ is zero-dimensional. Follows from the fact that the topology on $\mathbb{C}_{st}\mathfrak{X}$ is generated by a clopen subbase.

 $\mathbb{C}_{st}\mathfrak{X}$ is Hausdorff. Suppose \mathfrak{X} is a Hausdorff space and $h \neq h'$ in $\mathbb{C}_{st}\mathfrak{X}$. Then $h(a) \neq h'(a)$ for some $a \in \mathbb{Clp}\mathfrak{X}$. Without loss of generality assume that there is $x \in h(a)$ such that $x \notin h'(a)$. Since \mathfrak{X} is a Stone space and h'(a) is closed in \mathfrak{X} , there is a clopen $b \in \mathbb{Clp}\mathfrak{X}$ such that $x \in b \subseteq X \setminus h'(a)$. Let $d \coloneqq X \setminus b$, then $h'(a) \subseteq d$ but $h(a) \notin d$, because $x \notin d$. Therefore $h' \in \boxplus(a,d)$ and $h \notin \boxplus(a,d)$. Furthermore $h' \notin \otimes (a,b)$ and $h \notin \otimes (a,b)$, so $\mathbb{C}_{st}\mathfrak{X}$ is a Hausdorff space.

Showing that $\mathbb{C}_{st}\mathfrak{X}$ is a Stone space whenever \mathfrak{X} is a Stone space is more involved.

5.6 Lemma. Let $\mathfrak{X} = (X, \tau)$ be a Stone space. Then $\mathbb{C}_{st}\mathfrak{X}$ is compact.

Proof. Let \mathbb{C}' be the collection of all functions $h : \mathbb{Clp} \mathfrak{X} \to K\mathfrak{X}$ topologized by the subbase $\boxplus(a, b) := \{h \in \mathbb{C}'\mathfrak{X} \mid h(a) \subseteq b\}$ and $\otimes(a, b) := \{h \in \mathbb{C}'\mathfrak{X} \mid h(a) \cap b \neq \emptyset\}$, where a, b range over $\mathbb{Clp}\mathfrak{X}$. That is, we do not require the elements of $\mathbb{C}'\mathfrak{X}$ to satisfy (C1) and (C2) from definition 5.3. We will first show that $\mathbb{C}'\mathfrak{X}$ is compact, and then that $\mathbb{C}_{st}\mathfrak{X}$ is a closed subset of $\mathbb{C}'\mathfrak{X}$.

5.6.A Claim. Let $\mathfrak{X} = (X, \tau)$ be a compact topological space. Then $\mathbb{C}'\mathfrak{X}$ is compact as well.

Proof of claim. Suppose

$$\mathbb{C}'\mathfrak{X} = \bigcup_{i \in I} \boxplus (a_i, b_i) \cup \bigcup_{j \in J} \otimes (c_j, d_j)$$
(5.1)

is a cover. Let $A = \{a_i \mid i \in I\}$ and define B, C, D similarly. Let $D_c = \{d \in D \mid \&(c, d) \text{ is in the cover}\}$. Define

$$h: \mathbb{C} lp \mathfrak{X} \to K \mathfrak{X}: a \mapsto \begin{cases} \varnothing & \text{if } a \notin A \\ X & \text{if } a \in A \text{ and } a \notin C \\ X \smallsetminus \bigcup D_a & \text{if } a \in A \cap C \end{cases}$$

This element is in $\mathbb{C}'\mathfrak{X}$, so it must be in the cover (5.1).

If $h \in \otimes(c_j, d_j)$ for some $j \in J$, then we must have $h(c_j) \cap d_j \neq \emptyset$. If $c_j \notin A$ then $h(c_j) = \emptyset$ hence $h(c_j) \cap d_j = \emptyset$. So this cannot be the case. Therefore we must have $c_j \in A$ and $h(c_j) = (X \setminus \bigcup D_{c_j}) \cap d_j \neq \emptyset$. Since $d_j \in D_{c_j}$, this is a contradiction; see figure 5.1. So we must have $h \in \boxplus(a_i, \beta_i)$ for some $i \in I$.



Figure 5.1: $(X \setminus \bigcup D_{c_j}) \cap d_j = \emptyset$

If $h \in \boxplus(a_i, b_i)$ for some $i \in I$, then we must have $h(a_i) \subseteq b_i$. If $a_i \notin C$ then $h(a_i) = X$ hence $b_i = X$ (for otherwise $h \notin \boxplus(a_i, b_i)$) and we are done, because $\boxplus(a_i, X) = \mathbb{C}' \mathfrak{X}$. So suppose this is not the case, then we must have $h(a_i) = X \setminus \bigcup D_{a_i} \subseteq b_i$. This implies $b \cup \bigcup D_a = \mathfrak{X}$, so by compactness of \mathfrak{X} we find a finite number $d_1, \ldots, d_n \in D_a$ such that

$$\mathcal{B} \cup \mathcal{A}_1 \cup \cdots \cup \mathcal{A}_n = \mathfrak{X}$$

As a result

$$\boxplus (a, b) \cup \otimes (a, d_1) \cup \dots \cup \otimes (a, d_n) \tag{5.2}$$

 \diamond

is a finite subcover of (5.1). (By construction all basic clopens in (5.2) are indeed basic clopens in the cover (5.1).) To see this, we need to show that the finite cover in (5.2) indeed covers all of $\mathbb{C}' \mathfrak{X}$. Let $g \in \mathbb{C}' \mathfrak{X}$ be any element, then either g(a) touches one of the d_i or $g(a) \subseteq b$, so (5.2) is indeed a cover.

It follows from the Alexander subbase theorem that $\mathbb{C}'\mathfrak{X}$ is compact.

5.6.B Claim. Let \mathfrak{X} be a Stone space. Then $\mathbb{C}_{st}\mathfrak{X}$ is a closed subspace of $\mathbb{C}'\mathfrak{X}$.

Proof of claim. Let $h \in \mathbb{C}' \mathfrak{X}$ such that $h \notin \mathbb{C}_{st} \mathfrak{X}$. Then either there exists $a \in \mathbb{Clp} \mathfrak{X}$ such that $h(a) \notin a$ or there are $a \subseteq \beta$ in $\mathbb{Clp} \mathfrak{X}$ such that $h(\beta) \subseteq a$ and $h(a) \neq h(\beta)$.

In the first case, $\otimes(a, X \times a)$ is an open neighbourhood of h in $\mathbb{C}'\mathfrak{X}$ disjoint from $\mathbb{C}_{st}\mathfrak{X}$ and we are done. In the second case we consider two subcases: If $h(a) \notin h(b)$ there exists $x \in h(a)$ such that $x \notin h(b)$. Since \mathfrak{X} is (compact Hausdorff hence) regular and has a clopen subbase we can find disjoint clopens c and d such that $x \notin c$ and $h(b) \subseteq d$ (see figure 5.2). But then

$$h \in \boxplus(\mathcal{B}, a) \cap \otimes(a, c) \cap \boxplus(\mathcal{B}, d),$$

which is open in $\mathbb{C}'\mathfrak{X}$. Moreover, it is disjoint from $\mathbb{C}_{st}\mathfrak{X}$ because $a \subseteq b$ and for all $h' \in \boxplus(b, a) \cap \otimes(a, c) \cap \boxplus(b, d)$ we have $h(b) \subseteq a$ and $h(a) \neq h(b)$.

If $h(b) \notin h(a)$ we can use similar reasoning to find disjoint clopens c and d such that $h(a) \subseteq c$ and $x \in d$. Then the clopen set $\boxplus(b,a) \cap \boxplus(a,c) \cap \otimes(b,d)$ contains h and is disjoint from $\mathbb{C}_{st}\mathfrak{X}$. We conclude that $\mathbb{C}'\mathfrak{X} \setminus \mathbb{C}_{st}\mathfrak{X}$ is open, so $\mathbb{C}_{st}\mathfrak{X}$ is closed in $\mathbb{C}'\mathfrak{X}$.



Figure 5.2: The sets h(a), h(b), c and d in \mathfrak{X} .

Since closed subspaces of compact spaces are compact, this proves that $\mathbb{C}_{st}\mathfrak{X}$ is compact.

We now work our way towards proving that the category of descriptive conditional frames is isomorphic to the category $\mathsf{Coalg}(\mathbb{C}_{\mathsf{st}})$ of \mathbb{C}_{st} -coalgebras.

5.7 Lemma. Let $\mathfrak{X} = (X, \nu, A)$ be a descriptive conditional frame. Let τ_A be the topology on X generated by A and set $\mathfrak{X} = (X, \tau_A)$. Define $\gamma_{\nu} : X \to (\mathbb{Clp} \mathfrak{X} \to K\mathfrak{X})$ by $\gamma_{\nu}(x)(a) := \nu(x, a)$. Then (\mathfrak{X}, γ) is a \mathbb{C}_{st} -coalgebra.

Proof. The topological space \mathfrak{X} is zero-dimensional because it has a clopen basis, compact because the frame \mathfrak{X} is compact and Hausdorff because \mathfrak{X} is differentiated. So \mathfrak{X} is a Stone space.

We have $\gamma_{\nu}(x)(a) \in K\mathfrak{X}$ for all $a \in A$ because \mathfrak{X} is closed (that is, $\nu(x,a) \in K\mathfrak{X}$ for all $\in A$). Also $\gamma_{\nu}(x)$ satisfies (C1) and (C2) from definition 5.3 because ν satisfies conditions (i) and (ii) from definition 5.1.

We need to show that γ_{ν} is continuous. Since γ_{ν}^{-1} preserves taking complements, unions and intersections, it suffices to show that $\gamma_{\nu}^{-1}(\boxplus(a, \beta)) \in A$ for all $a, \beta \in A$. Let $\boxplus(a, \beta)$ be a clopen set in $\mathbb{C}_{st}\mathfrak{X}$, then

$$\gamma_{\nu}^{-1}(\boxplus(a, b)) = \{x \in X \mid \gamma_{\nu}(x) \in \boxplus(a, b)\}$$
$$= \{x \in X \mid \gamma_{\nu}(x)(a) \subseteq b\}$$
$$= \{x \in X \mid \nu(x, a) \subseteq b\}$$
$$= m_{\nu}(a, b) \in A.$$

This shows that γ_{ν} is continuous.

5.8 Lemma. Let $\mathfrak{X} = (X, \tau)$ and let (\mathfrak{X}, γ) be a \mathbb{C}_{st} -coalgebra. Define $\nu_{\gamma} : X \times \mathbb{Clp} \mathfrak{X} \to \mathbb{P}X$ by $\nu_{\gamma}(x, a) := \gamma(x)(a)$. Then $(X, \nu_{\gamma}, \mathbb{Clp} \mathfrak{X})$ is a descriptive conditional frame.

Proof. Evidently, ν_{γ} is a map from $X \times \mathbb{Clp} \mathfrak{X}$ to $\mathbb{P}X$ that satisfies (i) and (ii) from definition 5.1 (because γ satisfies (C1) and (C2) from definition 5.3). It is obvious that $\mathbb{Clp}\mathfrak{X}$ contains \emptyset and X and that it is closed under taking finite unions, finite intersections and complements. Let $a, b \in \mathbb{Clop}\mathfrak{X}$, then

$$m_{\nu_{\gamma}}(a, b) = \{x \in X \mid \nu_{\gamma}(x, a) \subseteq b\}$$

= $\{x \in X \mid \gamma(x)(a) \subseteq b\}$
= $\{x \in X \mid \gamma(x) \in \boxplus(a, b)\}$
= $\gamma^{-1}(x)(\boxplus(a, b)).$

This is clopen by continuity of γ . So $(X, \nu_{\gamma}, \mathbb{Clp} \mathfrak{X})$ is a general conditional frame.

The frame $(X, \nu_{\gamma}, \mathbb{Clp} \mathfrak{X})$ is closed because $\nu_{\gamma}(x, a) = \gamma(x)(a) \in K\mathfrak{X}$ for all $x \in X, a \in \mathbb{Clop} \mathfrak{X}$. It is compact and differentiated because \mathfrak{X} is compact and Hausdorff. Thus $(X, \nu_{\gamma}, \mathbb{Clp} \mathfrak{X})$ is descriptive.

It is obvious that the previous two lemmas establish a 1-1 correspondence between objects of DCF and objects of $Coalg(\mathbb{C}_{st})$. From now on we will use the terms descriptive conditional frame and \mathbb{C}_{st} -coalgebra interchangeably.

5.9 Proposition. Let (\mathfrak{X}, γ) and (\mathfrak{X}', γ') be two \mathbb{C}_{st} -coalgebras and let (X, ν, A) and (X', ν', A') be their corresponding descriptive conditional frames. Let $f : X \to X'$ be a map between sets. Then f is a general conditional frame morphism if and only if it is a \mathbb{C}_{st} -coalgebra morphism.

Proof. For all $x \in X$, $a \in A'$ we have

$$(\mathbb{C}_{st}f)(\gamma(x))(a) = f[\gamma(x)(f^{-1}(a))] = f[\nu(x, f^{-1}(a))].$$
(5.3)

Suppose f is a general conditional frame morphism, then f is continuous because $f^{-1}(a) \in A$ for all $a \in A'$ and it is a \mathbb{C}_{st} -coalgebra morphism because for all $x \in X$ and $a \in A'$ we have

$$\mathbb{C}_{st} f \circ \gamma(x)(a) = f[\nu(x, f^{-1}(a))] = \nu'(f(x), a) = \gamma'(f(x))(a)$$

The first equality holds by (5.3), the second one because f is a general conditional frame morphism, the third one by the correspondence between objects of DCF and $Coalg(\mathbb{C}_{st})$ given in lemma 5.7 and 5.8.

Conversely, suppose f is a \mathbb{C}_{st} -coalgebra morphism. Then $f^{-1}(a) \in A$ for all $a \in A'$ by continuity of f. Furthermore f is a general conditional frame morphism because for all $x \in X$ and $a \in A'$ we have

$$f[\nu(x, f^{-1}(a))] = \mathbb{C}_{st} f \circ \gamma(x)(a) = \gamma' \circ f(x)(a) = \nu'(f(x), a).$$

The first equality holds by (5.3), the second because f is a \mathbb{C}_{st} -coalgebra morphism and the third one because of the correspondence between objects given in lemma 5.7 and 5.8. This proves the proposition.

5.10 Theorem. We have an isomorphism of categories

$$\mathsf{DCF} \cong \mathsf{Coalg}(\mathbb{C}_{\mathsf{st}})$$

Proof. Follows from lemma 5.7 and 5.8 and proposition 5.9.

It follows directly from the definitions that descriptive conditional models are precisely \mathbb{C}_{st} -models. It is an easy exercise to show that general conditional frame morphisms are \mathbb{C}_{st} -model morphisms. This yields the following result:

5.11 Theorem. There is an isomorphism of categories

$$\mathsf{DCM} \cong \mathsf{Mod}(\mathbb{C}_{\mathsf{st}})$$

From now on we will identify descriptive conditional models and \mathbb{C}_{st} -models, and use the terminology interchangeably. We finish this subsection with a definition and remark about the conditional language on descriptive conditional models.

5.12 Definition. Define the clopen predicate liftings $\lambda^{\Rightarrow}, \lambda^{\downarrow} : \mathbb{Clp}^2 \to \mathbb{Clp} \circ \mathbb{C}_{st}$ by

$$\lambda_{\mathfrak{X}}^{\Rightarrow}(a, b) = \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid h(a) \subseteq b\} \quad \text{and} \quad \lambda_{\mathfrak{X}}^{\Downarrow}(a, b) = \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid h(a) \cap b \neq \emptyset\}.$$

Then $\mathcal{L}(\lambda^{\Rightarrow}, \lambda^{\downarrow})$ is given by

$$\varphi \coloneqq \perp \mid p \mid \varphi \land \psi \mid \varphi \Rightarrow \psi \mid \varphi \Downarrow \psi,$$

where $p \in \Phi$.

5.13 Remark. Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ be a \mathbb{C}_{st} -model, then

$$\mathfrak{X}, x \Vdash \varphi \Rightarrow \psi \quad \text{iff} \quad \gamma(x) \in \lambda^{\Rightarrow}(\llbracket \varphi \rrbracket^{\mathfrak{X}}, \llbracket \psi \rrbracket^{\mathfrak{X}})$$

and

$$\mathfrak{X}, x \Vdash \varphi \Downarrow \psi \quad \text{iff} \quad \gamma(x) \in \lambda^{\Downarrow}(\llbracket \varphi \rrbracket^{\mathfrak{X}}, \llbracket \psi \rrbracket^{\mathfrak{X}}).$$

Indeed, $\mathfrak{X}, x \Vdash \varphi \Rightarrow \psi$ iff $\mathfrak{X}, x \Vdash \neg(\varphi \Downarrow \neg \psi)$.

Moreover, the topology on $\mathbb{C}_{st}\mathfrak{X}$ is generated by $\lambda^{\Rightarrow}(a, b), \lambda^{\Downarrow}(a, b)$ where $a, b \in \mathbb{Clp}\mathfrak{X}$, so $\{\lambda^{\Rightarrow}, \lambda^{\Downarrow}\}$ is a characteristic set of clopen predicate liftings for \mathbb{C}_{st} . This allows us to use various results from chapter 2. In particular, it follows from proposition 2.29 that modal equivalence coincides with behavioural equivalence.

5.1.2 DUALITY

The goal of this section is to give a functor \mathbb{A}_{st} on BA, the category of Boolean algebras, such that the following diagram commutes,

We present a functor on BA and then show that it is the dual of \mathbb{C}_{st} .

5.14 Definition. Let *B* be a Boolean algebra. Let $\mathbb{A}_{st}B$ be the Boolean algebra generated by the elements $\Box(a, b), \Diamond(a, b)$ where $a, b \in B$ subject to the relations

- $(A1) \Box(a,a) = \mathsf{T}_{\mathbb{A}B} \qquad (A2) \diamondsuit(a, \bot_B) = \bot_{\mathbb{A}B}$
- $(A3) \Box(a, b) \land \Box(a, c) = \Box(a, b \land c) \qquad (A4) \diamondsuit(a, b) \lor \diamondsuit(a, c) = \diamondsuit(a, b \lor c)$

(A5)
$$\Box(a, b \lor c) \le \Box(a, b) \lor \diamondsuit(a, c)$$
 (A6) $\Box(a, b) \land \diamondsuit(a, c) \le \diamondsuit(a, b \land c)$

 $\begin{array}{ll} (\text{A7}) & \Box(a, b) \land \Box(a, c) = \Box(a, b) \land \Box(b, c) & (\text{A8}) & \Box(a, b) \land \diamondsuit(a, c) = \Box(a, b) \land \diamondsuit(b, c) \\ & \text{if } b \leq a & \text{if } b \leq a. \end{array}$

If $f: A \to B$ is a Boolean algebra morphism, define $\mathbb{A}_{st}f: \mathbb{A}_{st}A \to \mathbb{A}_{st}B$ on generators by $\mathbb{A}_{st}f(\Box(a, b)) = \Box(f(a), f(b))$ and $\mathbb{A}_{st}f(\diamondsuit(a, b)) = \diamondsuit(f(a), f(b))$. (One can easily see that the images of the generators of A under $\mathbb{A}_{st}f$ satisfy relations (A1) through (A8), so by remark 3.3 $\mathbb{A}_{st}f$ indeed defines a frame homomorphism.) Then \mathbb{A}_{st} defines an endofunctor on BA.

 \triangleleft

5.15 Remark. Relations (A5) and (A6) together imply $\Box(a, b) = \neg \diamond (a, \neg b)$.

Although it would have been possible to define $\mathbb{A}_{st}B$ using only the boxes as generators, we have chosen for this presentation, as it provides a smoother transition to the geometric case in section 5.2. Furthermore, we identify ultrafilters of a Boolean algebra B with homomorphisms from B to $2 = \{\mathsf{T}, \bot\}$.

5.16 Lemma. There is a one-one correspondence between ultrafilters on a Boolean algebra B and (Boolean algebra) homomorphisms $p: B \rightarrow 2$.

The proof of the next theorem is very similar to the proof of theorem 5.38, so we only give a proof sketch here. We decided to spell out the proof of theorem 5.38 because it is slightly more involved than the proof of the proposition below.

5.17 Proposition. Let \mathfrak{X} be a Stone space. Then $\mathfrak{uf}(\mathbb{A}_{st}(\mathfrak{clp}\mathfrak{X})) \cong \mathbb{C}_{st}\mathfrak{X}$.

Proof. We view ultrafilters of $\mathbb{A}_{st}(\mathfrak{clp} \mathfrak{X})$ as Boolean algebra homomorphisms $p : \mathbb{A}_{st}(\mathfrak{clp} \mathfrak{X}) \to 2$. Define a map

$$\varphi: \mathbb{C}_{\mathsf{st}} \mathfrak{X} \to \mathfrak{uf}(\mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X})): h \mapsto p_h,$$

where p_h is defined on generators by

$$p_h : \mathbb{A}_{\mathsf{St}}(\mathfrak{clp}\,\mathfrak{X}) \to 2 : \begin{cases} \Box(a, \delta) \mapsto \top & \text{iff } h(a) \subseteq \delta \\ \diamondsuit(a, \delta) \mapsto \top & \text{iff } h(a) \cap \delta \neq \emptyset \end{cases}$$

Conversely, define

$$\psi:\mathfrak{uf}(\mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X}))\to\mathbb{C}_{\mathsf{st}}\mathfrak{X}$$

as follows: For an ultrafilter $p: \mathbb{A}_{\mathsf{st}}(\mathfrak{clp} \mathfrak{X}) \to 2$ let $h_p: \mathfrak{clp} \mathfrak{X} \to K\mathfrak{X}$ be the map given by

$$h_p(a) = \mathfrak{X} \setminus \bigcup \{ \mathscr{b} \in \mathfrak{clp} \, \mathfrak{X} \mid p(\diamondsuit(a, \mathscr{b})) = \bot \}.$$

In order to show that these maps give rise to an isomorphism, we need to show that both φ and ψ are well defined, that they are mutually inverse and that φ is continuous. This is completely similar to the four claims in theorem 5.38.

The map φ from the previous theorem yields a map $\eta_{\mathfrak{X}} : \mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\mathfrak{X}) \to \mathfrak{clp}(\mathbb{C}_{\mathsf{st}}\mathfrak{X})$.

5.18 Definition. For a Stone space \mathfrak{X} let $\eta_{\mathfrak{X}}$ be the concatenation of \mathfrak{clp} , \mathfrak{uf} and $\mathfrak{clp}\varphi$, where φ is defined as in theorem 5.17, that is,

$$\eta_{\mathfrak{X}}: \mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X}) \xrightarrow{\mathfrak{clp}\circ\mathfrak{uf}} \mathfrak{clp}(\mathfrak{uf}(\mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X}))) \xrightarrow{\mathfrak{clp}\,\varphi} \mathfrak{clp}(\mathbb{C}_{\mathsf{st}}\mathfrak{X})$$

A routine calculation reveals that this map is given by $\Box(a, b) \mapsto \boxplus(a, b)$ and $\Diamond(a, b) \mapsto \Diamond(a, b)$.

The collection of isomorphisms $\eta_{\mathfrak{X}}$ constitute a natural isomorphism.

5.19 Proposition. The collection $\eta = (\eta \chi)_{\chi \in \text{Stone}}$ give a natural isomorphism

$$\eta: \mathbb{A}_{\mathsf{st}} \circ \mathfrak{clp} \to \mathfrak{clp} \circ \mathbb{C}_{\mathsf{st}}.$$

Proof. It follows from proposition 5.17 that $\eta_{\mathfrak{X}}$ is an isomorphism for each Stone space \mathfrak{X} . So we need to show naturality of η , that is, for all continuous maps $f: \mathfrak{X} \to \mathfrak{X}'$ in Stone, the following diagram commutes

$$\begin{array}{c} \mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X}_{\mathsf{st}}) & \stackrel{\mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,f)}{\longleftarrow} \mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,\mathfrak{X}') \\ \eta_{\mathfrak{X}} & \downarrow \\ \mathfrak{clp}(\mathbb{C}_{\mathsf{st}}\mathfrak{X}) & \stackrel{\mathbb{A}_{\mathsf{st}}(\mathfrak{clp}\,f)}{\longleftarrow} \mathfrak{clp}(\mathbb{C}_{\mathsf{st}}\mathfrak{X}') \end{array}$$

Let $\Box(a, b)$ be a generator in $\mathbb{A}_{\mathsf{st}}(\mathfrak{clp} \mathfrak{X}')$. Then

$$\begin{aligned} \mathfrak{clp}(\mathbb{C}_{\mathsf{st}}f)(\eta_{\mathfrak{X}'}(\square(a, \delta))) &= (\mathbb{C}_{\mathsf{st}}f)^{-1}(\boxplus(a, \delta)) & \text{Def of } \mathfrak{clp} \text{ and } \eta \\ &= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid (\mathbb{C}_{\mathsf{st}}f)(h) \in \boxplus(a, \delta)\} \\ &= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid (\mathbb{C}_{\mathsf{st}}f)(h)(a) \subseteq \delta\} & \text{Definition of } \boxplus \\ &= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid f[h(f^{-1}(a))] \subseteq \delta\} & \text{Definition of } \mathbb{C}f \\ &= \{h \in \mathbb{C}_{\mathsf{st}}\mathfrak{X} \mid h(f^{-1}(a)) \subseteq f^{-1}(\delta)\} \\ &= \boxplus(f^{-1}(a), f^{-1}(\delta)) & \text{Definition of } \boxplus \\ &= \eta_{\mathfrak{X}}(\square(f^{-1}(a), f^{-1}(\delta))) & \text{Definition of } \eta \\ &= \eta_{\mathfrak{X}}(\mathbb{A}_{\mathsf{st}}(\mathfrak{clp} f)(\square(a, \delta))) & \text{Def of } \mathfrak{clp} \text{ and } \mathbb{A}_{\mathsf{st}} \end{aligned}$$

The \diamond case follows from the fact that $\diamond(a, b) = \neg \Box (a, \neg b)$. Therefore η is a natural isomorphism.

5.20 Corollary. There is a dual equivalence

$$Alg(\mathbb{A}_{st}) \equiv^{op} Coalg(\mathbb{C}_{st}).$$

Proof. Follows from proposition 5.19 and lemma A.6.

5.1.3 BISIMULATIONS BETWEEN DESCRIPTIVE CONDITIONAL MODELS

In [7] the authors introduce a notion of conditional bisimilarity between conditional models. We modify their notion slightly to work well with \mathbb{C}_{st} -models (descriptive conditional models).

5.21 Definition. Let (\mathfrak{X}, γ) and (\mathfrak{X}', γ') be two descriptive conditional frames (viewed as \mathbb{C}_{st} -coalgebras). A closed subset $B \subseteq \mathfrak{X} \times \mathfrak{X}'$ is a **descriptive conditional bisimulation** if, for all *B*-coherent sets of clopens (a, a') and $(x, x') \in B$ we have

•
$$\gamma(x)(a) \subseteq B^{-1}[\gamma'(x')(a')]$$
 and $\gamma'(x')(a') \subseteq B[\gamma(x)(a)]$.

A conditional bisimulation between descriptive conditional models $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ is a conditional bisimulation between the underlying frames with the extra condition that

• $x \in V(p)$ iff $x' \in V'(p)$ for all $p \in \Phi$.

Two states in two descriptive conditional frames or models are called **descriptive conditional bisimilar** if there is a conditional bisimulation linking them. \triangleleft

5.22 Remark. We can rephrase the first bullet in definition 5.21 as follows: for all *B*-coherent pairs of clopens (a, a') and $(x, x') \in B$ we have

• for all $y \in \gamma(x)(a)$ there exists $y' \in \gamma'(x')(a')$ such that $(y, y') \in B$, and vice versa.

This wording resembles more the formulation of definition 5 in [7].

In section 2.3 we defined Λ -bisimulations for Stone coalgebras. A natural question is how these relate to descriptive conditional bisimulations. It turns out that, in terms of bisimilarity, they coincide. The remainder of this subsection is devoted to proving the following theorem.

5.23 Theorem. Let $(\mathfrak{X}, \gamma, V)$ and $(\mathfrak{X}', \gamma', V')$ be two descriptive conditional models, $x \in X$ and $x' \in X'$. Then the following are equivalent:

- (i) x and x' are descriptively conditionally bisimilar;
- (ii) x and x' are $\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\}$ -bisimilar;
- (iii) x and x' are modally equivalent;
- (iv) x and x' are behaviourally equivalent.

As a corollary of this theorem (combined with proposition 2.33) we find the following relation of descriptive conditional bisimilarity to Aczel-Mendler bisimilarity.

5.24 Corollary. If two states x and x' in two descriptive conditional frames are Aczel-Mendler bisimilar, then they are descriptive conditional bisimilar.

We isolate some of the implications of theorem 5.23 as separate propositions.

5.25 Proposition. Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}, \gamma', V')$ be two descriptive conditional models and $B \subseteq \mathfrak{X} \times \mathfrak{X}$ a descriptive conditional bisimulation. Then B is a $\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\}$ -bisimulation.

Proof. Let $(x, x') \in B$. By definition $x \in V(p)$ iff $x' \in V'(p)$. Suppose (a, a') and (b, b') are *B*-coherent. If $\gamma(x) \in \lambda_{\mathfrak{X}}^{\Rightarrow}(a, b)$ then $\gamma(x)(a) \subseteq b$ and hence

$$\gamma'(x')(a') \subseteq B[\gamma(x)(a)] \subseteq B[\delta] \subseteq \delta', \tag{5.4}$$

so $\gamma'(x') \in \lambda_{\chi'}(a', \beta')$. The first inclusion in (5.4) follows from the fact that *B* is a descriptive conditional bisimulation, the last one from the coherence of (β, β') . The converse direction is proven similarly. For λ^{\downarrow} , recall that lemma 2.18 states that $(\mathfrak{X} \times \beta, X' \times \beta')$ is *B*-coherent whenever (β, β') is, so

$$\begin{split} \gamma(x) \in \lambda_{\mathfrak{X}}^{\Downarrow}(a, \beta) & \text{iff} \quad \gamma(x) \notin \lambda_{\mathfrak{X}}^{\rightrightarrows}(a, \mathfrak{X} \smallsetminus \beta) \\ & \text{iff} \quad \gamma'(x') \notin \lambda_{\mathfrak{X}'}^{\rightrightarrows}(a, \mathfrak{X} \smallsetminus \beta) \quad \text{iff} \quad \gamma'(x') \in \lambda_{\mathfrak{X}'}^{\Downarrow}(a', \beta'). \end{split}$$

This proves the proposition.

5.26 Proposition. Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ and $\mathfrak{X}' = (\mathfrak{X}', \gamma', V')$ be two descriptive conditional models. Let $x \in X$ and $x' \in X'$ be modally equivalent. Then they are descriptively conditionally bisimilar.



Figure 5.3: The image of $\mathbb{U}\mathfrak{X} \smallsetminus a$ under B is \mathfrak{G}' .

Proof. Let $B \subseteq \mathfrak{X} \times \mathfrak{X}'$ be the relation of modal equivalence. We will show that B is a descriptive conditional bisimulation. Let $w \in X$ and $w' \in X'$ be modally equivalent and let $(a, a') \in \mathbb{Clp}(\mathfrak{X}) \times \mathbb{Clp}(\mathfrak{X}')$ be *B*-coherent. Assume towards a contradiction that *B* does not satisfy the bisimulation property for *a* and *a'*, then we claim that this implies that *w* and *w'* are not modally equivalent. We will show this in a couple of steps:

- Step 1. Show that B is closed in $\mathfrak{X} \times \mathfrak{X}'$.
- Step 2. Build a formula α that will play the role of a and a' in the antecedent. That is $a \subseteq \llbracket \alpha \rrbracket^{\mathfrak{X}} \subseteq a \cup \mathfrak{r}$ with \mathfrak{r} a set of "non-relevant" states, and similarly $a' \subseteq \llbracket \alpha \rrbracket^{\mathfrak{X}'} \subseteq a' \cup \mathfrak{r}'$.
- Step 3. Show that $\gamma(w)(\llbracket \alpha \rrbracket^{\mathfrak{X}}) = \gamma(w)(a)$ and $\gamma'(w')(\llbracket \alpha \rrbracket^{\mathfrak{X}'}) = \gamma(w')(a')$.
- Step 4. Find a contradiction with the assumption that B does not satisfy the bisimulation property for a and a'.

Step 1. Suppose $(x, x') \notin B$, then there exists φ such that $\mathfrak{X}, x \Vdash \varphi$ and $\mathfrak{X}', x' \Vdash \neg \varphi$, so $(x, x') \in \llbracket \varphi \rrbracket^{\mathfrak{M}} \times \llbracket \neg \varphi \rrbracket^{\mathfrak{M}'}$ which is open in $\mathfrak{X} \times \mathfrak{X}'$ and clearly disjoint from B. Therefore B is closed in $\mathfrak{X} \times \mathfrak{X}'$.

Step 2. Let $\[mbox{\boldmath δ} = B^{-1}(\[mbox{\boldmath X}' \times a')\]$ and $\[mbox{\boldmath τ} = \[mbox{\boldmath X} \times (a \cup b)\]$. Observe that $\[mbox{\boldmath b}\]$ is disjoint from a, for otherwise $B[a] \notin a'$. Moreover, elements in $\[mbox{\boldmath b}\]$ are not modally equivalent to elements in a, because this would imply that $B[a] \notin a'$. Observe also that elements in $\[mbox{\boldmath τ}\]$ are not related to anything in $\[mbox{\boldmath X}',\]$ that is, they are not modally equivalent to any element in $\[mbox{\boldmath X}'.\]$ With this notation, $\[mbox{\boldmath W}\]$ is the disjoint union of $a, \[mbox{\boldmath b}\]$ and $\[mbox{\boldmath $\vmath τ},\]$

$$\mathbb{U}\mathfrak{X} = a \cup b \cup r.$$

In a similar way we can write $\mathbb{U}\mathfrak{X}' = a' \cup b' \cup z'$. See figure 5.3.

The set β is closed in \mathfrak{X} . To see this, note that

$$\mathcal{B} = B^{-1}[\mathcal{X}' \setminus a'] = p[(\mathcal{X}' \times a') \cap B],$$

where $p: \mathfrak{X} \times \mathfrak{X}' \to \mathfrak{X}$ is projection to the first coordinate. Observe that $(\mathfrak{X} \times a') \cap B$ is closed in $\mathfrak{X} \times \mathfrak{X}'$. As \mathfrak{X}' is compact, p is a closed map and \mathfrak{b} is closed in \mathfrak{X} . For a similar reason \mathfrak{b}' is closed in \mathfrak{X}' .

Since elements of a are not modally equivalent to elements of $b \cup (\mathfrak{X}' \smallsetminus a')$, for each $x \in a$ and $y \in b \cup b' \cup t'$ there exists $\varphi_{x,y}$ such that $\mathfrak{X}, x \Vdash \varphi_{x,y}$ and $\mathfrak{X}, y \Vdash \neg \varphi_{x,y}$ when $y \in b$ and $\mathfrak{X}', y \Vdash \neg \varphi_{x,y}$ when $y \in b' \cup t'$. Fix $x \in a$. Then $b \subseteq \bigcup_{y \in b} [\![\neg \varphi_{x,y}]\!]^{\mathfrak{X}}$ is an open

covering of the closed set \mathcal{B} . By compactness there exists a finite subset $I_x \subseteq \mathcal{B}$ such that $\mathcal{B} \subseteq \bigcup_{y \in I_x} [\![\neg \varphi_{x,y}]\!]^{\mathfrak{X}}$. Similarly we have $\mathcal{B}' \cup \mathfrak{c}' \subseteq \bigcup_{y \in \mathcal{B}' \cup \mathfrak{c}'} [\![\varphi_{x,y}]\!]^{\mathfrak{X}'}$ and by compactness we find a finite $I'_x \subseteq \mathcal{B}' \cup \mathfrak{c}'$ such that $\mathcal{B}' \cup \mathfrak{c}' \subseteq \bigcup_{y \in I'_x} [\![\varphi_{x,y}]\!]^{\mathfrak{X}'}$ is a finite subcover. Set

$$\chi_x\coloneqq \bigwedge_{y\in I_x\cup I_x'}\varphi_{x,y}.$$

Then $\mathfrak{X}, x \Vdash \chi_x$ and for all $y \in \mathfrak{b} \cup \mathfrak{b}' \cup \mathfrak{c}'$ we have $y \not\models \chi_x$. Next, we find an open covering

$$a \subseteq \bigcup_{x \in a} \llbracket \chi_x \rrbracket^{\mathfrak{X}}$$

and since a is closed, hence compact, there exists a finite set $J \subseteq a$ such that $a \subseteq \bigcup_{x \in J} [\![\chi_x]\!]^{\mathfrak{X}}$. Set

$$\chi \coloneqq \bigvee_{x \in J} \chi_x.$$

Then χ has the property that $\mathfrak{X}, x \Vdash \chi$ for all $x \in a$ and $\mathfrak{X}, y \not\models \chi$ for $y \in b$ and $\mathfrak{X}', y' \not\models \chi$ for $y' \in b' \cup z'$. In a symmetric way we can define a formula χ' which is true for all $x \in a'$ and false for all $y \in b' \cup b \cup z$.

Let $\alpha \coloneqq \chi \lor \chi'$. Then

$$a \subseteq \llbracket \alpha \rrbracket^{\mathfrak{X}} \subseteq a \cup \mathfrak{r} \quad \text{and} \quad a' \subseteq \llbracket \alpha \rrbracket^{\mathfrak{X}'} \subseteq a' \cup \mathfrak{r}'.$$
 (5.5)

Step 3. Our next goal is to show $\gamma(w)(\llbracket \alpha \rrbracket^{\mathfrak{X}}) = \gamma(w)(a)$. Let $z \in \gamma(w)(\llbracket \alpha \rrbracket^{\mathfrak{X}})$. By condition (C1) from definition 5.3 we know $z \in \llbracket \alpha \rrbracket^{\mathfrak{X}}$ and by equation (5.5) this means either $z \in a$ or $z \in \mathfrak{r}$.

If $z \in z$ then z is not modally equivalent to any $x \in \mathbf{X}'$. Therefore we can make a formula β that is true everywhere in \mathbf{X}' but that is not true on z: For each $x \in \mathbf{X}'$, there exists φ_x such that $\mathfrak{X}, z \Vdash \varphi$ and $\mathfrak{X}', x \not\models \varphi_x$, so the clopen sets $[\![\neg \varphi_x]\!]^{\mathfrak{X}'}$ (where x ranges over \mathbf{X}') form an open cover of \mathbf{X}' . By compactness there exists a finite subcover, say, $\mathbf{X}' = [\![\neg \varphi_1]\!]^{\mathfrak{X}'} \cup \cdots \cup [\![\neg \varphi_n]\!]^{\mathfrak{X}'}$. Set

$$\beta \coloneqq \neg \varphi_1 \lor \cdots \lor \neg \varphi_n,$$

then $\mathfrak{X}, z \not\models \beta$ while $\mathfrak{X}', x \models \beta$ for all $x \in \mathfrak{X}'$. In particular this means $\gamma(w)(\llbracket \alpha \rrbracket^{\mathfrak{X}}) \notin \llbracket \beta \rrbracket^{\mathfrak{X}}$ and $\gamma'(w')(\llbracket \alpha \rrbracket^{\mathfrak{X}'}) \subseteq \llbracket \beta \rrbracket^{\mathfrak{X}'}$. Therefore $\mathfrak{X}, w \not\models \alpha \Rightarrow \beta$ while $\mathfrak{X}', w' \models \alpha \Rightarrow \beta$. A contradiction with the assumption that w and w' are modally equivalent. We conclude that $z \in a$.

As $a \subseteq [\![\alpha]\!]^{\mathfrak{X}}$ and $\gamma(w)([\![\alpha]\!]^{\mathfrak{X}}) \subseteq a$, it follows form condition (C2) from definition 5.3 that

$$\gamma(w)(\llbracket lpha
rbracket^{\mathfrak{X}}) = \gamma(w)(a)$$

In a similar way one may show that $\gamma'(w')(\llbracket \alpha \rrbracket^{\mathfrak{X}'}) = \gamma'(w')(a')$.

Step 4. Now suppose for a contradiction that B does not satisfy the bisimulation property for the sets a and a'. Then without loss of generality, there is $x \in \gamma(w)(a)$ such that for all $x' \in \gamma'(w')(a')$ we have $(x, x') \notin B$. For each $x' \in \gamma'(w')(a')$ let $\varphi_{x'}$ be such that $\mathfrak{X}, x \Vdash \neg \varphi_{x'}$ and $\mathfrak{X}', x' \Vdash \varphi_{x'}$. Then the union

$$\bigcup_{x'\in\gamma'(w')(a')} \llbracket \varphi_{x'} \rrbracket^{\mathfrak{X}}$$

is an open cover of $\gamma'(w')(a')$. Since $\gamma'(w')(a')$ is closed, hence compact, there is a finite set $I \subseteq \gamma'(w')(a')$ such that $\bigcup_{x' \in I} \llbracket \varphi_{x'} \rrbracket^{\mathfrak{X}'}$ covers $\gamma'(w')(a')$. Therefore the formula $\psi = \bigvee_{x' \in I} \varphi_{x'}$ that is false at x and true everywhere in $\gamma'(w', \llbracket \alpha \rrbracket^{\mathfrak{X}'})$. But then $\mathfrak{X}, w \not\models \alpha \Rightarrow \psi$ and $\mathfrak{X}', w' \models \alpha \Rightarrow \psi$, a contradiction. This shows that B must be a descriptive conditional bisimulation.

Recall theorem 5.23; the proof is now easy:

5.23 Theorem. Let $(\mathfrak{X}, \gamma, V)$ and $(\mathfrak{X}', \gamma', V')$ be two descriptive conditional models, $x \in X$ and $x' \in X'$. Then the following are equivalent:

- (i) x and x' are descriptively conditionally bisimilar;
- (ii) x and x' are $\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\}$ -bisimilar;
- (iii) x and x' are modally equivalent;
- (iv) x and x' are behaviourally equivalent.

Proof of theorem 5.23. The implication (i) \Rightarrow (ii) follows from proposition 5.25, (ii) \Rightarrow (iii) is lemma 2.32, (iii) \Rightarrow (i) is proposition 5.26 and (iii) \Leftrightarrow (iv) is proposition 2.29. \Box

A notion of bisimulation is called structural if it does not rely on the truth set of any formulas apart from the propositional variables. Structural bisimulations are preferred over non-structural ones. (For a discussion about this in the scope of conditional logic see e.g. [15].) In particular Λ -bisimulations (see definition 2.30) are non-structural. Theorem 5.23 provides a structural charactarization of $\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\}$ -bisimilarity (and modal equivalence and behavioural equivalence).

5.2 GEOMETRIC CONDITIONAL LOGIC

In this section we develop (the frames for) geometric conditional logic. We first generalise the conditional frames from example 2.10 to geometric conditional frames. We show how geometric conditional frames can be viewed as coalgebras for a functor \mathbb{C}_{kh} . Thereafter, we define two open predicate liftings that correspond to the binary modalities \Rightarrow and \downarrow . In subsection 5.2.2 we give a functor on Frm whose restriction to KRFrm is dual to \mathbb{C}_{kh} .

5.2.1 Geometric conditional frames

5.27 Definition. A topological conditional frame is a triple (X, ν, A) where X is a set, $A \subseteq \mathbb{P}X$ is a topology on X and $\nu : X \times A \to \mathbb{P}X$ is a map, such that

• A is closed under the maps $m_{\nu}, n_{\nu} : A \times \mathbb{P}X \to \mathbb{P}X$ defined by

$$m_{\nu}(a, b) \coloneqq \{x \in X \mid \nu(x, a) \subseteq b\}, \quad n_{\nu}(a, b) \coloneqq \{x \in X \mid \nu(x, a) \cap b \neq \emptyset\},\$$

- $\nu: X \times A \to K \mathfrak{X}$ satisfies for all $x \in X$, $a, b \in A$;
 - (i) if $a \cap b = \emptyset$, then $\nu(x, a) \cap b = \emptyset$,
 - (ii) if $a \subseteq b$ and $\nu(x, b) \subseteq a$, then $\nu(x, a) = \nu(x, b)$.

72

 \triangleleft
5.28 Definition. Let $\mathfrak{X} = (X, \nu, A)$ be a topological conditional frame and denote by \mathfrak{X} the set X topologised by A. We say that \mathfrak{X} is **differentiated** if for all $x, y \in X$ there exist disjoint $a, b \in A$ such that $x \in a$ and $y \in b$, **closed** if $\nu(x, a)$ is closed in \mathfrak{X} for all $x \in X$ and $a \in A$, and **compact** if \mathfrak{X} is compact.

A geometric conditional frame is a compact, closed and differentiated topological conditional frame. A geometric conditional frame morphism from (X, ν, A) to (X', ν', A') is a map $f: X \to X'$ such that $f^{-1}(a') \in A$ for all $a' \in A'$ (i.e. f is continuous) and $f[\nu(x, f^{-1}(a))] = \nu'(f(x), a)$ for all $x \in X, a \in A'$. Write GCF for the category of geometric conditional frames and geometric conditional frame morphisms. \triangleleft

Let us take a coalgebraic perspective.

5.29 Definition. For a topological space \mathfrak{X} , let $\mathbb{C}_{top}\mathfrak{X}$ be the collection of maps $h : \Omega\mathfrak{X} \to K\mathfrak{X}$ such that for all $x \in \mathfrak{X}$ and $a, b \in \Omega\mathfrak{X}$ we have

(C1) if $a \cap b = \emptyset$ then $h(a) \cap b = \emptyset$; and

(C2) if $a \subseteq b$ and $h(b) \subseteq a$ then h(a) = h(b).

Endow $\mathbb{C}_{top} \mathfrak{X}$ with the topology generated by the subbase

$$\boxplus (a, b) \coloneqq \{h \in \mathbb{C}_{top} \mathfrak{X} \mid h(a) \subseteq b\}, \quad \& (a, b) \coloneqq \{h \in \mathbb{C}_{top} \mathfrak{X} \mid h(a) \cap b \neq \emptyset\}$$

where a and β range over the opens of \mathfrak{X} . For a continuous map $f: \mathfrak{X} \to \mathfrak{X}'$ between topological spaces, define $\mathbb{C}_{top} f: \mathbb{C}_{top} \mathfrak{X} \to \mathbb{C}_{top} \mathfrak{X}'$ by

$$\mathbb{C}_{\mathsf{top}}f(h)(a) = \overline{f[h(f^{-1}(a))]}.$$

(The overline denotes closure.) Then \mathbb{C}_{top} defines an endofunctor on Top.

 \triangleleft

5.30 Remark. (i) Condition (C1) can be reformulated as $h(a) \subseteq \overline{a}$. The intuition behind this is as follows: The opens of a topological space will serve as the interpretants of geometric modal formulae and $\nu(x)(\llbracket \varphi \rrbracket)$ indicates the relevant states of $\llbracket \varphi \rrbracket$ for x. If the truth set $\llbracket \varphi \rrbracket$ of some formula φ is disjoint from the truth set $\llbracket \psi \rrbracket$ of ψ , then we require that $\varphi \Rightarrow \psi$ is false, i.e. $h(\llbracket \varphi \rrbracket) \cap \llbracket \psi \rrbracket = \emptyset$. However, we want to allow the situation where every state in $\llbracket \varphi \rrbracket$ is relevant for x, so $a \subseteq h(a)$. The set h(a) must be closed, and the smallest closed set containing a is the closure \overline{a} . Therefore, we tolerate $h(a) = \overline{a}$.

(ii) If $f : \mathfrak{X} \to \mathfrak{X}'$ is a morphism in KHaus then $f[h(f^{-1}(a))] \subseteq \mathfrak{X}'$ is closed, so $\overline{f[h(f^{-1}(a))]} = f[h(f^{-1}(a))]$. Since *a* is open in \mathfrak{X}' , $f^{-1}(a)$ is open in \mathfrak{X} . By construction $h(f^{-1}(a))$ is closed in \mathfrak{X} , and since \mathfrak{X} is compact and \mathfrak{X}' is Hausdorff, *f* is a closed map so $f[h(f^{-1}(a))]$ is closed in \mathfrak{X}' .

5.31 Lemma. Let $f : \mathfrak{X} \to \mathfrak{X}'$ be a continuous map between topological spaces. Then $\mathbb{C}_{top}f : \mathbb{C}_{top}\mathfrak{X} \to \mathbb{C}_{top}\mathfrak{X}'$ is a well-defined continuous map.

Proof. We need to show that for each $h \in \mathbb{C}_{top} \mathfrak{X}$, the conditions (C1) and (C2) from definition 5.29 hold, and that $\mathbb{C}_{top} f$ is continuous. Checking (C1) and (C2) is similar to example 2.21.

For continuity, it suffices to check that for all $a, b \in \Omega \mathfrak{X}'$, the sets $(\mathbb{C}_{top} f)^{-1}(\boxplus(a, b))$ and $(\mathbb{C}_{top} f)^{-1}(\otimes(a, b))$ are open in $\mathbb{C}_{top}\mathfrak{X}$. It can be shown by a straightforward computation that

$$(\mathbb{C}_{\mathsf{top}}f)^{-1}(\boxplus(a, \delta)) = \boxplus(f^{-1}(a), f^{-1}(\delta)) \quad \text{and} \quad (\mathbb{C}_{\mathsf{top}}f)^{-1}(\otimes(a, \delta)) = \otimes(f^{-1}(a), f^{-1}(\delta))$$

Since f is continuous, $f^{-1}(a)$ and $f^{-1}(b)$ are open in \mathfrak{X} , hence $\boxplus(f^{-1}(a), f^{-1}(b))$ and $\otimes(f^{-1}(a), f^{-1}(b))$ are open in $\mathbb{C}_{top}\mathfrak{X}$. This proves that $\mathbb{C}_{top}f$ is continuous. \Box

5.32 Lemma. If \mathfrak{X} is a compact Hausdorff space, then so is $\mathbb{C}_{top}\mathfrak{X}$.

Proof. \mathbb{C}_{top} is Hausdorff. Suppose $h \neq h'$ in $\mathbb{C}_{top} \mathfrak{X}$, then $h(a) \neq h'(a)$ for some open set a of \mathfrak{X} . Without loss of generality assume there exists $x \in h'(a)$ such that $x \notin h(a)$. Since h(a) is closed and \mathfrak{X} is compact Hausdorff, hence regular, there exist disjoint open neighbourhoods u, v of h(a) and x respectively. Now we have $h \in \boxplus(a, u)$, $h' \notin \boxplus(a, u)$ and $h \notin \otimes (a, v)$, $h' \in \otimes (a, v)$. For any $j \in \mathbb{C}_{top} \mathfrak{X}$, whenever $j \in \boxplus(a, u)$ we have $v \cap j(a) = \emptyset$ so $j \notin \otimes (a, v)$. Therefore $\boxplus(a, u)$ and $\otimes (a, v)$ are disjoint and $\mathbb{C}_{top} \mathfrak{X}$ is Hausdorff.

 \mathbb{C}_{top} is compact. The proof of this is similar to the proof of lemma 5.6.

5.33 Definition. Let \mathbb{C}_{kh} denote the restriction of the functor \mathbb{C}_{top} to KHaus. Lemma 5.32 entails this is an endofunctor on KHaus.

5.34 Proposition. GCF \cong Coalg(\mathbb{C}_{kh}).

Proof. There is an obvious bijection between objects of GCF and objects of $\mathsf{Coalg}(\mathbb{C}_{\mathsf{kh}})$. Let (X, ν, A) and (X', ν', A') be two geometric conditional frames and let (\mathfrak{X}, γ) and (\mathfrak{X}', γ') be the corresponding \mathbb{C}_{kh} -coalgebras. Let $f: X \to X'$ be a function. We claim that f is a geometric conditional frame morphism if and only if it is a \mathbb{C}_{kh} -coalgebra morphism.

Suppose f is a geometric conditional frame morphism. Then f is continuous as $f^{-1}(a') \in A$ for all $a' \in A'$ and A' is precisely the set of opens in \mathfrak{X} . Let $x \in X$ and $a \in \Omega \mathfrak{X}$, then

$$(\mathbb{C}f)(\gamma(x))(a) = f[\gamma(x)(f^{-1}(a))] = f[\nu(x, f^{-1}(a))] = \nu'(f(x), a) = \gamma'(f(x))(a),$$

so f is a \mathbb{C}_{kh} -coalgebra morphism. The converse direction is similar.

5.35 Remark. The functor \mathbb{C}_{kh} coincides with the restriction to KHaus of the sober Kupke-Kurz-Pattinson lift (from section 4.1) $\ddot{\mathbb{C}}_{\{\lambda \Rightarrow, \lambda^{\Downarrow}\}}$ of \mathbb{C} with respect to the predicate liftings $\lambda^{\Rightarrow}, \lambda^{\Downarrow}$ (from example 2.22). This can be proved in a way similar to example 4.9 and using the duality from theorem 5.38 below.

We close this subsection with a brief look at the open predicate liftings that constitute geometric conditional logic.

5.36 Definition. Define the open predicate liftings $\lambda^{\Rightarrow}, \lambda^{\Downarrow}: \Omega^2 \to \Omega \circ \mathbb{C}_{\mathsf{top}}$ by

$$\lambda_{\mathfrak{X}}^{\rightrightarrows}(a, b) = \{h \in \mathbb{C}_{top} \mathfrak{X} \mid h(a) \subseteq b\} \quad \text{and} \quad \lambda_{\mathfrak{X}}^{\Downarrow}(a, b) = \{h \in \mathbb{C}_{top} \mathfrak{X} \mid h(a) \cap b \neq \emptyset\}.$$

Recall that we write Φ for a set of proposition letters. Then $\text{GML}(\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\})$ is given by

$$\varphi ::= \bot \mid p \mid \varphi_1 \land \varphi_2 \mid \bigvee_{i \in I} \varphi_i \mid \varphi_1 \Rightarrow \varphi_2 \mid \varphi_1 \Downarrow \varphi_2,$$

where $p \in \Phi$.

Let $\mathfrak{X} = (\mathfrak{X}, \gamma, V)$ be a \mathbb{C}_{top} -model, then

$$\mathfrak{X}, x \Vdash \varphi \Rightarrow \psi \quad \text{iff} \quad \gamma(x)(\llbracket \varphi \rrbracket^{\mathfrak{X}}) \subseteq \llbracket \psi \rrbracket^{\mathfrak{X}}$$

and

$$\mathfrak{X}, x \Vdash \varphi \Downarrow \psi \quad \text{iff} \quad \gamma(x)(\llbracket \varphi \rrbracket^{\mathfrak{X}}) \cap \llbracket \psi \rrbracket^{\mathfrak{X}} \neq \emptyset.$$

Moreover, the topology on $\mathbb{C}_{top} \mathfrak{X}$ is generated by $\lambda^{\Rightarrow}(a, b), \lambda^{\downarrow}(a, b)$ where $a, b \in \Omega \mathfrak{X}$, so $\{\lambda^{\Rightarrow}, \lambda^{\downarrow}\}$ is a characteristic set of open predicate liftings for \mathbb{C}_{top} .

 \triangleleft

5.2.2 An Isbell dual for \mathbb{C}_{kh}

We will now work towards a dual functor for \mathbb{C}_{kh} on the frame side, analogous to theorem 3.41. Recall from definition 3.7 that for an element a in a frame F the negation is defined as $\sim a = \bigvee \{b \in F \mid a \land b = \bot\}$.

5.37 Definition. Let *B* be a frame. Define AB to be the frame generated by the set $\{\Box(a, b), \diamondsuit(a, b) \mid a, b \in B\}$ subject to the following relations:

$$\begin{array}{ll}
\text{(A1)} & \Box(a, b) = \intercal_{\mathbb{A}B} \text{ if } b \lor \sim a = \intercal \\
\text{(A3)} & \Box(a, b) \land \Box(a, c) = \Box(a, b \land c) \\
\text{(A4)} & \diamondsuit(a, b) \lor \diamondsuit(a, c) = \diamondsuit(a, b \lor c) \\
\text{(A5)} & \Box(a, b) \land \Box(a, c) = \Box(a, b) \lor \diamondsuit(a, c) \\
\text{(A5)} & \Box(a, b) \land \Box(a, c) = \Box(a, b) \lor \diamondsuit(a, c) \\
\text{(A6)} & \Box(a, b) \land \diamondsuit(a, c) \leq \diamondsuit(a, b \land c) \\
\text{(A7)} & \Box(a, b) \land \Box(a, c) = \Box(a, b) \land \Box(b, c) \\
& \text{if } b \leq a \\
\text{(A9)} & \bigvee_{b \in A} \Box(a, b) = \Box(a, \bigvee A) \\
& \text{if } A \text{ is directed} \\
\text{(A2)} & \diamondsuit(a, b) = \bigsqcup(a, b) \land \Box(a, c) \\
\text{(A2)} & \diamondsuit(a, b) = \bot_{\mathbb{A}B} \text{ if } a \land b = \bot \\
\text{(A2)} & \diamondsuit(a, b) = \bot_{\mathbb{A}B} \text{ if } a \land b = \bot \\
\text{(A2)} & \diamondsuit(a, b) = \bot_{\mathbb{A}B} \text{ if } a \land b = \bot \\
\text{(A2)} & \diamondsuit(a, b) = \bot_{\mathbb{A}B} \text{ if } a \land b = \bot \\
\text{(A2)} & \diamondsuit(a, b) = \bot_{\mathbb{A}B} \text{ if } a \land b = \bot \\
\text{(A2)} & \diamondsuit(a, b) = (a, b \lor c) \\
\text{(A4)} & \diamondsuit(a, b) \lor \diamondsuit(a, c) = \diamondsuit(a, b \lor c) \\
\text{(A5)} & \Box(a, b) \land (a, c) = \Box(a, b) \land \diamondsuit(b, c) \\
\text{ if } b \leq a \\
\text{(A9)} & \bigvee_{b \in A} \Box(a, b) = \Box(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \diamondsuit(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \diamondsuit(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \diamondsuit(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \diamondsuit(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \diamondsuit(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \bigvee(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \bigvee(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \curlyvee(a, b) = \bigvee(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \lor(a, b) = \bigvee(a, \bigvee A) \\
\text{ if } A \text{ is directed} \\
\text{(A10)} & \bigvee_{b \in A} \lor(a, b) = \bigvee(a, \bigvee A) \\
\text{(A10)} & \bigvee(a, \boxtimes A) \lor(a, \boxtimes A) \\
\text{(A10)} & \bigvee(a, \boxtimes A) \lor(a, \boxtimes A) \lor(a, \boxtimes A) \\
\text{(A10)} & \bigvee(a, \boxtimes A) \lor(a, \boxtimes A) \lor(b, \boxtimes A) \lor(a, \boxtimes A) \lor(a$$

If $f: B \to B'$ is a frame homomorphism, define $\mathbb{A}f: \mathbb{A}B \to \mathbb{A}B'$ on generators by $\mathbb{A}f(\Box(a, b)) = \Box(f(a), f(b))$ and $\mathbb{A}f(\diamondsuit(a, b)) = \diamondsuit(f(a), f(b))$. (By remark 3.3 $\mathbb{A}_{st}f$ is well defined.) \triangleleft

We will now show that there is a duality between \mathbb{C}_{kh} and a restriction of A. The proof of the next theorem is somewhat similar to the proof of theorem 3.41 and to the proof of proposition III4.6 in [29]. The main difference with the proof in [29] is the fact that we use binary modalities instead of unary ones. Therefore our maps (that constitute the aforementioned duality) are defined differently. The second argument of each of the modalities is treated similarly to the proof in [29].

5.38 Theorem. If $\mathfrak{X} = (X, \tau)$ is a compact Hausdorff space then

$$\mathfrak{pt}(\mathbb{A}(\mathfrak{opn}\mathfrak{X})) \cong \mathbb{C}_{\mathsf{kh}}\mathfrak{X}.$$

Proof. Define a map

$$\varphi: \mathbb{C}_{\mathsf{kh}} \mathfrak{X} \to \mathfrak{pt}(\mathbb{A}(\mathfrak{opn} \mathfrak{X})): h \mapsto p_h,$$

where p_h is defined on generators by

$$p_h : \mathbb{A}(\mathfrak{opn} \mathfrak{X}) \to 2 : \begin{cases} \Box(a, b) \mapsto \top & \text{iff } h(a) \subseteq b \\ \diamondsuit(a, b) \mapsto \top & \text{iff } h(a) \cap b \neq \emptyset \end{cases}$$

Conversely, for a point $p \in \mathfrak{pt}(\mathbb{A}(\mathfrak{opn}\mathfrak{X}))$ define $h_p : \Omega\mathfrak{X} \to K\mathfrak{X}$ by

$$h_p(a) = \mathfrak{X} \setminus \bigcup \{ \mathfrak{b} \in \Omega \mathfrak{X} \mid p(\diamondsuit(a, \mathfrak{b})) = \bot \}.$$

This gives rise to a map $\psi : \mathfrak{pt}(\mathbb{A}(\mathfrak{opn}\mathfrak{X})) \to \mathbb{C}_{\mathsf{kh}}\mathfrak{X}$.

Note the absence of the box in the definition of h_p . This is less surprising that it may seem at first sight: Diamonds and boxes interact via relation (A5) and (A6). It follows from claim 5.38.C below that it is also possible to define $h_p(a)$ using boxes, namely via

$$h_p(a) = \bigcap \{ \beta \in \Omega \mathfrak{X} \mid p(\Box(a, \beta)) = \top \}.$$

We will show that both φ and ψ are well defined, that they are mutually inverse and that φ is continuous. Throughout, we write $X \coloneqq \mathbb{U} \mathfrak{X}$.

5.38.A Claim. For each $h \in \mathbb{C}_{\mathsf{kh}} \mathfrak{X}$, the map p_h indeed defines a point.

Proof of claim. Since a point is simply a frame homomorphism $\mathbb{A}_{kr}(\mathfrak{opn} \mathfrak{X}) \to 2$ and p_h is defined on generators, by remark 3.3 it suffices to show that the $p(\Box(a, b))$ and $p(\diamondsuit(a, b))$, where a, b range over $\Omega \mathfrak{X}$, satisfy relations (A1) through (A10).

- (A1) Observe that $\sim a = X \setminus \overline{a}$. Let β be such that $\beta \cup (X \setminus \overline{a}) = X$. Then $\overline{a} \subseteq \beta$. Since for all h in $\mathbb{C}_{\mathsf{kh}} \mathfrak{X}$ we have $h(a) \subseteq \overline{a}$, we know $p_h(\Box(a, \beta)) = \top$.
- (A2) Let $a, b \in \Omega \mathfrak{X}$ and suppose $a \cap b = \emptyset$. As a and b are open, we have $\overline{a} \cap b = \emptyset$. Since $h(a) \subseteq \overline{a}$ it follows that $h(a) \cap b = \emptyset$, so $p_h(\diamondsuit(a, b)) = \bot$.
- (A3) For all $a, b, c \in \Omega \mathfrak{X}$ we have $p_h(\Box(a, b \cap c)) = \top$ iff $h(a) \subseteq b \cap c$ iff $[h(a) \subseteq b$ and $h(a) \subseteq c$] iff $[p_h(\Box(a, b)) = \top$ and $p_h(\Box(a, c)) = \top$] iff $p_h(\Box(a, b) \land \Box(a, c)) = \top$.
- (A4) For all $a, b, c \in \Omega \mathfrak{X}$ we have $p_h(\diamondsuit(a, b) \lor \diamondsuit(a, c)) = \top$ iff $[p_h(\diamondsuit(a, b)) = \top$ or $p_h(\diamondsuit(a, c)) = \top$] iff $[h(a) \cap b \neq \emptyset$ or $h(a) \cap c \neq \emptyset$] iff $h(a) \cap (b \cup c) \neq \emptyset$ iff $p_h(\diamondsuit(a, b \cup c)) = \top$.
- (A5) Suppose $p_h(\Box(a, \delta \cup c)) = \top$. If $p_h(\Box(a, \delta)) = \top$ we are done, so suppose otherwise. Then $p_h(\Box(a, \delta)) = \bot$, so $h(a) \subseteq \delta \cup c$ and $h(a) \notin \delta$. Then $h(a) \cap c \neq \emptyset$ and hence $p_h(\diamondsuit(a, c)) = \top$.
- (A6) Suppose $p_h(\Box(a, b)) = \top$ and $p_h(\diamondsuit(a, c)) = \top$. Then $h(a) \cap c \neq \emptyset$, so there is some $x \in h(a)$ such that $x \in c$. Since $h(a) \subseteq b$ this x must be in b, and hence $h(a) \cap (b \cap c) \neq \emptyset$, so that $h_p(\diamondsuit(a, b \cap c)) = \top$.
- (A7) Suppose $b \subseteq a$. If $h(a) \subseteq b$ then by (C2) from definition 5.29 h(a) = h(b), hence $h \in \Box(a,c)$ iff $h \in \Box(b,c)$. So if $p_h(\Box(a,b)) = \top$ we have $p_h(\Box(a,c)) = p_h(\Box(b,c))$, which proves that (A7) holds.
- (A8) Suppose $b \subseteq a$. If $h(a) \subseteq b$ then by (C2) from definition 5.29 h(a) = h(b), hence $h \in \Diamond(a,c)$ iff $h \in \Diamond(b,c)$. So if $p_h(\Box(a,b)) = \top$ we have $p_h(\Diamond(a,c)) = p_h(\Diamond(b,c))$ for all c, which proves that (A8) holds.
- (A9) We need to show that $\bigvee_{\delta \in A} p_h(\Box(a, \delta)) = p_h(\Box(a, \bigcup A))$. Suppose $\bigvee_{\delta \in A} p_h(\Box(a, \delta)) = \top$, then $p_h(\Box(a, \delta)) = \top$ for some $\delta \in A$. So for this δ we have $h(a) \subseteq \delta$. Since $\delta \subseteq \bigcup A$ this implies $h(a) \subseteq \bigcup A$ and hence $p_h(a, \bigcup A) = \top$.

Conversely, suppose $p_h(\Box(a, \bigcup A)) = \top$, then $h(a) \subseteq \bigcup A$. The sets in A form an open cover of h(a) and since h(a) is closed it has a finite subcover. Since A is a directed set, there is some $b \in A$ containing the union of this finite subcover, and therefore $h(a) \subseteq b$. But then $p_h(\Box(a, b)) = \top$ and hence $\bigvee_{b \in A} p_h(\Box(a, b)) = \top$.

(A10) We need to show that $\bigvee_{\delta \in A} p_h(\diamondsuit(a, \delta)) = p_h(\diamondsuit(a, \bigcup A))$. Suppose $\bigvee_{\delta \in A} p_h(\diamondsuit(a, \delta)) = \top$, then there is some $\delta \in A$ such that $p_h(\diamondsuit(a, \delta)) = \top$, so $h(a) \cap \delta \neq \emptyset$. Since $\delta \subseteq \bigcup A$ this implies $h(a) \cap (\bigcup A) \neq \emptyset$, hence $p_h(\diamondsuit(a, \bigcup A)) = \top$.

Suppose $\bigvee_{b \in A} p_h(\diamondsuit(a, b)) = \bot$. Then $h(a) \cap b = \varnothing$ for all $b \in A$. Therefore every $b \in A$ is contained in the open set $X \smallsetminus h(a)$, hence $\bigcup A \subseteq X \smallsetminus h(a)$. This implies $h(a) \cap (\bigcup A) = \varnothing$, so $p_h(\diamondsuit(a, \bigcup A)) = \bot$.

So for each $h \in \mathbb{C}_{kh} \mathfrak{X}$, the map p_h is indeed a point.

 \diamond

5.38.B Claim. For each point $p \in pt(\mathbb{A}(\mathfrak{opn} \mathfrak{X}))$ the map h_p is an element of $\mathbb{C}_{kh}\mathfrak{X}$.

Proof of claim. It is clear that $h_p(a) \in K\mathfrak{X}$, because it is the complement of a union of open sets. We need to show that h_p satisfies conditions (C1) and (C2) from definition 5.29.

First (C1). Let $a \in \Omega \mathfrak{X}$. Using (A2) we find $p(\diamondsuit(a, \sim a)) = p(\bot) = \bot$ so $\bigcup \{c \mid p(\diamondsuit(a, c)) = \bot\} \supseteq \sim a$ and hence

$$h_p(a) = X \setminus \bigcup \{ c \mid p(\diamondsuit(a, c)) = \bot \} \subseteq X \setminus \neg a = \overline{a}.$$

Now (C2). Suppose $b \subseteq a$ and $h_p(a) \subseteq b$. We need to show that $h_p(a) = h_p(b)$. Abbreviate $C_a = \{c \in \Omega \mathfrak{X} \mid p(\Diamond(a,c)) = \bot\}$. Then $\bigcup C_a \in \Omega \mathfrak{X}$. Moreover, C_a is directed, because $p(\Diamond(a,c)) = p(\Diamond(a,c')) = \bot$ implies $p(\Diamond(a,c \cup c')) = p(\Diamond(a,c)) \lor p(\Diamond(a,c')) = \bot \lor \bot = \bot$. Now (A10) implies $p(a, \bigcirc C_a) = \bot$.

Since $h_p(\delta) \subseteq a$ we have $X = a \cup (X \setminus h_p(\delta)) = a \cup \bigcup C$. Because $p(a, X) = \top$ and $p(a, \bigcup C_a) = \bot$ it follows from (A5) that $p(\Box(a, \delta)) = \top$. Therefore, by (A8), $p(\diamondsuit(a, d)) = p(\diamondsuit(\delta, d))$ for all d so $h_p(a) = h_p(\delta)$. Clearly this implies $h_p(a) = h_p(\delta)$, as desired.

5.38.C Claim. For every point p we have

- (i) $h_p(a) \cap b \neq \emptyset$ iff $p(\diamondsuit(a, b)) = \top$; and
- (*ii*) $h_p(a) \subseteq \beta$ iff $p(\Box(a, \beta)) = \top$.

Proof of claim. Abbreviate $C_a = \{c \in \Omega \mathfrak{X} \mid p(\diamondsuit(a,c)) = \bot\}$. We have seen in the proof of claim 5.38.B that $p(\diamondsuit(a, \bigcup C_a)) = \bot$. It follows from (A4) that for any $d \subseteq \bigcup C_a$ we have $p(\diamondsuit(a,d)) = \bot$. By definition $p(\diamondsuit(a,d)) = \bot$ implies $d \subseteq C_a$. Therefore

$$h_p(a) \cap d \neq \emptyset$$
 iff $d \notin \bigcup C_a$ iff $p(\diamondsuit(a,d)) \neq \bot$ iff $p(\diamondsuit(a,b)) = \top$.

This proves (i).

Proving (ii) requires some more work. Suppose $h_p(a) \subseteq b$. Then $b \cup \bigcup C_a = X$ so $p(\Box(a, b \cup \bigcup C_a)) = \top$. By (A5) we have $p(\Box(a, b \cup \bigcup C_a)) \leq p(\Box(a, b)) \vee p(\diamondsuit(a, \bigcup C_a))$ and since $p(\diamondsuit(a, \bigcup C_a)) = \bot$, we must have $p(\Box(a, b)) = \top$.

For the converse, suppose $h_p(a) \notin \beta$. Then $X \setminus \bigcup C_a \notin \beta$, so

$$\delta \cup \bigcup C_a \neq X.
 \tag{5.6}$$

 \diamond

If $\delta' \leq \delta$ then by lemma 3.8 $\sim \delta' \vee \delta = \top$, so by (5.6) $\sim \delta' \notin \bigcup C$ and $h_p(a) \cap \sim \delta \neq \emptyset$; hence by part (i) of this proof $p(\diamondsuit(a, \sim \delta')) = \top$. Relation (A8) gives

$$p(\Box(a, b')) \land p(\diamondsuit(a, \neg b')) \le p(\diamondsuit(a, b' \land \neg b')) = p(\diamondsuit(a, \bot)) = \bot$$

and since $p(\diamondsuit(a, \sim b')) = \top$ this implies $p(\Box(a, b')) = \bot$. It then follows from (A10) and regularity of \mathfrak{X} that

$$p(\Box(a, b)) = p(\Box(a, \bigcup_{b' \leqslant b} b')) = p(\bigcup_{b' \leqslant b} \Box(a, b')) = \bigcup_{b' \leqslant b} p(\Box(a, b')) = \bot$$

We may conclude that $p(\Box(a, b)) = \top$ iff $h_p(a) \subseteq b$.

5.38.D Claim. The maps φ and ψ define a bijection between $\mathbb{C}_{kh} \mathfrak{X}$ and $\mathfrak{pt}(\mathbb{A}(\mathfrak{opn} \mathfrak{X}))$.

Proof of claim. We will show that for all $p \in \mathfrak{pt}(\mathbb{A}(\mathfrak{opn} \mathfrak{X}))$ we have $p = p_{h_p}$ and for all $h \in \mathbb{C}_{kh} \mathfrak{X}$ we have $h = h_{p_h}$. We start with showing $p = p_{h_p}$.

It suffices to show that p and p_{h_p} coincide on the generators of $\mathbb{A}(\mathfrak{opn}\mathfrak{X})$. It follows from the definition of p_h (for a point p) and claim 5.38.C that

$$p(\Box(a, b)) = \top$$
 iff $h_p(a) \subseteq b$ iff $p_{h_p}(\Box(a, b)) = \top$

and

$$p(\diamondsuit(a, b)) = \top$$
 iff $h_p(a) \cap b \neq \varnothing$ iff $p_{h_p}(\diamondsuit(a, b)) = \top$.

Therefore $p = p_{h_p}$.

Now let us prove $h = h_{p_h}$. It suffices to show that for each $a, b \in \Omega \mathfrak{X}$ we have $h(a) \subseteq b$ iff $h_{p_h}(a) \subseteq b$. By definition of p_h we have $h(a) \subseteq b$ iff $p_h(\Box(a, b)) = \top$. It follows from claim 5.38.C that $p_h(\Box(a, b)) = \top$ iff $h_{p_h}(a) \subseteq b$. This proves the claim. \diamondsuit

5.38.E Claim. The map φ is continuous.

Proof. The opens of $\mathfrak{pt}(\mathbb{A}(\mathfrak{opn}\mathfrak{X}))$ are generated by $\Box(a, b) = \{p \mid p(\Box(a, b)) = \top\}$ and $\Diamond(a, b) = \{p \mid p(\Diamond(a, b)) = \top\}$. We have

$$\varphi^{-1}(\widetilde{\Box(a, \delta)}) = \varphi^{-1}(\{p \mid p(\Box(a, \delta)) = \mathsf{T}\}) = \{h \in \mathbb{C}_{\mathsf{kh}} \mathfrak{X} \mid h(a) \subseteq \delta\} = \boxplus(a, \delta)$$

and similarly $\varphi^{-1}(\widehat{\diamond(a, b)}) = \hat{\diamond(a, b)}$. This proves continuity of φ .

This completes the proof of the theorem.

An immediate corollary of this theorem is the fact that the functor \mathbb{A} on Frm preserves compact regularity. Let \mathbb{A}_{kr} be the restriction of \mathbb{A} to KRFrm. Theorem 5.38 entails that the frames $\mathbb{A}_{kr}(\mathfrak{opn}\mathfrak{X})$ and $\mathfrak{opn}(\mathbb{C}_{kh}\mathfrak{X})$ are isomorphic for every $\mathfrak{X} \in \mathsf{KHaus}$.

5.39 Definition. For a compact Hausdorff space \mathfrak{X} , let $\eta_{\mathfrak{X}}$ be the map

$$\mathbb{A}_{\mathrm{kr}}(\mathfrak{opn}\,\mathfrak{X}) \xrightarrow{\mathfrak{opn}\,\mathfrak{opt}} \mathfrak{opn}(\mathfrak{pt}(\mathbb{A}_{\mathrm{kr}}(\mathfrak{opn}\,\mathfrak{X}))) \xrightarrow{\mathfrak{opn}\,\varphi} \mathfrak{opn}(\mathbb{C}_{\mathrm{kh}}\mathfrak{X}).$$

It is routine to verify that, on generators, this is given by $\Box(a, b) \mapsto \boxplus(a, b)$ and $\diamondsuit(a, b) \mapsto \diamondsuit(a, b)$.

It turns out that the the collection $(\eta_{\mathfrak{X}})_{\mathfrak{X}\in\mathsf{KHaus}}$ constitutes a natural isomorphism between $\mathbb{A}_{\mathsf{kr}} \circ \mathfrak{opn}$ and $\mathfrak{opn} \circ \mathbb{C}_{\mathsf{kh}}$.

5.40 Proposition. Let $\eta = (\eta \chi)_{\chi \in \mathsf{KHaus}}$. Then $\eta : \mathbb{A}_{\mathsf{kr}} \circ \mathfrak{opn} \to \mathfrak{opn} \circ \mathbb{C}_{\mathsf{kh}}$ is a natural isomorphism.

Proof. It follows from theorem 5.38 that $\eta_{\mathfrak{X}}$ is an isomorphism for each compact Hausdorff sapce \mathfrak{X} , so we only need to show naturality. Let $f : \mathfrak{X} \to \mathfrak{X}'$ be a continuous map between compact Hausdorff spaces. We need to show that the following diagram commutes

 \diamond

Let $\Box(a, b)$ be a generator in $\mathbb{A}_{kr}(\operatorname{Clop} \mathfrak{X}')$. Then

$$\begin{aligned} \mathfrak{opn}(\mathbb{C}_{\mathsf{kh}}f)(\eta_{\mathfrak{X}'}(\square(a, \delta))) &= (\mathbb{C}_{\mathsf{kh}}f)^{-1}(\boxplus(a, \delta)) & \text{Def of } \mathfrak{opn} \text{ and } \eta_{\mathfrak{X}'} \\ &= \{h \in \mathbb{C}_{\mathsf{kh}}\mathfrak{X} \mid (\mathbb{C}_{\mathsf{kh}}f)(h) \in \boxplus(a, \delta)\} \\ &= \{h \in \mathbb{C}_{\mathsf{kh}}\mathfrak{X} \mid (\mathbb{C}_{\mathsf{kh}}f)(h)(a) \subseteq \delta\} & \text{Definition of } \boxplus \\ &= \{h \in \mathbb{C}_{\mathsf{kh}}\mathfrak{X} \mid f[h(f^{-1}(a))] \subseteq \delta\} & \text{Definition of } \mathbb{C}_{\mathsf{kh}}f \\ &= \{h \in \mathbb{C}_{\mathsf{kh}}\mathfrak{X} \mid h(f^{-1}(a)) \subseteq f^{-1}(\delta)\} \\ &= \boxplus(f^{-1}(a), f^{-1}(\delta)) & \text{Definition of } \boxplus \\ &= \eta_{\mathfrak{X}}(\square(f^{-1}(a), f^{-1}(\delta))) & \text{Definition of } \eta_{\mathfrak{X}} \\ &= \eta_{\mathfrak{X}}(\mathbb{A}_{\mathsf{kr}}(\mathfrak{opn} f)(\square(a, \delta))) & \text{Def of } \mathbb{A}_{\mathsf{kr}} \text{ and } \mathfrak{opn} \end{aligned}$$

With a similar argument it can be shown that

$$(\mathfrak{opn}(\mathbb{C}_{\mathsf{kh}}f)(\eta_{\chi'}(\diamondsuit(a, \beta))) = \eta_{\chi}(\mathbb{A}_{\mathsf{kr}}(\mathfrak{opn} f)(\diamondsuit(a, \beta))).$$

This proves that η is a natural transformation.

Applying lemma A.6 to proposition 5.40 yields the following corollary.

5.41 Corollary. There is a dual equivalence

$$Alg(\mathbb{A}_{kr}) \equiv^{op} Coalg(\mathbb{C}_{kh}).$$

This section provided an initial attempt at the development of geometric conditional logic. We finish this section on the claim that there are still many interesting questions concerning geometric conditional logic, some of which can be found in chapter 6.

Conclusion

We have started building a framework for coalgebraic *geometric* logic and investigated some examples of concrete funtors. There are still many unanswered and interesting questions. We outline possible directions for further research.

Bisimulations In [6] the authors define Λ -bisimulations (which are inspired by [17]) between set coalgebas. In this thesis we define Λ -bisimulations between Stone coalgebras (section 2.3) and between sober coalgebras (section 3.4). This raises the question whether a more uniform treatment of Λ -bisimulations is possible, which encompasses all these cases.

Lifting functors In section 4.1 we give a method to lift a set functor \mathbb{T} together with a set of predicate liftings Λ for \mathbb{T} to an endofunctor $\ddot{\mathbb{T}}_{\Lambda}$ on **Sob**. We know three instances where this lifted functor preserves the compact Hausdorff property: when lifting the powerset functor, the monotone functor and the conditional functor (with respect to a suitable Λ). An interesting question is whether we can find general conditions (on \mathbb{T} or Λ) which imply preservation properties of $\ddot{\mathbb{T}}$, like preserving compactness or the Hausdorff property. This search could be inspired by the cases where we know that the compact Hausdorff property is preserved.

Besides, in [57] the authors give a method for lifting a certain class of set functors to endofunctors on Frm, which leads to geometric logic with an added modality ∇ . We suspect that there is a connection with our way of lifting set functors to sober functors, in the sense that, for a cleverly chosen set of predicate liftings Λ , the lift of \mathbb{T} to a functor on Frm is dual to $\ddot{\mathbb{T}}_{\Lambda}$.

Examples in theory As mentioned at the end of chapter 3, it is not completely clear how the concrete examples of functors we have seen fit in the general theory that we developed. An interesting direction for further research is to investigate how examples fit within this general theory.

Proof theory Throughout this thesis we have focussed on the semantics of certain coalgebraic logics. An interesting course of action is to investigate the syntactic side of the logics involved. One could either try to do this in full generality, or (as a first step) develop proof theory for examples such as monotone and conditional geometric logic.

Populaire samenvatting

We schetsen wat een set-coalgebra is en hoe zulke coalgebras verschillende structure beschrijven.

Coalgebras Laat \mathbb{T} een formule zijn die van elke verzameling X een (andere) verzameling $\mathbb{T}X$ maakt (dat noemen we een functor). Een voorbeeld van zo'n formule is de machtsverzameling functor \mathbb{P} , die stuurt een verzameling X naar de machtsverzameling $\mathbb{P}X = \{V \mid V \subseteq X\}$. Een \mathbb{T} -coalgebra is dan simpelweg een paar (X, γ) van een verzameling X en een afbeelding $\gamma : X \to \mathbb{P}X$. Coalgebras zijn een algemeen raamwerk om structuren te beschrijven. We geven twee voorbeelden van zulke structuren en hoe ze als coalgebra gezien kunnen worden.

Transitie systemen Een transitie systeem is een verzameling X met daarop een relatie R. Een relatie is een deelverzameling van $X \times X$. Als x en y twee punten in X zijn dan zeggen we "x ziet y" als $(x, y) \in R$.

Een voorbeeld van een transitie systeem is



Dit is een weergave van (X, R) met $X = \{x_1, x_2, x_3\}$ en

$$R = \{(x_1, x_2), (x_1, x_3), (x_2, x_1), (x_2, x_3)\}.$$

We kunnen dit zien als een \mathbb{P} -coalgebra door $\gamma_R : X \to \mathbb{P}X$ te definiëren als $\gamma(x_i) = \{x_j \in X \mid (x_i, x_j) \in R\}$. Dan beschrijft de \mathbb{P} -coalgebra (X, γ_R) precies het transitie system (X, R); dat wil zeggen, als we alleen (X, γ_R) zouden weten kunnen we daaruit aflezen wat (X, R) is en andersom.

In het algemeen corresponderen transitie systemen één-op-één met P-coalgebras.

Gelabelde transitie systemen Stel nu dat we aan elke pijl een label toevoegen. Zij A een collectie labels waaruit we kunnen kiezen. Een gelabeld transitie systeem een paar (X, L) met X een verzameling en $L \subseteq X \times A \times X$ een gelabelde relatie. Een voorbeeld van een gelabeld transitie systeem is



Dit is (X_1, L_1) met $X_1 = \{x_1, x_2, x_3\}$ en

$$L_1 = \{(x_1, a_2, x_2), (x_1, a_4, x_3), (x_2, a_1, x_1), (x_2, a_3, x_3)\}.$$

Om dit als een coalgebra te zien kunnen we niet meer \mathbb{P} gebruiken maar moeten we een andere functor, \mathbb{L} , maken.

Voor een verzameling X definiëren we

$$\mathbb{L}X = \mathbb{P}(A \times X).$$

We kunnen het gelabelde transitie systeem (X_1, L_1) nu beschouwen als L-coalgebra: Laat $\gamma_{L_1}(x_i) = \{(a, x_j) \mid (x_i, a, x_j) \in L\}$. Dan is

$$\gamma_{L_1}: X \to \mathbb{P}X: \begin{cases} x_1 \mapsto \{(a_2, x_2), (a_4, x_3)\} \\ x_2 \mapsto \{(a_1, x_1), (a_3, x_3)\} \\ x_3 \mapsto \emptyset \end{cases}$$

en dan kunnen we uit het paar (X, γ_{L_1}) het gelabelde transitie systeem (X, L) achterhalen. In het algemeen corresponderen gelabelde transitie systemen precies met Lcoalgebras.

Deze scriptie Coalgebras zoals hierboven, maar in een meer algemene vorm, vormen de basis van deze thesis. We definiëren logica op coalgebras en bestuderen eigenschappen van verschillende klassen coalgebras.

Bibliography

- [1] S. Abramsky. A cook's tour of the finitary non-well-founded sets, 1988. Invited Lecture at BCTCS.
- [2] S. Abramsky. Coalgebras, Chu spaces, and representations of physical systems. Journal of Philosophical Logic, 42(3):551–574, Jun 2013.
- [3] P. Aczel. *Non-well-founded Sets.* Center for the Study of Language and Information Publication Lecture Notes. Cambridge University Press, 1988.
- [4] E.W. Adams. The logic of conditionals. Springer Netherlands, 1975.
- [5] M.A. Armstrong. *Basic Topology*. Undergraduate Texts in Mathematics. Springer-Verlag New York, 2013.
- [6] Z. Bakhtiari and H.H. Hansen. Bisimulation for weakly expressive coalgebraic modal logics. In 7th Conference on Algebra and Coalgebra in Computer Science, CALCO 2017, June 12-16, 2017, Ljubljana, Slovenia, pages 4:1–4:16, 2017.
- [7] A. Baltag and G. Cinà. Bisimulation for conditional modalities. *Studia Logica*, 106(1):1–33, Feb 2018.
- [8] G. Bezhanishvili, N. Bezhanishvili, and J. Harding. Modal compact Hausdorff spaces. Journal of Logic and Computation, 25(1):1–35, 2015.
- [9] N. Bezhanishvili, G. Fontaine, and Y. Venema. Vietoris bisimulations. Journal of Logic and Computation, 20(5):1017–1040, 2010.
- [10] P. Blackburn, M. de Rijke, and Y. Venema. *Modal Logic*. Cambridge Tracts in Theoretical Computer Science. Cambridge University Press, 2001.
- [11] C. Boutilier. Conditional logics of normality as modal systems. In Proceedings of the 8th National Conference on Artificial Intelligence. Boston, Massachusetts, July 29 - August 3, 1990, 2 Volumes., pages 594–599, 1990.
- [12] S. Burris and H.P. Sankappanavar. A Course in Universal Algebra. Springer, 1981.
- [13] B.F. Chellas. Modal Logic: An Introduction. Cambridge University Press, 1980.
- [14] C. Cirstea, A. Kurz, D. Pattinson, L. Schröder, and Y. Venema. Modal logics are coalgebraic. In Samson Abramsky, Erol Gelenbe, and Vladimiro Sassone, editors, *Visions of Computer Science*, *BCS International Academic Research Conference* (*BCS 2008*), pages 129–140. British Computer Society, 2008.

- [15] L. Demey. Some remarks on the model theory of epistemic plausibility models. In Computing Research Repository - CORR, volume 21, 03 2010.
- [16] S. Enqvist and S. Sourabh. Bisimulations for coalgebras on Stone spaces. Journal of Logic and Computation, 2018.
- [17] D. Gorín and L. Schröder. Simulations and bisimulations for coalgebraic modal logics. CoRR, abs/1303.2467, 2013.
- [18] G. Grätzer. Universal Algebra. Springer-Verlag New York, 1979.
- [19] D.F. Gruber and J.S. Sparks. First observation of fluorescence in marine turtles. American Museum Novitates, 3845:1–8, 2015.
- [20] H.H. Hansen. Monotonic modal logics, 2003. Master's thesis, Institute for Logic, Language and Computation, University of Amsterdam.
- [21] H.H. Hansen and C. Kupke. A coalgebraic perspective on monotone modal logic. Electronic Notes in Theoretical Computer Science, 106:121–143, 2004.
- [22] H.H. Hansen, C. Kupke, and E. Pacuit. Neighbourhood structures: bisimilarity and basic model theory. *Logical Methods in Computer Science*, 5(2), April 2009.
- [23] I. Hasuo and B. Jacobs. Context-free languages via coalgebraic trace semantics. In J.L. Fiadero, N. Harman, M. Roggenbach, and J.J.M.M. Rutten, editors, Proceedings of the 1st Conference on Algebraic and Coalgebraic Methods in Computer Science (CALCO 2005), volume 3629 of Lecture Notes in Computer Science, pages 175–193. Springer, 2005.
- [24] B. Jacobs. Many-sorted coalgebraic modal logic: a model-theoretic study. Theoretical Informatics and Applications, 35(1), 2001.
- [25] B. Jacobs. The temporal logic of coalgebras via galois algebras. Mathematical Structure in Computer Science, 12:875–903, 2002.
- [26] B. Jacobs. Coalgebraic walks, in quantum and Turing computation. In M. Hofmann, editor, *Foundations of Software Science and Computational Structures*, pages 12–26, Berlin, Heidelberg, 2011. Springer Berlin Heidelberg.
- [27] B. Jacobs. Introduction to Coalgebra: Towards Mathematics of States and Observation. Cambridge Tracts in Theoretical Computer Science. Cambridge University Press, 2016.
- [28] B. Jacobs and J.J.M.M. Rutten. A tutorial on (co)algebras and (co)induction. EATCS Bulletin, 62:62–222, 1997.
- [29] P.T. Johnstone. Stone Spaces. Cambridge Studies in Advanced Mathematics. Cambridge University Press, 1982.
- [30] P.T. Johnstone. Vietoris locales and localic semilattices. In R.-E. Hoffmann and K.H. Hofmann, editors, *Continuous lattices and their applications*, Lecture Notes in Pure and Applied Mathematics, pages 155–180. Dekker, New York, 1985. Proceedings of a conference held at the University of Bremen, July 2-3, 1982.

- [31] P.T. Johnstone. Sketches of an Elephant: A Topos Theory Compendium. Number v. 2 in Oxford Logic Guides. Clarendon Press, 2002.
- [32] P.T. Johnstone and S.J. Vickers. Preframe representations present. In A. Carboni, M.C. Pedicchio, and G. Rosolini, editors, *Category Theory: Proceedings of* the International Conference held in Como, Italy, July 22-28, 1990, Lecture Notes in Mathematics, pages 193–212. Springer Berlin Heidelberg, 2006.
- [33] J.G. Kemény and J.L. Snell. *Finite Markov chains*. University series in undergraduate mathematics. Van Nostrand, 1960.
- [34] S. Kripke. Semantical considerations on modal logic. Acta Philosophica Fennica, 16:83–94, 1963.
- [35] C. Kupke, A. Kurz, and D. Pattinson. Algebraic semantics for coalgebraic logics. *Electronic Notes in Theoretical Computer Science*, 106:219 – 241, 2004. Proceedings of the Workshop on Coalgebraic Methods in Computer Science (CMCS).
- [36] C. Kupke, A. Kurz, and D. Pattinson. Ultrafilter extensions for coalgebras. In J. Luiz Fiadeiro, N. Harman, M. Roggenbach, and J. Rutten, editors, *Algebra and Coalgebra in Computer Science*, pages 263–277, Berlin, Heidelberg, 2005. Springer Berlin Heidelberg.
- [37] C. Kupke, A. Kurz, and Y. Venema. Stone coalgebras. Theoretical Computer Science, 327(1):109–134, 2004. Selected Papers of CMCS '03.
- [38] C. Kupke and D. Pattinson. Coalgebraic semantics of modal logics: An overview. *Theoretical Computer Science*, 412(38):5070 – 5094, 2011. CMCS Tenth Anniversary Meeting.
- [39] A. Kurz, A. Palmigiano, and Y. Venema. Coalgebra and logic: A brief overview. Journal of Logic and Computation, 20(5):985–990, 2010.
- [40] S.M. Lane. Categories for the Working Mathematician. Graduate Texts in Mathematics. Springer New York, 1998.
- [41] J.J. Lee. Exclusive video: First "glowing" sea turtle found. https://news.nationalgeographic.com/2015/09/150928-sea-turtleshawksbill-glowing-biofluorescence-coral-reef-ocean-animals-science/, September 2015. Accessed: June 12, 2018.
- [42] D. Lewis. *Counterfactuals*. Harvard University Press, 1973.
- [43] L.S. Moss. Coalgebraic logic. Annals of Pure and Applied Logic, 96(1):277–317, 1999.
- [44] L.S. Moss and I.D. Viglizzo. Harsanyi type spaces and final coalgebras constructed from satisfied theories. *Electronic Notes in Theoretical Computer Science*, 106:279– 295, 2004. Proceedings of the Workshop on Coalgebraic Methods in Computer Science (CMCS).
- [45] J.R. Munkres. Topology. Featured Titles for Topology Series. Prentice Hall, Incorporated, 2000.

- [46] J.R. Norris. Markov chains. Cambridge series in statistical and probabilistic mathematics. Cambridge University Press, 1998.
- [47] D. Pattinson. Coalgebraic modal logic: soundness, completeness and decidability of local consequence. *Theoretical Computer Science*, 309(1):177–193, 2003.
- [48] J.J.M.M. Rutten. Automata and coinduction (an exercise in coalgebra). In Proceedings of the 9th International Conference on Concurrency Theory, CONCUR '98, pages 194–218, London, UK, UK, 1998. Springer-Verlag.
- [49] J.J.M.M. Rutten. Universal coalgebra: a theory of systems. Theoretical Computer Science, 249((1)):3–80, 2000.
- [50] L. Schröder. Expressivity of coalgebraic modal logic: The limits and beyond. In V. Sassone, editor, *Foundations of Software Science and Computational Structures*, pages 440–454, Berlin, Heidelberg, 2005. Springer Berlin Heidelberg.
- [51] L. Schröder, D. Pattinson, and D. Hausmann. Optimal tableaux for conditional logics with cautious monotonicity. In *Frontiers in Artificial Intelligence and Applications*, volume 215, pages 707–712, 01 2010.
- [52] A. Sokolova. Probabilistic systems coalgebraically: A survey. Theoretical Computer Science, 412(38):5095–5110, 2011. CMCS Tenth Anniversary Meeting.
- [53] R.C. Stalnaker. A theory of conditionals. Americal Philosophical Quarterly, pages 98–112, 1968.
- [54] F. Veltman. Logic for conditionals. PhD thesis, University of Amsterdam, 1985.
- [55] Y. Venema. Algebras and coalgebras. In P. Blackburn, J. Van Benthem, and F. Wolter, editors, *Handbook of Modal Logic*, volume 3 of *Studies in Logic and Practical Reasoning*, pages 331–426. Elsevier, 2007.
- [56] Y. Venema. Coalgebra and modal logic, 2017. Course notes.
- [57] Y. Venema, S.J. Vickers, and J. Vosmaer. Generalised powerlocales via relation lifting. Mathematical Structures in Computer Science, 23(1):142–199, 2013.
- [58] Y. Venema and J. Vosmaer. Modal logic and the Vietoris functor. In G. Bezhanishvili, editor, *Leo Esakia on Duality in Modal and Intuitionistic Logics*, pages 119–153, Dordrecht, 2014. Springer Netherlands.
- [59] S.J. Vickers. Topology via Logic. Cambridge University Press, New York, NY, USA, 1989.
- [60] S.J. Vickers. Geometric theories and databases. In Applications of Categories in Computer Science, volume 177, pages 288–314. Cambridge University Press, 1992.
- [61] S.J. Vickers. Geometric logic in computer science. In G. Burn, S. Gay, and M. Ryan, editors, *Theory and Formal Methods 1993*, pages 37–54, London, 1993. Springer London.
- [62] S.J. Vickers. Locales and toposes as spaces. In M. Aiello, I.E. Pratt-Hartmann, and J.F.A.K. van Benthem, editors, *Handbook of Spatial Logics*, pages 429–496, Secaucus, NJ, USA, 2007. Springer-Verlag New York, Inc.

- [63] S.J. Vickers. The connected vietoris powerlocale. *Topology and its Applications*, 156(11):1886–1910, 2009. A Conference in honour of Peter Collins and Mike Reed.
- [64] L. Vietoris. Bereiche zweiter ordnung. Monatshefte für Mathematik und Physik, 32(1):258–280, Dec 1922.
- [65] B. Trancón y Widemann and M. Hauhs. Distributive-law semantics for cellular automata and agent-based models. In A. Corradini, B. Klin, and C. Cïrstea, editors, *Conference on Algebra and Coalgebra in Computer Science (CALCO 2011)*, pages 344–358, Berlin, 2011. Springer Berlin.

Index

 Λ -bisimulation, 15 Kupke-Kurz-Pattinson lift \mathbb{T} -model, 17 Set to Sob, 49 Set to Stone, 52 Aczel-Mendler bisimulation, 13, 14 labelled transition system, 10 base category, 9 language, 14, 17 behavioural equivalence, 13 language:geometric logic, 29 bisimilarity, see bisimulation lift via pro-completion, 55 bisimulation Λ -bisimulation, 15, 42 modal equivalence, 18, 29 Aczel-Mendler bisimulation, 13, 14, 17 model, 14 coalgebra bisimulation, 13 over Stone-coalgebras, 17 descriptive conditional, 68 model morphism, 14, 17 structural, 72 monotone frame, 11 bounded morphism monotone modal logic, 15 between conditional frames, see condimonotone neighbourhood frame, 11 tional frame morphism normal modal logic, 15 between Kripke frames, 11 between monotone neighbourhood frames, predicate lifting, 14 11 clopen, 17 open, 29 coalgebra, 9 topological, 17 coalgebra bisimulation, 13 presentation, 25 coalgebra morphism, 9 conditional frame, 12 selection function frame, 13 conditional frame morphism, 12 similarity type, 14 conditional logic, 16 state space, 9 structural bisimulation, 72 descriptive conditional bisimulation, 68 structure map, 9 descriptive conditional frame, 61 system, 9 descriptive conditional model, 61 descriptive frame, 21 topological conditional frame, 72 descriptive monotone frame, 22 topological predicate lifting, 17 transition map, 9 final object, 13 transition system, 10 general conditional frame, 61 weighed transition system, 11 general conditional model, 61 general monotone frame, 22 geometric conditional frame, 73 Kripke frame, 11

A

Appendix

A.1 NOTATION

Throughout this thesis, Φ denotes a denumerable set of proposition letters. Also we have number of notational conventions. Sets come in many forms and, when the setting is not yet set, are written as capital letters X, X', Y. Elements of a set are denoted by lower case letters x, x', y, u. Subsets of a given set X are written in lower case calligraphy, a, b. Collections of subsets of a given set are written as W, W', V. Topological spaces are indicated by bold upper case calligraphic letters, $\mathbf{X}, \mathbf{X}', \mathbf{Y}$ and their underlying sets by the upper case normal font of the letter.

Categories are denoted in this font. Usually C stands for an arbitrary category. We fix some categories:

- Set is the category of sets and functions;
- Top is the category of topological spaces and continuous functions;
- Top₀, Sob, KTop, KSob, KHaus and Stone are the full subcategories of Top with as objects *T*₀-spaces, sober spaces, compact spaces, compact sober spaces, compact Hausdorff spaces and Stone spaces respectively;
- BA is the category of Boolean algebras and Boolean algebra homomorphisms;
- Frm is category of frames and frame homomorphisms;
- SFrm, KFrm, KSFrm and KRFrm are the full subcategories of Frm with objects spatial frames, compact frames, compact spatial frames and compact regular frames respectively.

More categories are defined as we go.

Functors are usually denoted by a blackboard font letter, $\mathbb{A}, \mathbb{B}, \mathbb{C}$. An arbitrary functor is written as \mathbb{T} and its the domain and range should be clear from the context. The following functors will be used throughout the thesis:

- U: Top → Set is the forgetful functor sending a topological space to its underlying set;
- \mathbb{Z} : Frm \rightarrow Set is the forgetful functor sending a frame to its carrier in Set, in particular this restricts to a functor BA \rightarrow Set;
- \mathbb{P} : Set \rightarrow Set is the covariant powerset functor;

• \mathbb{Q} : Set \rightarrow BA is the contravariant functor sending a set to its powerset Boolean algebra and a function on Set to its inverse image.

Some functors do not adhere to the rule of being a blackboard letter:

- $\mathfrak{uf} : \mathsf{BA} \to \mathsf{Stone}$ is the contravariant functor which sends a Boolean algebra B to the Stone space of ultrafilters topologised by $\{\widetilde{a} \mid a \in B\}$, where $\widetilde{a} = \{u \in \mathfrak{uf} B \mid a \in u\}$, and a homomorphism $f : B \to B'$ to $\mathfrak{uf} f : \mathfrak{uf} B' \to \mathfrak{uf} B$ defined by $(\mathfrak{uf} f)(u') = \{f^{-1}(a') \mid a' \in u'\};$
- $clp : Stone \rightarrow BA$ is the contravariant functor which sends a Stone space to its Boolean algebra of clopen sets and a continuous function f to f^{-1} ;
- pt : Frm → Top is the contravariant functor which sends a frame to its space of points, this is defined in definition 3.13;
- $\mathfrak{opn}: \mathsf{Top} \to \mathsf{Frm}$ is the contravariant functor which sends a topological space to its frame of open sets and a continuous map f to f^{-1} , see definition 3.6.

The functors $\mathfrak{u}\mathfrak{f}$ and $\mathfrak{c}\mathfrak{l}\mathfrak{p}$ constitute Stone duality; the functors $\mathfrak{p}\mathfrak{t}$ and \mathfrak{opn} constitute a duality between sober spaces and spatial frames. Furthermore we have the abbreviations

- $\mathbb{P} = \mathbb{Z} \circ \mathbb{Q}$: Set \rightarrow Set, known as the contravariant powerset functor;
- $\mathbb{C}lp = \mathbb{Z} \circ \mathfrak{clp}$: Stone \rightarrow Set sends a Stone space to its set of clopen subsets;
- $\Omega = \mathbb{Z} \circ \mathfrak{opn} : \mathsf{Top} \to \mathsf{Set}$ sends a topological space to its set of open subsets.

Along the way, more functors will be defined when required.

We often use diamonds and boxes. When we use these to as formal symbols who generate a frame we use an "empty" box and diamond, \Box and \diamondsuit . When using boxes and diamonds to define open or clopen sets in some topological space, we use "crossed" boxes and diamonds, \boxplus and \diamondsuit .

Sometimes we have split up long proofs into several claims. The claims are then numbered with letters within the proposition or theorem. The proof of such a claim is closed by a diamond, \diamond , instead of a box, in order to differentiate between to proof and it subproofs.

A.2 The monotone functor on Frm

Recall that the monotone functor $\mathbb{M} : \mathsf{Frm} \to \mathsf{Frm}$ is defined as follows: For a frame F the frame $\mathbb{M}F$ is the frame generated by the set $\mathbb{M}F = \{\Box a, \Diamond a \ (a \in F)\}$ subject to the relations

$(M1) \ \Box(a \land b) \le \Box a$	$(M2) \diamond a \leq \diamond (a \lor b)$
(M3) $\Box a \land \diamondsuit b \leq 0$ whenever $a \land b \leq 0$	(M4) $\Box a \lor \diamondsuit b \ge 1$ whenever $a \lor b \ge 1$
$(M5) \Box \lor A = \lor \{ \Box a \mid a \in A \}$	$(M6) \diamondsuit \bigvee A = \bigvee \{\diamondsuit a \mid a \in A\},\$

where $a, b \in F$ and A is a directed subset of F. For a homomorphism $f: F \to F'$ define $\mathbb{M}f: \mathbb{M}F \to \mathbb{M}F'$ on generators by

$$\Box a \mapsto \Box f(a), \quad \diamondsuit a \mapsto \diamondsuit f(a)$$

We have seen that \mathbb{M} preserves regularity in proposition 3.40. Via the duality from theorem 3.41 it follows from lemma 3.38 that \mathbb{M} preserves compactness as well. But we can also prove this fact without using the duality to move the problem to the topological side. Before we prove this, we give an equivalent definition of $\mathbb{M}F$, which is inspired by theorem 4.2 from [57]. We then show that this equivalent frame is compact whenever Fis. The proof is similar to the proof of theorem 4.2 from [57].

A.1 Definition. Let D be a frame. Define

$$\begin{split} \mathbb{M}'D \coloneqq \operatorname{Fr}\langle \wp_{\omega}D \times \wp_{\omega}D \quad (\operatorname{qua} \lor \operatorname{semilattice}) \mid \\ & (i') \quad (\gamma \cup \{a \land b\}, \delta) \leq (\gamma \cup \{a\}, \delta) \\ & (ii') \quad (\gamma, \delta \cup \{a\}) \leq (\gamma, \delta \cup \{a \lor b\}) \\ & (iii') \quad (\gamma \cup \{a\}, \delta) \land (\gamma, \delta \cup \{b\}) \leq (\gamma, \delta) \quad \text{if } a \land b = 0 \\ & (iv') \quad \forall \leq (\gamma \cup \{a\}, \delta \cup \{b\}) \quad \text{if } a \lor b = 1 \\ & (v') \quad (\gamma \cup \{\bigvee A\}, \delta) \leq \bigvee_{a \in A} (\gamma \cup \{a\}, \delta) \\ & (vi') \quad (\gamma, \{\bigvee A\} \cup \delta) \leq \bigvee_{a \in A} (\gamma, \{a\} \cup \delta) \\ & \rangle. \end{split}$$

Here we use \top to denote the top element of $\mathbb{M}'D$ and 1 for the top element of D. The join structure is given by $(\gamma, \delta) \lor (\gamma', \delta') = (\gamma \cup \gamma', \delta \cup \delta')$.

A.2 Lemma. Let D be a frame. Then $MD \cong M'D$.

Proof. Define

$$\mu: \mathbb{M}D \to \widehat{\mathbb{M}}D: \left\{ \begin{array}{l} \Box a \mapsto (\{a\}, \emptyset) \\ \diamondsuit a \mapsto (\emptyset, \{a\}) \end{array} \right.$$

and

$$\eta:\widehat{\mathbb{M}}D\to\mathbb{M}D:(\gamma,\delta)\mapsto\bigvee_{c\in\gamma}\Box c\vee\bigvee_{d\in\delta}\diamondsuit d.$$

These maps obviously give a bijection. We will show that both assignments preserve the relations from definition 3.39 and A.1, wherefore they can be lifted to frame homomorphisms. Moreover, both maps are frame homomorphisms.

The map μ is a homomorphism because it is defined on the generators and is welldefined for it preserves the relations from definition 3.39.

We first show that the assignment μ preserves (i) – (vi).

(i) First (i),

$$\mu(\Box(a \land b)) = (\{a \land b\}, \varnothing) \stackrel{(i')}{\leq} (\{a\}, \varnothing) = \mu(\Box a).$$

(ii) Second,

$$\mu(\diamondsuit a) = (\emptyset, \{a\}) \stackrel{(ii')}{\leq} (\emptyset, \{a \lor b\}) = \mu(\diamondsuit(a \lor b)).$$

(iii) For (iii), suppose $a \wedge b = 0$, then

$$\mu(\Box a \land \Diamond b) = (\{a\}, \emptyset) \land (\emptyset, \{b\}) \stackrel{(iii')}{\leq} (\emptyset, \emptyset) = \mu(0).$$

(iv) Now suppose $a \lor b = 1$, then

$$\mu(1) = \top \stackrel{(iv')}{\leq} = (\{a\}, \emptyset) \lor (\emptyset, \{b\}) = \mu(\Box a) \lor \mu(\diamondsuit b) = \mu(\Box a \lor \diamondsuit b).$$

(v) For (v), first observe

$$\mu(\Box \bigvee A) = (\{\bigvee A\}, \varnothing) \stackrel{(v')}{\leq} \bigvee_{a \in A} (\{a\}, \varnothing) = \bigvee_{a \in A} \mu(\Box a) = \mu(\bigvee_{a \in A} \Box a).$$

From (i') it follows that

$$(\{a\}, \emptyset) = (\{(\bigvee A) \land a\}, \emptyset) \le (\{\bigvee A\}, \emptyset)$$

so that $\mu(\Box a) \leq \mu(\Box \lor A)$ and hence $\mu(\bigvee_{a \in A} \Box a) = \bigvee_{a \in A} \mu(\Box a) \leq \mu(\Box \lor A)$.

(vi) From (vi') we get $\mu(\diamondsuit \lor A) \le \mu(\lor_{a \in A} \diamondsuit a)$. From (ii') it follows that $(\varnothing, \{a\}) \le (\varnothing, \lor A)$ so that $\mu(\diamondsuit a) \le \mu(\diamondsuit \lor A)$ for all $a \in A$ and hence $\mu(\lor_{a \in A} \diamondsuit a) = \bigvee_{a \in A} \mu(\diamondsuit a) \le \mu(\diamondsuit \lor A)$. This proves equality.

In order to show that η defines a frame homomorphism we need to prove that it preserves joins and relation (o') – (vi'). The preservation of joins is obvious. For (i'),

$$\begin{split} \eta(\gamma \cup \{a \land b\}, \delta) &= \bigvee_{c \in \gamma} \Box c \lor \Box (a \land b) \lor \bigvee_{d \in \delta} \diamondsuit d \\ &\leq \bigvee_{c \in \gamma} \Box c \lor \Box a \lor \bigvee_{d \in \delta} \diamondsuit d \\ &= \eta(\gamma \cup \{a\}, \delta). \end{split}$$

In a similar way (ii') can be treated.

For (iii'), suppose $a \wedge b = 0$, then

$$\eta(\gamma \cup \{a\}, \delta) \land \eta(\gamma, \delta \cup \{b\}) = \left(\bigvee \Box c \lor \Box a \bigvee \Diamond d\right) \land \left(\bigvee \Box c \lor \bigvee \Diamond d \lor \Diamond b\right)$$
$$= \bigvee \Box c \lor \bigvee \Diamond d \lor (\Box a \land \Diamond b)$$
$$\leq \bigvee \Box c \lor \bigvee \Diamond d \lor 0$$
$$= \eta(\gamma, \delta).$$

The dual notion (iv'), can be treated similarly.

For (v'),

$$\eta(\gamma \cup \{\bigvee A\}, \delta) = \bigvee_{c \in \gamma} \Box c \lor \Box \bigvee A \lor \bigvee_{d \in \delta} \diamondsuit d$$
$$= \bigvee_{c \in \gamma} \Box c \lor \bigvee_{a \in A} \Box a \lor \bigvee_{d \in \delta} \diamondsuit d$$
$$= \bigvee_{a \in A} (\bigvee \Box c \lor \Box a \lor \bigvee \diamondsuit d)$$
$$= \bigvee_{a \in A} (\gamma \cup \{a\}, \delta).$$

Lastly (vi') is similar.

A.3 Theorem. Suppose D is compact. Then MD is compact.

Proof. The frame $\mathbb{M}D$ is compact iff there is a preframe homomorphism $\varphi : \mathbb{M}D \to 2$ that is right adjoint to the unique frame homomorphism $!: 2 \to \mathbb{M}D$ (cf. [57, Theorem 4.2]).

By proposition A.2 $\mathbb{M}D \cong \mathbb{M}'D$, so we may work with the latter. Because all the relations in definition A.1 are join stable, we can use the preframe coverage theorem (theorem 5.1 in [32]) to find

 $\mathbb{M}'D \cong \operatorname{PreFr}(\wp_{\omega}D \times \wp_{\omega}D \text{ (qua poset)} | \text{ same relations as definition A.1 }).$

Define $\varphi : \mathbb{M}'D \to 2$ by

$$\varphi(\gamma, \delta) = \begin{cases} 1 & \text{iff there are } c \in \gamma \text{ such that } c \lor (\lor \delta) = 1 \\ 0 & \text{otherwise} \end{cases}$$

We need to check that φ is indeed a pre-frame homomorphism. Since φ is defined on generators, it suffices to show that it preservers the relations (i') – (vi'), because if it does it can be lifted in a unique way to a frame homomorphism $\mathbb{M}'D \to 2$. We check that φ preserves the relations one by one.

- (i') Suppose $\varphi(\gamma \cup \{a\}, \delta) = 0$, then $c \lor \lor \delta = 0$ for all $c \in \gamma$ and $(a \land b) \lor \lor \delta \le a \lor \lor \delta = 0$.
- (ii') If $\varphi(\gamma, \delta \cup \{a\}) = 1$, then $c \lor \lor (\delta \cup \{a\}) = 1$ for some $c \in \gamma$, so $c \lor \lor (\delta \cup \{a \lor b\}) \ge c \lor \lor (\delta \cup \{a\}) = 1$.
- (iii') Suppose $\varphi(\gamma \cup \{a\}, \delta) = 1$ and $\varphi(\gamma, \delta \cup \{b\}) = 1$. Then either there is some $c \in \gamma$ such that $c \vee \bigvee \delta = 1$, which implies $\varphi(\gamma, \delta) = 1$, or $a \vee \bigvee \delta = 1$. In the latter case, note that we also have some $c' \in \gamma$ such that $c' \vee \bigvee \delta \vee b = 1$, so that

$$c' \lor \bigvee \delta = \bigcup \lor \bigvee \delta \lor (a \land b) = (a \lor \bigvee \delta \lor c') \land (c' \lor \bigvee \delta \lor b) = 1 \land 1 = 1.$$

The first equality holds because $a \wedge b = 0$. Again we find $\varphi(\gamma, \delta) = 1$.

- (iv') If $a \lor b = 1$, then $a \lor \lor (\delta \cup \{b\}) = 1$ so $\varphi(\gamma \cup \{a\}, \delta \cup \{b\}) = 1$.
- (v') Suppose $\varphi(\gamma \cup \{\bigvee A\}, \delta) = 1$, then either $c \vee (\bigvee \delta) = 1$ for some $c \in \gamma$, or $1 = (\bigvee A) \vee (\bigvee \delta) = \bigvee_{a \in A} (a \vee (\bigvee \delta))$ (note that the latter is indeed a directed set, because A is). By compactness of D this gives $a \vee (\bigvee \delta) = 1$ for some $a \in A$. So both cases yield $\varphi(\bigvee_{a \in A} (\gamma \cup \{a\}, \delta)) = 1$.
- (vi') Suppose $\varphi(\gamma, \{\sqrt[n]{A}\} \cup \delta) = 1$, then, for some $c \in \gamma$, we have

$$1 = c \lor \bigvee (\{ \lor A\} \cup \delta) = \lor (c \lor a \lor \bigvee \delta)$$

and by compactness we must have $c \lor \bigvee(\{a\} \cup \delta) = 1$ for one of the *a*. (The set $\{c \lor a \lor \bigvee \delta \mid a \in A\}$ is directed and by (ii').)

Lastly, we need to check that φ is right-adjoint to $!: 2 \to \mathbb{M}'L$ (defined by $1 \mapsto \mathsf{T} = (1,1)$ (note that (1,1) is in the equivalence class of T), and $0 \mapsto (\emptyset, \emptyset)$). It suffices to show that $\varphi(!(p)) \ge p$ and $!(\varphi(\gamma, \delta)) \le (\gamma, \delta)$. For the first, suppose p = 1, then !(p) is the equivalence class of (1,1) and $\varphi(!(p)) = 1$. For the second, if $\varphi(\gamma, \delta) = 1$, then there are $c \in \gamma$ such that $c \lor (\lor \delta) = 1$ (in particular $\delta \neq \emptyset$) and hence

$$1 = (\{1\}, \delta) = (\{c \lor (\bigvee \delta)\}, \delta) \le (\{c\}, \delta) \le (\gamma, \delta).$$

The first inequality follows from recalling that δ is a finite set and applying (vi') repeatedly.

Proposition 3.40 and theorem A.3 combined yield the following result.

A.4 Corollary. The functor \mathbb{M} on Frm preserves compactness.

A.3 FOR REFERENCE

A.5 Lemma. Let C and D be categories and $\mathbb{I} : C \to D$, $\mathbb{J} : D \to C$ two functors that constitute a dual equivalence between C and D. Let $\mathbb{S} : C \to C$ and $\mathbb{T} : D \to D$ be endofunctors. The following are equivalent

- (i) there is a natural isomorphism $\eta : \mathbb{S} \circ \mathbb{J} \to \mathbb{J} \circ \mathbb{T}$;
- (ii) there is a natural isomorphism $\mu : \mathbb{T} \circ \mathbb{I} \to \mathbb{I} \circ \mathbb{S}$.

Proof. (i) to (ii): define μ by

$$\mathbb{TI} \xrightarrow{}{}^{\mathbb{I}_{\mathbb{TI}}} \mathbb{IJTI} \xrightarrow{}{}^{\mathbb{I}\eta_{\mathbb{I}}} \mathbb{ISJI} \xrightarrow{}{}^{\mathbb{IS}\epsilon} \mathbb{IS}.$$

These are all natural isomorphisms. (ii) to (i) is similar.

A.6 Lemma. Let C and D be categories and $\mathbb{I} : C \to D$, $\mathbb{J} : D \to C$ two contravariant functors that constitute a dual equivalence between C and D. Let $\epsilon : \mathrm{Id} \to \mathbb{J} \circ \mathbb{I}$ and $1 : \mathrm{Id} \to \mathbb{I} \circ \mathbb{J}$ be the corresponding natural isomorphisms. Let $\mathbb{S} : C \to C$ and $\mathbb{T} : D \to D$ be endofunctors. Suppose there is is a natural isomorphism $\eta : \mathbb{S} \circ \mathbb{J} \to \mathbb{J} \circ \mathbb{T}$. Then there is a dual equivalence

$$\mathsf{Alg}(\mathbb{S}) \equiv^{\operatorname{op}} \mathsf{Coalg}(\mathbb{T}).$$

Proof. We define two functors $\mathbb{X} : \mathsf{Alg}(\mathbb{S}) \to \mathsf{Coalg}(\mathbb{T})$ and $\mathbb{Y} : \mathsf{Coalg}(\mathbb{T}) \to \mathsf{Alg}(\mathbb{S})$ and we will give two natural isomorphisms $\mathrm{Id}_{\mathsf{Alg}(\mathbb{S})} \cong \mathbb{Y} \circ \mathbb{X}$ and $\mathrm{Id}_{\mathsf{Coalg}(\mathbb{T})} \cong \mathbb{X} \circ \mathbb{Y}$.

Definition of X. By lemma A.5 we get a natural isomorphism $\mu : \mathbb{T} \circ \mathbb{I} \to \mathbb{I} \circ \mathbb{S}$.

Let $\delta: \mathbb{S}X \to X$ be an S-algebra. Applying \mathbb{I} gives $\mathbb{I}\delta: \mathbb{I}X \to \mathbb{I}\mathbb{S}X$ and composition with μ_X^{-1} gives

$$\mathbb{I}X \xrightarrow{\mathbb{I}\delta} \mathbb{I}\mathbb{S}X \xrightarrow{\mu_X^{-1}} \mathbb{T}\mathbb{I}X.$$
(A.1)

So if we put $\gamma_{\delta} \coloneqq \mu_X^{-1} \circ \mathbb{I}\delta$ the pair $(\mathbb{I}X, \gamma_{\delta})$ is a \mathbb{T} -coalgebra. Define $\mathbb{X}(X, \delta) \coloneqq (\mathbb{I}X, \gamma_{\delta})$. Let $f : X \to X'$ be a \mathbb{S} -algebra morphism from (X, δ) to (X', δ') . The the following diagram commutes,

Moreover, the following diagram commutes

$$\begin{array}{c|c} \mathbb{I}X & \stackrel{\mathbb{I}f}{\longleftarrow} \mathbb{I}X' \\ \mathbb{I}\delta & & \downarrow \mathbb{I}\delta' \\ \mathbb{I}SX & \stackrel{\mathbb{I}Sf}{\longleftarrow} \mathbb{I}SX' \\ \mu_X^{-1} & & \downarrow \mu_X^{-1} \\ \mathbb{T}\mathbb{I}X & \stackrel{\mathbb{T}If}{\longleftarrow} \mathbb{T}\mathbb{I}X \end{array}$$

Commutativity of the upper square follows from applying I to the diagram in (A.2) and commutativity of the lower square follows from the fact that μ is a natural isomorphism.

This shows that $\mathbb{I}f$ is a \mathbb{T} -coalgebra morphism from $(\mathbb{I}X, \gamma_{\delta})$ to $(\mathbb{I}X', \gamma_{\delta'})$. Define $\mathbb{X}f := \mathbb{I}f$.

Definition of \mathbb{Y} . Define $\mathbb{Y} : \mathsf{Coalg}(\mathbb{T}) \to \mathsf{Alg}(\mathbb{S})$ in a similar way, where $\mathbb{Y}(X, \gamma) = (\mathbb{J}X, \delta_{\gamma})$ and $\mathbb{Y}f = \mathbb{J}f$. Here δ_{γ} is the composition

$$\mathbb{SJ}X \xrightarrow{\eta_X} \mathbb{JT}X \xrightarrow{\mathbb{J}\gamma} \mathbb{J}X. \tag{A.3}$$

The natural isomorphism $\xi : \mathrm{Id}_{\mathsf{Alg}(\mathbb{S})} \to \mathbb{Y} \circ \mathbb{X}$. We need an algebra isomorphism $\xi_{(X,\delta)} : (X,\delta) \to \mathbb{Y} \circ \mathbb{X}(X,\delta) = (\mathbb{JI}X, \delta_{\gamma_{\delta}})$. That is, the following diagram must commute

$$\begin{array}{c} \mathbb{S}X \xrightarrow{\mathbb{S}\xi_{(X,\delta)}} \mathbb{SJI}X \\ \downarrow^{\delta} & \downarrow^{\delta_{\gamma_{\delta}}} \\ X \xrightarrow{\xi_{(X,\delta)}} \mathbb{JI}X \end{array}$$

Let $\xi_{(X,\delta)} := \epsilon_X$. We claim that with this definition the diagram above commutes. Obviously this yields an isomorphism. For commutativity:



Since I and J form an adjunction, we have $\mathrm{Id}_{\mathbb{J}} = \epsilon_{\mathbb{J}}^{-1} \circ \mathbb{J}^{1^{-1}}$ (see e.g. [40] section IV.1 theorem 1), so $\epsilon_{\mathbb{J}}^{-1} = \mathbb{J}_{1}$.

The upper square commutes trivially. The middle square commutes by definition of μ . The squares left and right of the middle square commute because ϵ is a natural transformation, as does the left lower square. For the middle right square, recall $\epsilon_J^{-1} = \mathbb{J}_1$. The middle and right lower squares commutes by definition of γ_{δ} and δ_{γ} . It follows that

$$\epsilon_X \circ \delta = \delta_{\gamma_\delta} \circ \mathbb{S}(\epsilon_X),$$

as desired. This gives the natural isomorphism $\mathrm{Id}_{\mathsf{Alg}(\mathbb{S})} \to \mathbb{Y} \circ \mathbb{X}$.

The natural isomorphism $\zeta : \mathrm{Id}_{\mathsf{Coalg}}(\mathbb{T}) \to \mathbb{X} \circ \mathbb{Y}$. This can be treated similar to ξ .

Conclusion. This proves that there is a dual equivalence $Alg(\mathbb{S}) \equiv^{op} Coalg(\mathbb{T})$.

As we often work with Stone duality and Isbell duality, the previous lemma provides a powerful tool. A.7 Lemma. (i) Pullbacks in Sob are computed as in Top.

(ii) Pullbacks in KHaus are computed as in Top.

Proof. (i) Let

$$\mathfrak{X} \xrightarrow{f} \mathfrak{Y} \xleftarrow{f'} \mathfrak{X}' \tag{A.4}$$

be a cospan in Sob. The pullback of this diagram in Top is $\mathcal{P} = \{(u, u') \in \mathfrak{X} \times \mathfrak{X}' \mid f(u) = f'(u')\}$ viewed as a subspace of $\mathfrak{X} \times \mathfrak{X}'$. Let $\pi : \mathcal{P} \to \mathfrak{X}$ and $\pi' : \mathcal{P} \to \mathfrak{X}'$ be projections. Since dual equivalences send limits to colimits and vice versa the diagram

is a pullback diagram in Top. By proposition 3.14 the cospan

$$\mathfrak{pt}(\mathfrak{opn}\,\mathfrak{X}) \xrightarrow{\mathfrak{pt}(\mathfrak{opn}\,f)} \mathfrak{pt}(\mathfrak{opn}\,\mathfrak{Y}) \xrightarrow{\mathfrak{pt}(\mathfrak{opn}\,f')} \mathfrak{pt}(\mathfrak{opn}\,\mathfrak{X}')$$

is isomorphic to the cospan in (A.4), hence $\mathfrak{pt}(\mathfrak{opn}\mathcal{P})$ is also a pullback of the cospan in (A.4). Since pullbacks are unique up to isomorphism $\mathcal{P} \cong \mathfrak{pt}(\mathfrak{opn}\mathcal{P})$. By definition 3.13 the latter is sober, therefore \mathcal{P} is sober.

(ii) Suppose the diagram of (A.4) is in KHaus and we take the pullback \mathcal{P} in Top. The product of two Hausdorff spaces is a Hausdorff space and the subspace of a Hausdorff space is Hausdorff, so \mathcal{P} is Hausdorff. Suppose $(u, u') \in \mathcal{X} \times \mathcal{X}'$ is not in \mathcal{P} . Then $f(u) \neq f'(u')$ in \mathcal{Y} so there exist disjoint open sets a, a' in \mathcal{Y} containing u and u' respectively. Then $a \times a'$ is open in $\mathcal{X} \times \mathcal{X}'$ and disjoint from \mathcal{P} . Therefore \mathcal{P} is closed. As the product of two compact sets is again compact and closed subsets of compact sets are compact, \mathcal{P} is compact.

A.8 Lemma. Let $f : A \to B$ be a Boolean algebra homomorphism. Then f is injective if and only if $f^{-1}(T_B) = \{T_A\}$.